

Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 13320

Negative Childhood Experiences and Risk Aversion: Evidence from Children Exposed to Domestic Violence

Marco Castillo

JUNE 2020



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 13320

Negative Childhood Experiences and Risk Aversion: Evidence from Children Exposed to Domestic Violence

Marco Castillo Texas A&M University and IZA

JUNE 2020

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Negative Childhood Experiences and Risk Aversion: Evidence from Children Exposed to Domestic Violence

Using a longitudinal study of 1,900 Peruvian children, I show that children who grow up in a household where mothers report experiencing domestic violence are more risk averse and have lower cognitive development. Risk attitudes are measured with an incentivized experiment. The effect of domestic violence on risk attitudes is not mediated by cognitive development and suggests that early negative experiences in life can directly influence the risk attitudes of children. This experience is associated with other behavioral changes as well, including lower physical activity and higher BMI.

JEL Classification:	C93, J13
Keywords:	domestic violence, cognitive development, risk preferences,
	children

Corresponding author:

Marco Castillo Department of Economics Texas A&M University College Station, TX 77843 USA E-mail: marco.castillo@tamu.edu

1 Introduction

According to the United Nations Children's Fund (UNICEF), 133 to 275 million children are exposed to domestic violence every year (UNICEF, 2006).¹ Exposure to domestic violence has been associated with a myriad of negative outcomes that include impaired cognitive development (Koenen, Moffitt, Caspi, Taylor, and Purcell, 2003), an increase in externalizing and internalizing behaviors (Emery, 2011; Fantuzzo and Mohr, 1999; Kernic, Wolf, Holt, McKnight, Huebner, and Rivara, 2003; Osofsky, 1999), and an increased likelihood of poor health conditions (Brown, Anda, Tiemeier, Felitti, Edwards, Croft, and Giles, 2009; Felitti, Anda, Nordenberg, Williamson, Spitz, Edwards, Koss, and Marks, 1998). Research suggests that adverse childhood experiences might not only be associated with lasting changes in the nervous, endocrine and immune systems (Danese and McEwen, 2012), but also with lifetime earnings through their effect on cognitive and non-cognitive abilities (Carrell and Hoekstra, 2010; Currie and Tekin, 2006; Gertler, Heckman, Pinto, Zanolini, Vermeersch, Walker, Chang, and Grantham-Zmcgregor, 2013; Heckman, Pinto, and Savelyev, 2013; Heckman, Stixrud, and Urzua, 2006). In this paper, I show that growing up in a household where domestic violence occurs can directly alter the risk attitudes of children.

The result is important because children make many consequential and unsupervised decisions and these decisions are correlated with their preferences (Bertrand and Pan, 2013; Castillo, Jordan, and Petrie, 2018, 2019; Castillo, Ferraro, Jordan, and Petrie, 2011; Golsteyn, Gronqvist, and Lindahl, 2014; Segal, 2013; Sutter, Kocher, Glaetzle-Ruetzler, and Trautmann, 2013). Early adverse experiences can then affect future outcomes by changing the way children evaluate options. In particular, increased risk aversion might be costly in the long-term if it prevents a child from taking advantage of opportunities available and/or if it stunts personal development. Given the prevalence of domestic violence, knowing the consequences of growing up in a toxic environment on child development is important.

This paper aims to improve our understanding of the determinants of individual preferences and the role the household environment plays in this process. Despite the fact that the household appears to be the most important environment in which a child's views, attitudes and capabilities are formed, we know little about how this environment relates to

¹According to (UNICEF, 2006), domestic violence or intimate partner violence refers to a pattern of assaultive and coercive behaviors including physical, sexual and psychological attacks, as well as economic coercion used by adults or adolescents against their current or former intimate partners. Examples of physical abuse include slapping, shaking, beating with a fist or object, strangulation, burning, kicking and threats with a knife. Sexual abuse includes coerced sex through threats or intimidation or through physical force, forcing unwanted sexual acts, forcing sex in front of others and forcing sex with others. Psychological abuse involves isolation from others, excessive jealousy, control of activities, verbal aggression, intimidation through destruction of property, harassment or stalking, threats of violence and constant belittling and humiliation.

individual preferences. This vacuum is especially surprising given that it is known that early experiences are crucial for the development of the child (Cirulli, Berry, and Alleva, 2003; Davidson and McEwen, 2012; Gunnar and Quevedo, 2007; Huttenlocher, 1979; Thompson and Nelson, 2001) and that traumatic events experienced as an adult can affect the risk attitudes of adults (Callen, Isaqzadeh, Long, and Sprenger, 2014; Eckel, el Gamal, and Wilson, 2009; Malmendier and Nagel, 2011; Voors, Nillesen, Verwimp, Bulte, Lensink, and van Soest, 2012). There is evidence that children's preferences differ by race and gender (Bettinger and Slonim, 2007; Cardenas, Dreber, von Essen, and Ranehill, 2012; Castillo et al., 2011; Gneezy and Rustichini, 2004; Levin and Hart, 2003), yet it is an open question whether early home experiences affect preferences as well. Living in a household with domestic violence is a natural starting point to investigate the relation between household environment and risk or economic preferences because of the large body of research showing a negative association with child development. Domestic violence is not only a widespread phenomenon, but it is one that highlights the fact that parents do not always engage in activities that are welfare enhancing for their children.

The findings in this paper are derived from a unique longitudinal study of a random sample of 1,900 Peruvian children whose households were surveyed when they were one, five, eight and twelve years old.² In the third visit, the risk attitudes of the children were measured by their selection of one of six possible lotteries that increased in mean and variance and included a sure payment option.³ The instrument used to measure risk preferences has many desirable characteristics: it applies standardized procedures, children made decisions without parental supervision, and they were paid in tokens that were redeemable for prizes of the child's choosing. Harbaugh, Krause, and Vesterlund (2002); Levin and Hart (2003); Moreira, Matsushita, and da Silva (2010); Schlottmann (2001); Weller, Levin, and Denburg (2011) show that children not only can handle these types of questions, but also that preferences can be elicited in this way. The present study investigates how the choices in this experiment relate to measures of domestic violence experienced by mothers within the household.

Measurement problems and endogeneity typically make it difficult to determine how parents' decisions affect the development of children. The evaluation of the effect of households with domestic violence on children's preferences is no exception.⁴ To deal with these issues,

 $^{^{2}}$ The panel is part of the Young Lives study on childhood poverty (http://www.younglives.org.uk/). The panel study also includes a survey when the children were 12 years of age.

 $^{^{3}\}mathrm{The}$ study did not collect information on time preferences.

⁴I construct measures of domestic violence based on reports by the child's mother as well as reports of seeking assistance to deal with this problems. While mothers' self-reports of domestic violence might be biased (Aizer, 2010; Ellsberg, Heise, Pena, Agurto, and Winkvist, 2001), they have the advantage of including events that the child was too young to remember and events that children less affected by the events might likely forget. According to these measures, 16.7% of the households in the sample reported

I follow Heckman et al. (2013) to establish a measurement system to measure domestic violence, parental investments and the child's cognitive and non-cognitive development. This is feasible because of the rich set of data available. To deal with the potential endogeneity of the household environment and parental investment, I follow two strategies.

First, I construct instruments for both parental investments and domestic violence. The instrument for parental investment is based on the introduction of a conditional cash transfer program (CCT) in Peru. Following the approach of Dahl and Lochner (2012) and Gruber and Saez (2002), a simulated instrument of changes in income is constructed using the introduction and deployment timing of the program. To deal with the potential endogeneity of the household environment, I build on Aizer (2010) to construct an instrument for domestic violence based on improvements in the gender wage gap. I confirm that the incidence of domestic violence decreases as the gender wage gap decreases. Using the introduction and timing of the opening of Women Emergency Centers (WECs) across Peru, a second instrument for domestic violence, I show that the timing of WECs is uncorrelated with pre-existing trends in domestic violence but predicts a drop in future levels of domestic violence. Using these constructed instruments in the analysis, I show that growing up in a household where domestic violence occurs leads a child to more risk-averse behavior and slower cognitive development.

Second, I estimate a structural model of child cognitive and preference development. The model is useful because it affords the treatment of both parental investments and domestic violence as endogenous variables. It also allows to model the trade-off between engaging in domestic violence and child development. Following Del Boca, Flinn, and Wiswall (2014), I estimate the model using simulated method of moments. The estimated model captures many of the features of the data, including the impact of domestic violence on risk preferences.

The model estimates show that parents care about both the cognitive development and the risk preferences of their children. Parents act as if they value their child to have the ability to take risks. The model is also helpful in interpreting various results in a coherent framework and allows for policy exercises to assess potential ways to foster balanced development of children. The estimates of the model are broadly in line with those in the literature (e.g Del Boca et al. (2014)). I test the robustness of this model by estimating a second structural model in which husbands use pecuniary and non-pecuniary (violence) to extract effort from their wives. This is a strategic model that follows Weinberg (2001)'s analysis of corporal punishment of children. This model fits the data less precisely than the unitary model, however, it shows that the effect of violence on risk preferences is robust

domestic violence at least once and 3.3% reported it at least twice.

to alternative modelling assumptions. Overall, the constructed instrument and structural modeling approaches show remarkable consistency and robustness of the results.

The main finding of the paper is that the existence of domestic violence in the household is positively associated with risk aversion in children. The estimated effect can be as large as -0.66SD decrease on the average lottery choice. Also, the measure of domestic violence is associated with a significant decrease in the cognitive development of the child. The estimated effect can be as large as -0.3SD decrease of the measure of cognitive development. To increase confidence in the results, I explore if these negative effects manifest in other outcomes as well. Children in households with domestic violence are less physically active, spend less time playing and have higher BMIs. These patterns are also present in the survey given 4 years after the experiment. Increased risk aversion and inactivity point to domestic violence leading to internalizing behaviors.

It is important to note that the effect of domestic violence on children's development and behavior has the potential to be underestimated when addressing measurement and endogeneity issues. This would be consistent with the existence of compensatory behaviors in the household and with domestic violence being equivalent to the absence of parental investment. I present evidence that a composite index of maternal investment that includes investment in time and accounts for domestic violence rationalizes the main patterns in the data.

The emerging picture from this study is that an important component of parenting is providing children a safe home environment. This coincides with the recent literature trying to unpack several components of parental investments (e.g. Aizer and Cunha, 2012; Del Boca et al., 2014). I provide evidence that the effect of domestic violence mimics that of a loss in maternal investment in time and attention. To my knowledge, this is the first paper to show that parents can affect their child's risk preferences and behavior, even if done unintentionally.

An important open question is whether risk aversion is a maladaptive response to domestic violence. For instance, an increase in risk aversion might be accompanied by a reduction in the willingness to compete. If so, a possible economic cost of domestic violence would include not only the loss in potential human capital accumulation but also the opportunity cost of not choosing more advantageous, but perhaps riskier, options. While this might keep a child out of trouble, it might also prevent him/her from acquiring social skills, which are useful in the job market as well as in life in general. Similarly, if learning requires taking risks, exposure to domestic violence might have additional costs by discouraging experimentation.

The analysis shows that there might be ways to mitigate the prevalence of domestic

violence, such as through the improvement of women's labor prospects (Aizer, 2010) or an increase in the cost of exercising violence against women. Some interventions might be less costly than others. For instance, there is evidence that information campaigns are effective in deterring violence against women (see Ellsberg, Arango, Morton, Gennari, Kiplesund, Contreras, and Watts (2014); Pronyk, Hargreaves, Kim, Morison, Phetla, Watts, Busza, and Porter (2006)). Given the potential long-term effect of domestic violence on abilities and preferences, additional research on cost effective ways to prevent domestic violence is needed.

The paper is organized as follows. Section 2 provides background literature. Section 3 presents a simple intra-household model with parental investments in child development and domestic violence. Section 4 describe the estimation procedures and assumptions. Section 5 describes the sample and measures used in the analysis. Section 6 present estimates using an instrumental variable approach. Section 7 present structural estimations. Section 8 concludes. Additional material is provided in a series of appendices.

2 Background literature

There is a large body of animal and human research showing that early life experiences have long-term impacts on cognition and behavior. For instance, children exposed to stress and emotional deprivation experience changes in the nervous, endocrine and immune systems (Brown et al., 2009; Danese and McEwen, 2012; Danese, Pariante, Caspi, Taylor, and Poulton, 2007; Felitti et al., 1998; Gunnar and Quevedo, 2007). Research also shows that children living in a household with domestic violence can exhibit behavioral changes such as internalizing and externalizing behaviors (Emery, 2011; Fantuzzo and Mohr, 1999; Kernic et al., 2003; Osofsky, 1999) and impaired cognitive development (Koenen et al., 2003), thereby potentially worsening economic outcomes in life (Gertler et al., 2013; Heckman et al., 2013, 2006).

A mother's depression, which is strongly associated with domestic violence, can have similar effects because it has been found to affect the quality of the relationship with the child (Murray and Cooper, 1997; Tomlinson, Cooper, and Murray, 2005). Moreover, there is evidence from non-human species (e.g. rats and primates) that maternal care affects fearfulness in rats (Caldji, Tannenbaum, Sharma, Francis, Plotsky, and Meaney, 1998) and that naturally occurring variations in maternal care can alter the expression of genes that regulate behavioral and endocrine responses to stress (Francis, Diorio, Plotsky, and Meaney, 2002; Meaney, 2001). There is also evidence that these effects are causal (Meaney, Aitken, Bodnoff, Iny, Tatarewicz, and Sapolsky, 2013) and that early adverse conditions can be diminished with improved care later in life (Chisholm, Carter, Ames, and Morison, 1995; Gertler et al., 2013; Nelson, Zeanah, Fox, Marshall, Smyke, and Guthrie, 2007).

This literature suggests at least two channels through which exposure to domestic violence might affect risk preferences. The first channel is through its effect on cognitive development. Studies with adults show a consistent negative relationship between cognitive ability and risk aversion (Burks, Carpenter, Goette, and Rustichini, 2009; Dohmen, Falk, Huffman, and Sunde, 2010). More risk aversion is therefore expected among children exposed to domestic violence if the relationship between cognitive ability and preferences appears in early childhood.

A second potential channel through which domestic violence might affect risk preferences is by its effect on stress. For instance, Chen, Cohen, and Miller (2010) observe that the positive relationship between lower socieconomic status and higher levels of cortisol among children was partially mediated by their perceptions of threat and by family chaos. Bair-Merritt, Johnson, Okelo, and Page (2012) show that children exposed to intimate partner violence present elevated levels of cortisol. Finally, Korte (2001) discusses how the hormonal system regulating fear and anxiety can become maladaptive when it is chronically unbalanced due to stress. This is likely to occur in the face of prolonged stress and is important because recent evidence from Kandasamy, Hardy, Page, Schaffner, Graggaber, Powlson, Fletcher, Gurnell, and Coates (2014) shows that an experimentally-induced elevation of cortisol levels over a period of 8 days causes subjects to behave more risk aversely.⁵ If children exposed to domestic violence have elevated levels of cortisol due to stress, we should expect them to behave more risk aversely.⁶ Importantly, absent an increase in the base level of cortisol, threat of violence might itself affect behavior. Haushofer and Fehr (2014) show that risk aversion can be induced by the threat of receiving random levels of high or low electrical shocks. The threat of electrical shocks is a way to experimentally induce fear and stress. Finally, Harrison, List, and Towe (2007) also show that background risk can increase risk aversion.

The current literature suggests that children exposed to domestic violence might behave more risk aversely because they experience more stress and/or more uncertainty. The data used in this study do not include measures of stress, however, the prevailing evidence suggests this could be an underlying mechanism for the observed risk-averse behavior.

⁵This effect was found to be larger among men.

⁶Gunnar and Quevedo (2007) shows that prolonged exposure to stress can lead to *lower* levels of cortisol (hypocortisolism). However, this condition is associated with a lower responsiveness to incentives. Evidence of this effect in children can be found in Ouellet-Morin, Danese, Bowes, Shakoor, Ambler, Pariante, Papadopoulos, Caspi, Moffitt, and Arseneault (2011).

3 Modelling parents decisions and children's preferences

Why would parents engage in behaviors that jeopardize the development of their children if they care about them? In this section, I describe a simple household model in which domestic violence is treated as a consumption good that produces a negative externality. I use this model to motivate the trade-off faced by parents between indulging in potentially harmful habits and aiding the development of their child. This highlights the endogeneity of both parental investments as well as domestic violence. It also stresses that parents always have to make trade-offs between their own consumption and that of their children. Estimates of the model will be discussed Section 7 after presenting estimates of the effect of domestic violence on child outcomes using instrumental variables in Section 6. I will also discuss results using alternative modelling assumptions like the strategic use of domestic violence to extract effort from wives. All approaches lead to similar conclusions.

Parents maximize utility subject to budget and technology constraints. The following program summarizes the decisions of parents:

$$\operatorname{Max}_{C,l_m,l_f,h_m,h_f,\tau_m,V,I} \alpha_c ln(C) + \alpha_m ln(l_m) + \alpha_f ln(l_f) + \alpha_V ln(V) + \alpha_\theta ln(\theta_1) + \alpha_\sigma ln(\sigma_1)$$

subject to:

$$\begin{array}{l} \text{(i)} \ C+I+p_VV \leq w_mh_m + w_fh_f + Y \\ \text{(ii)} \ T = h_m + l_m + \tau_m \\ \text{(iii)} \ T = h_f + l_f \\ \text{(iv)} \ ln\theta_1 = A + a_\theta ln(\theta_0) + a_{\theta_m}ln(\theta_m) + a_{\tau_m}ln(\tau_m) + a_Ilog(I) + a_Vlog(V) + a_XX + \epsilon_\theta \\ \text{(v)} \ ln\sigma_1 = B + b_\sigma ln(\sigma_0) + b_{\theta_m}ln(\theta_m) + b_{\tau_m}ln(\tau_m) + b_Ilog(I) + b_Vlog(V) + b_XX + \epsilon_\sigma \end{array}$$

where C is household consumption, I are parental investments in money, Y is predetermined income, l_m is the mother's leisure time and l_f is the father's leisure time, τ_m is the mother's investment in time in the child, h_m is the mother's working hours and h_f is the father's working hours. V is an index of domestic violence, θ_1 is an index of cognitive development at the end of the decision period and θ_0 is a measure of cognitive development at the time decisions are made. σ_1 is a measure of the child's risk-taking behavior at the end of the period and σ_0 is a measure of risk-taking behavior at the beginning of the decision period. In this case, σ_0 is not observed by the researcher. Parents face five constraints, a budget constraint, two time constraints and two technological constraints. The production functions depend on τ_m , I, θ_0 , V and a set of additional covariates X.⁷

Some comments about the model specification are in order. First, I assume a Cobb-Douglas utility function and Cobb-Douglas production functions for simplicity. They pro-

⁷In the estimation, X includes an index of mother's capital (θ_m), family size (*hh*) and wealth (ω).

duce closed-form solutions, and as I will discuss in Section 7, they capture many features of the data. For presentation purposes, I will delay discussion of the estimation of the model until later. Second, domestic violence is modelled as a consumption good. This assumption is strong in the sense that the possibility that violence might be used to extract effort from a spouse is excluded. However, I discuss in Appendix F estimates of a model that allows for the instrumental use of violence, and this yields similar results. I also assume that there is a 'market' for violence. This assumption aims to represent the existence of social costs associated with exercising 'violence' on others (e.g. lost friendships, purchase of presents to repair relationships).

The model makes clear that if parents care about the cognitive development and preferences of their children, parental investments and domestic violence should be considered endogenous variables. This is the basic insight from the economics literature on child development (e.g. Cunha, Heckman, and Schennach, 2010; Heckman et al., 2013). The model also suggests that exogenous variations in income and the cost of exercising domestic violence can help identify the causal impact of both parental investments and domestic violence on children's development. I will present evidence consistent with this prediction in section 6.

4 Methods

Testing the empirical implications of the model presents several challenges. First, both parental investments and domestic violence are endogenous variables. Second, cognitive development, parental investments and domestic violence are not directly observable. In this section, I adapt the framework from Heckman et al. (2013) to extract latent variables from the survey data.⁸

I consider a simple linear approximation to determine the cognitive abilities (θ_1) and risk preferences of the child at age 8. In particular, the empirical model for outcome y_i^k of child *i* is,

$$y_{1,i}^k = \alpha_{I_0}^k I_{0,i} + \alpha_{V_0}^k V_{0,i} + \alpha_{\theta_0}^k \theta_{0,i} + \alpha_{\theta_M}^k \theta_{M,i} + X_i' \beta_i^k + \varepsilon_i^k \tag{1}$$

In this expression, X'_i represents a set of additional determinants of outcomes and ε^k_i represents factors affecting the child that are not observed by the researcher.

The empirical model above is limiting. The research on child development (Cunha et al., 2010) demonstrates that there are important complementarities between different inputs in

⁸Attanasio, Meghir, and Nix (2015) use a related approach with the Young Lives sample from India. In their study, they estimate a model of child development similar to that in Cunha et al. (2010). They consider parental investments in money and time. Only the survey from Peru has information on domestic violence.

the production of cognitive and non-cognitive abilities. The simplifying assumptions of the model make the estimation tractable, and I discuss the results in light of these assumptions.

4.1 Measurement and Estimation

The model described above has four main components: parental investments, mother's human capital, domestic violence, cognitive development and risk attitudes. Except for risk attitudes, which are measured with an experiment, all the other variables are assumed to be measured with error. In other words, the estimation assumes that neither parental investments, mother's human capital, domestic violence or cognitive development are directly observed, but that only variables imperfectly measuring them are observed. The assumption that risk attitudes are measured without error is made out of necessity because the data do not have alternative measures of either risk attitudes or other preferences. The fact that the results are consistently strong despite this shortcoming suggests that this assumption is not limiting. Table 2 shows the descriptive statistics of all the measures used in the estimations and the variables to which they are dedicated.

I follow Heckman et al. (2013) to develop a measurement system for the latent variables in the model. Appendix B shows how to use derived measures of the latent factor in regression analysis and how to correct these estimates to address the fact that latent factors are measured with error. The basic assumption of the model is the existence of measures $\hat{M}_{m_j}^j, m_j \in \mathcal{M}_j$ for each of the latent factors $j \in \mathcal{I}$. Measures are standardized, and the basic measurement equation is:

$$\hat{M}_{m_j}^j = \lambda_{m_j}^j \theta^j + \varepsilon_{m_j}^j, j \in \mathcal{I}, m_j \in \mathcal{M}_j \tag{2}$$

I further assume that latent variables are zero mean with covariance matrix Σ and that $COV(\varepsilon_{m_j}^j, \varepsilon_{m_k}^k) = 0$, all $j, k \in \mathcal{I}, m_j \in \mathcal{M}_j, m_k \in \mathcal{M}_k$. That is, I assume that the noise terms are uncorrelated for all measures. Under the additional normalization assumption that $\lambda_1^j = 1$ for all $j \in \mathcal{I}$, Heckman et al. (2013) show that the measurement system can be identified from the covariances of measures if there are at least three measures per latent factor.⁹

The vector of measures for person *i* can be expressed as $M_i = \Lambda \theta_i + \varepsilon_i$, with $COV(\varepsilon_i, \varepsilon_i) = \Omega$ and ε_i independent of θ_i . An unbiased estimator of θ_i is:

$$\hat{\theta}_i = (\Lambda' \Omega^{-1} \Lambda)^{-1} \Lambda' \Omega^{-1} M_i \tag{3}$$

⁹In the absence of serial correlation, only two measures are needed. I identify the parameters of the cognitive development of a child at age 5 under this assumption due to the fact that in the second round of the study only two measures of cognitive development were available.

This formula is used in the analysis.

5 Sample selection and Data

This section discusses the sample and variables used in the empirical analysis.

5.1 Sample

The original sample included 2,052 children ages 6 to 18 months. The sample was selected in a series of steps. First, 1,818 districts were ranked according to the 2000 Fondo Nacional de Compensacion y Desarrollo Social (FONCODES, 2001) poverty index. The index aggregates information on infant mortality, housing, schooling, roads, and access to services. Districts ranked at the top 5% of the distribution were excluded to over-represent poor districts. Each of the remaining districts was subdivided in geographical areas of similar population size and then 20 of these units were selected for the study. Each of these 20 units was further subdivided into census tracks and one track was selected at random in each unit. All the households in the selected track were visited to identify if the household had a child in the desired age range. Finally, neighboring census tracks were visited until 100 eligible households were surveyed. Escobal and Flores (2008) compared the current sample with the 2000 Demographic and Health Survey, the 2001 Living Standard Measurement Survey and the 2005 Population census. They find that the sample of children are slightly richer than these other samples. While there is no information regarding refusals to participate in the study, Outes-Leon and Dercon (2008) document that attrition between the first and second round of the study is small and mostly random. The households were visited in four waves: 2002, 2006, 2009 and 2012.

5.2 Experiment

In the third wave of the study (2009), children were asked to choose one out of 6 possible lotteries that paid in tokens depending on a coin flip. The lotteries increased the mean and variance of payoffs (Binswanger, 1980; Eckel and Grossman, 2008) and were simple enough for the children to understand. The lotteries were paid to promote truthful revelation of preferences (Harbaugh et al., 2002), and the tokens were redeemable for prizes (i.e. stickers) of the child's choosing. Only 24 children of a total of 1,943 interviewed in the third wave have missing data on the lottery task. Of these 24 children, only one is reported to have refused to answer the lottery question. The instrument the children faced is shown in Figure 1. The first option is a sure payment. Then, to distinguish between risk neutral and risk taking children, the last option only increased the variance of the lottery but keeps the expected payoff constant. Figure 2 shows the distribution of lottery choices for boys and girls (panel a) and the distribution of lottery choices by children with and without reports of domestic violence in the 3rd survey (panel b). Children in households reporting domestic violence in the 3rd survey behave markedly different than children in households where domestic violence is not reported.

5.3 Measures of exposure to domestic violence

The measure of domestic violence in the first wave of the survey is based on the mother's answer to the question: "When [your partner] gets drunk does he hit you?" This question was answered in the affirmative 6 percent of the time. In the second and third waves of the survey, the measure of domestic violence is based on the mother's answer to the question: "When [a family member] gets drunk does he/she turn aggressive?" This question was answered in the affirmative 8 and 7 percent of the time in the second and third waves of the survey. The wording in the second and third survey has the advantage of being more inclusive. This might explain why the prevalence of events of domestic violence is slightly larger in the latter surveys. For instance, none of the 162 mothers who were divorced or separated in the first survey reported an incident of domestic violence, but 5 out the 128 mothers who were divorced or separated in the second and third survey reported an incident of domestic violence.

Information from the third survey suggests, however, that answers likely refer to the behavior of the mother's partner.¹⁰ According to the third wave of the survey, 12% of the adults in the household were grand or great-grand parents and 17% percent were other adults (i.e. uncles/aunts, siblings, cousins, etc.). The percent of households ever reporting a case of domestic violence is 16.7%, and the percent of households reporting cases at least 2 times is 3.3%. These numbers are comparable to those found in developed countries (Hedin and Janson, 2000; McFarlane, Parker, and Soeken, 1996). The third wave of the survey also includes a question on whether the mother ever asks for help due to domestic violence problems. Twelve and a half percent responded in the affirmative. This variable will be used as an additional measure of domestic violence. Finally, the presence of smokers (other than the mother) will also be used as an additional indicator of the quality of the home environment.¹¹

According to the World Health Organization (WHO, 2013), women who have been physically or sexually abused by their partners are almost twice as likely to experience de-

¹⁰Question about domestic violence are not available in the fourth wave of the survey.

¹¹The correlation between the answer to the domestic violence question and the answer to the question about asking for help is 0.146 (p-value < 0.001). The correlation with the number of smokers is 0.118 (p-value < 0.001).

pression. I confirm a similar pattern in the data by comparing indicators of the mother's depression using a 20 Yes/No questionnaire developed by the World Health Organization (WHO, 1994) for this purpose. A mother is considered to be at risk of depression if she responds yes to at least 8 of these 20 questions. According to this measure, mothers reporting domestic violence were about twice as likely to show signs of depression compared to other mothers (50% v. 28%, p-value < 0.001 in the first survey and 24% v. 12%, p-value < 0.001 in the second survey).¹² The measure used in this paper reproduces previous results of the effect of domestic violence on mothers' depression. Importantly, the effect does not seem to be due to the existence of time-invariant omitted variables. A regression of the change in the index of depression on the change in the reports of domestic violence, the change in the wealth index, and the change in marital status shows that domestic violence is associated with a 12 percentage points increase (p-value = 0.002) in the likelihood of being depressed.

The total number of reported instances of domestic violence is negatively correlated with the mother's years of education (r = -0.09, p-value < 0.001), the wealth index (r = -0.08, p-value < 0.001), being a single mother during the first survey (r = -0.06, p-value = 0.004) and positively correlated with the number of children (r = 0.09, p-value = 0.001), the father's childhood experience of domestic violence (r = 0.08, p-value = 0.003), the mother's childhood experience of domestic violence (r = 0.09, p-value = 0.001), and recent experience of a bad shock (r = 0.06, p-value = 0.009). Similar results are obtained using the disaggregated reported instances of domestic violence.

5.4 Measures of cognitive development and controls

Five measures of cognitive development are available in the 2nd and 3rd wave of the study. Children were administered the Peabody Picture Vocabulary Test (PPVT) at 5 and 8 years of age. The test's main objective is to measure vocabulary acquisition from 2.5 years of age to adulthood and consists of giving a person a stimulus word to match with a picture. The questions increase in the level of difficulty as the test progresses. There is evidence that the PPVT is strongly correlated with measures of intelligence (Campbell, Bell, and Keith, 2001). The average number of correct answers was 29.1 (s.d. 17.8) at 5 years of age and 46.7 (s.d. 13.5) at 8 years of age. Cueto and Leon (2012); Cueto, Leon, Guerrero, and Munoz (2009) provide detailed information on the validity of all tests for the current sample of children. For comparison, results on similar samples of children from India, Vietnam and Ethiopia are 27.4 (s.d. 21.1), 37.0 (s.d. 18.2) and 21.4 (s.d. 12.4) at age 5 and 49.2 (s.d. 26.7), 76.9 (s.d. 23.8) and 68.4 (s.d. 36.8) at age 8.

Children were also administered the Cognitive Development Assessment (CDA) devel-

 $^{^{12}}$ Answers to questions about depression are not available in the third survey.

oped by the International Evaluation Association. The test has three components: spatial relations, quantity and time, but only the quantity portion of the test was collected by the survey. The quantity portion of the test requires the child to indicate which picture in a set of pictures best represents a description given by the examiner. Notions such as a few, most, half, many, equal, etc. are evaluated by asking questions such as: 'point to the plate that has a few cupcakes.' This test was administered in the second survey and had on average 8.4 (s.d. 2.2) correct answers. For comparison, results on similar samples of children from India, Vietnam and Ethiopia are 9.4 (s.d. 2.6), 9.8 (s.d. 2.5) and 8.2 (s.d. 3.0).

Finally, math and reading tests were administered in the third wave of the study. The math test measures basic quantitative and number notions, including questions on counting, knowledge of numbers, number discrimination, and the use of basic operations. Questions were read by the field worker with the aid of cards, so that poor reading skills would not impact the test. The second section of the test measures the ability to perform basic mathematics operations with numbers (i.e. addition, subtraction, multiplication and division). The average number of correct answers was 11.9 (s.d. 4.9) out of 29 questions. The reading test included writing and reading comprehension. The average number of correct answers was 7.7 (s.d. 3.0) out of 13 questions.

The pairwise correlations of the instruments are all significantly different from zero. The smallest is 0.379 between the cognitive test and math test. The largest is 0.657 between the two PPVT tests. The Cronbach scale reliability coefficient across all 5 measures is 0.697. Factor analysis confirms that the highest eigenvalue for all these measures is 3.042. Parallel analysis (Horn, 1965) suggest the existence of a sole common latent factor. In the analysis, I construct two indices, one for cognitive development at 8 and one for cognitive development at 5.

To make the measures comparable across children, the effect of age, gender and location of the child at the time of measurement is removed using linear regressions. The measures of the mother's human capital are years of schooling, the ability to read in her mother's language and whether she is a Spanish speaker (the dominant language of Peru). Household structure is approximated by household size and whether both parents live together. I also include a wealth index to control for the child's socio-economic status. The index has been shown to be a good substitute, and sometimes a better alternative, to measures of household consumption (Filmer and Pritchett, 1999, 2001). The index has three main components: housing quality, consumer durables and services. The index is calculated according to the information in the first survey and has a mean of 0.42 (s.d. 0.19, min 0.03, max 0.83).

The surveys have only a handful of indicators of the child's field behavior. All the surveys includes measures of the child's BMI. These are based on measurements performed by the enumerators who were trained for the task. The enumerators used a scale to measure weight and a metric measuring stick for height. The later surveys include the average time per day a child spends playing and studying.¹³ Finally, the third survey includes the mother's answers to the following two questions: "During the last 7 days, on how many days was [Name] physically active for at least 60 minutes at one time? (Examples for physical activity would be running, biking, dancing, football, digging, carrying water, or other activities) 00 = 0 days, 01= 1 day, 02= 2 days 03= 3 days 04=4 days 05=5 days 06=6 days 07= 7 days (every day)" and "How much time does [Name] spend during a typical day sitting (school, work, watching TV and sitting with friends)? 01= Less than 1 hour per day, 02= 1 to 2 hours a day 03= 3 to 4 hours a day 04= 5 to 7 hours a day 05= more than 7 hours". I use these variables to explore the robustness of the main results.

5.5 Instrumental variables

The theory section suggests that, conditional on wealth and family structure, variations in current income and the cost of domestic violence should influence parental investment and domestic violence. I follow Dahl and Lochner (2012) and Gruber and Saez (2002) to construct a simulated instrument of change in income due to the introduction and timing of deployment of a conditional cash transfer (CCT) program in poor areas of Peru. The CCT program started in 2005 and was deployed in phases according to overall poverty levels. The CCT program defined eligibility using a known formula which was based on family characteristics (the actual formula used for eligibility is presented in Appendix B).¹⁴ Since families could potentially manipulate their characteristics and location in order to qualify for the program, I calculate eligibility based on the characteristics of the household at the time of the first wave of the Young Lives (YL) study. The YL study started several years prior to policy discussions leading to the CCT program which makes calculations based on lagged variables less susceptible to manipulation.

Due to the timing of the policy, mainly after the second wave of the YL study, I can simulate changes in income between the second and third wave.¹⁵ In particular, I use income in the second wave and the wealth index in the first wave of the YL study to

¹³The actual wording of the question is: "Now, think about the rest of NAMEs day. I want you to tell me how much time NAME spent on the following activities during a typical day?"

¹⁴Conditions varied by the age of participant. Members of households with children younger than five years of age or with a pregnant or lactating woman present were required to attend regular health care visits. Children aged 6 to 14 years old who had not completed primary school were required to attend school 85% of school days. Beneficiary households received transfers of 100 soles (about US\$30) each month regardless of household composition, representing ~15% of household spending (Andersen, Reynolds, Behrman, Crookston, Dearden, Escobal, Mani, Sanchez, Stein, and Femald, 2015)

¹⁵Unfortunately, I cannot calculate changes in income between the first and later rounds of the YL since data on income were not collected in the first round.

predict pre-CCT income in the third wave. I use a fifth-order polynomial of the second wave income, a dummy variable for reporting zero income and the wealth index in the first wave to predict pre-transfer income in the third wave (as in Dahl and Lochner (2012)). I use the simulated eligibility, based on the actual formula by the Peruvian government and household characteristics in the first wave of the panel, and the deployment schedule of the CCT, to predict whether a family would have received a transfer. The predicted income plus the predicted transfer are used to calculate a predicted income in the third wave of the study. Predicted changes in income across waves are calculated by subtracting the income in the second wave from the predicted income in the third wave.

The predicted change in income is used as an instrument for parental investment. Predicted changes in income are less likely to be correlated with pre-existence differences across sites which in turn are correlated with the CCT deployment schedule.¹⁶ In the estimations, I augment the set of controls by including a polynomial of second wave income and whether the family reported no income in the second wave of the study. I do this to reduce biases due to factors such as measurement error, regression to the mean, and serially correlated income (see Dahl and Lochner (2012)). Finally, I also include the value of the eligibility index to make sure that the results are due to changes in income and not due to eligibility. Credibility of this instrument is anchored by pre-CCT program variables and the fact that they are unlikely to correlate with anticipated shocks. The main drawback of the variable is its conditionality. However, the CCT program would be expected to overestimate the effect of parental investment because the conditionality was based on parents providing more investment (e.g. increased schooling). If domestic violence is significant because it is picking up parental investment, then this approach should make that explanation less plausible.

To account for changes in the bargaining position of women, I construct a measure of the gender wage gap. I rely on a national representative sample of Peruvian households (ENAHO) to calculate the market wages of men and women. The Peruvian National Household Survey (ENAHO) collects information from a random sample of 2,200 households every month in order to assess living standards. The 2009 sample included 26,988 household and 17,285 individual observations on wages.¹⁷ To diminish the possibility that local characteristics influence local wages, I use the average of a neighboring province in the calculation of both men and women's wages. This reduces the chances that local prevailing levels of violence against women manifest themselves as lower wages due to productivity losses.

I also consider a second instrument for domestic violence: whether a Woman Emergency Center (WEC) was opened in the YL site between 2005 and 2008 (prior to the 3rd wave

¹⁶The estimations are robust to the inclusion of site-level average wealth level.

¹⁷There is a community survey associated with the Young Lives study that contains some information on wages. However, this information is collected from informants, and it is selective and incomplete.

of the survey). I do this because recent research showing the effect of WEC in reducing domestic violence in Peru (Trako, Sviatschi, and Kavanaugh, 2018) and my conversations with officials at the Peruvian Ministry of Women and Disadvantage Populations suggest that the WECs were not opened in response to changes in cases of domestic violence but rather because of the availability of space for a center. Table A.1 in Appendix A uses several Peruvian Demographic and Health Surveys (DHS) to show that past levels of domestic violence do not predict the creation of WECs. There is no consecutive yearly DHS data prior to 2004, so the estimations use data from year 2004 onwards. In the estimations, I use changes in the availability of a WEC, rather than the existence of a WEC, to reduce the potential influence of pre-existing differences across sites.

6 Results

This section presents the estimations on the relationship between domestic violence and risk aversion. The evidence suggests that being in a household where a mother reports experiencing domestic violence increases the level of risk aversion of the child.

6.1 Measurement

Table 1 presents basic descriptive statistics of sample, and Table 2 presents information on the relation between the measurement variables and the latent factors they represent. Table 3 presents the information content of each latent variable measure. I confirm that, for this sample, measures of latent variables are noisy. The table shows that measures of cognitive development are more informative than those associated with parental investment, domestic violence and the household environment. Explanatory factor analysis suggests that the current grouping of variables is appropriate. With the exception of parental investment, for which there is some evidence of a second latent variable and is discussed further in Section 6.4, I find that at most one latent factor exists. The largest eigenvalue is 2.25 (second largest 0.40) for the measures of cognitive development at age 8, 2.11 (second largest 0.49) for cognitive development at age 12, 1.56 (second largest 1.17) for parental investment, 1.35 (second largest 0.96) for domestic violence, 1.47 (second largest 0.53) for cognitive development at age 5, and 1.98 (second largest 0.59) for mother's capital.¹⁸

 $^{^{18}{\}rm These}$ calculations are performed after the effects of age, sex and location have been removed from each measure.

6.2 Preliminary results

Table 4 presents correlation coefficients between the lottery decision, indices of the mother's psychological malaise and an indicator variable that equals 1 if the child had a serious injury (e.g. burns, broken bones, etc.).¹⁹ The decision in the risk experiment is correlated with the gender of the child, reports of domestic violence in the 2nd and 3rd surveys and the manner in which mothers react to misbehavior by the child. Reasoning with the child is negatively correlated with taking risks in the experiment, and using corporal punishment is positively correlated with taking risks in the experiment. The mother's psychological malaise is positively correlated with reports of domestic violence. This is also positively correlated with a child's injuries and cognitive development at age 5. This is consistent with the mother's wellbeing being important in providing effective care for the child. Finally, mothers with higher indices of psychological malaise are more likely to report asking for help on issues related to domestic and child abuse in the third wave of the panel.

Table 5 presents fixed-effect regressions of the mother's psychological malaise index, the child's serious injuries and cognitive development on reports of domestic violence made by the mothers. These regressions can be implemented due to the fact that there are repeated measures of some of these variables. Reports of domestic violence are positively correlated with the mother's psychological malaise and negatively correlated with measures of the child's cognitive development. Domestic violence is not correlated with a child's serious injuries, but serious injuries are positively correlated with the mother's psychological malaise. These estimates show that the relationship between domestic violence, the mother's wellbeing and the child's development are not due to omitted time-invariant variables. While this is suggestive that the relationship between domestic violence and child development is not spurious, these regressions do not address the problem of endogeneity. For instance, the health of the mother and a lower than expected cognitive development of the child might trigger domestic violence. Results that address the issue of endogeneity are presented next.

6.3 Instrumental variable estimates

This section presents the IV regressions of the effect of parental investments and domestic violence on the child outcomes of interest. Table 6 shows the first- and second-stage regressions for cognitive development and lottery choice for completeness.²⁰ Predicted changes in income strongly predict parental investment and the gender wage ratio, and the introduc-

¹⁹The indices are the common factor of the 20-item signs of depression questionnaire for the 1st, 2nd and 4th survey and the life satisfaction questionnaire for the 3rd survey.

 $^{^{20}}$ As discussed in Appendix B, parameter estimates might be biased due to measurement error. The estimates presented in the paper do not correct for this possibility for clarity of exposition and due to the fact that the results are similar if a correction is used.

tion of a WEC predict a reduction in the prevalence of domestic violence. The instrumental variable approach helps uncover the importance of parental investments but also shows that the effect of intimate partner violence is robust. The identified effect of domestic violence does not appear to be due to an underestimation of the effect of parental investment.

Tables A.2 and A.3 in Appendix A show the effect of domestic violence first, and domestic violence and parental investment second, using simulated changes in income, provincelevel female-male wage gaps and whether a WEC was opened, for a larger set of outcomes. These tables show a strong correlation between the measure of domestic violence and cognitive development, risk aversion, BMI and play both contemporaneously and four years after risk preferences were measured.

Table A.4 presents the estimates excluding the change in the presence of a WEC as an instrument. The results are similar, but they are less precisely estimated. Table A.5 presents the estimates on the subsample of households that would have likely qualified for a CCT if the program was universal at the time of its inception. This provides a comparison group closer to those predicted to receive a CCT earlier. Because the threshold used to assign CCTs was significantly lower after 2007 and the qualification process was relaxed to make it easier to receive a transfer, I use a lower threshold than the original one (0.600 instead of 0.756). Table A.5 shows that the results are comparable despite the loss of power. Finally, Table A.6 in Appendix A compares the coefficient estimates of the equation for cognitive development and lottery decisions accounting for measurement error. Measurement error occurs due to the fact that estimated latent variables are only approximations. Formulas to correct for this bias in the IV regressions are provided in Appendix B. Parameter estimates are very close to the estimates without the measurement error correction suggesting that measurement error is not severe in the sample.

In sum, the estimates using instrumental variables point to a direct effect of domestic violence on cognitive development and risk aversion that is separate from parental investments.

6.4 Domestic violence and parental investments

The measure of parental investment combines both time and money investments. Del Boca et al. (2014) have shown that parental investment of time and money are different and not equally effective. It stands to reason that more refined measures of parental investments might explain child outcomes better. Table A.7 shows confirmatory factor analysis conducted on the measures of monetary parental investment, time investment and domestic violence. Importantly, the analysis shows that these are three different concepts. I next estimate the effect of investment of time and domestic violence on cognitive development and risk attitudes separately.

Table A.8 shows the negative symmetry between parental investment in time and domestic violence. Empirically, however, I find that variables affecting a mother's investment in time also affect the presence of domestic violence. In other words, it is not possible to identify simultaneously the separate effect of investments of time and domestic violence given the data. Figure A.1 uses the approach suggested by Chernozhukov and Hansen (2008a) to identify the set of parameters for parental investment of time and domestic violence that is not rejected by the data.²¹ Figure A.1 shows that for both cognitive development and lottery choices, the hypothesis that the coefficients of both parental investment in time and domestic violence are simultaneously equal to zero can be rejected.

This analysis suggests creating a *domestic violence adjusted* index of parental investment in time. In particular, I construct such an indicator by simply subtracting the index of domestic violence from the index of parental investment in time. This particular assumption is consistent with the estimates presented in Figure A.1. Table A.9 shows regression results using this indicator and an indicator of parental investment of money. I follow the instrumental variable approach described in the previous sections. Table A.9 shows that parental investment of time minus domestic violence consistently affects indicators of child development and behavior. For instance, a one-standard deviation in the domestic violence adjusted index of mother's investment of time leads to approximately a 0.4SD increase in the cognitive development index and a 0.6SD increase in the lottery choice. While this is suggestive that one mechanism through which domestic violence affects children is by diminishing the effect of mothers' time investments, other combinations of these two indicators are not rejected by the data either.

This analysis suggests that identifying the effect of domestic violence separate from the complexity of parental investments is challenging. This would amount to finding instrumental variables that affect domestic violence but do not affect any of the instances of parental investment. The next section addresses this identification problem by estimating a structural model in which parental investments of different types and domestic violence are endogenous. The structural approach provides evidence that both mothers' investments of time and money and domestic violence are both consequential for the development of a child. While structural estimates depend heavily on function form assumptions, they provide support for the idea of adjusting measures of parental investments by the presence of domestic violence.

²¹I discuss this approach in Appendix B

7 Structural Estimates

In this section, I discuss the estimation of a structural household model in which parental investments and domestic violence are endogenous. As mentioned in Section 3, I assume a Cobb-Douglas utility function and Cobb-Douglas production functions for simplicity. I also assume that there is a 'market' for violence. This assumption aims to represent the existence of social costs associated with exercising 'violence' on others (e.g. lost friendships and purchase of presents to repair relationships). Finally, it is clear from the model that if domestic violence has a negative effect on the development of the child, parents will never engage in violence if parameter $|\alpha_V|$, the utility from domestic violence, is smaller than $|\alpha_{\theta}a_V + \alpha_{\sigma}b_V|$, the utility-weighted loss in child development. In the estimation, I assume that the utility parameter for violence has a lower bound. This is equivalent to assuming that parents will refrain from violence if it is too harmful, but there is always a remnant demand for 'violence' that is unavoidable.

Preference parameters α_i are assumed to be positive and add up to one. To model heterogeneity in preferences, and therefore heterogeneity in behavior, it is assumed that they are drawn from a multivariate distribution.²² The log of wages are assumed to depend on years of education and age. Predetermined income is assumed to be random, and bounded from below at zero, which is largely consistent with the data. The price of violence p_V is assumed to depend on the education of the parents. All income related processes are allowed to be correlated. The parameters of the cognitive development and risk preferences equation are assumed to be fixed and equal for all households. In this model, parental investments and domestic violence are endogenous, and their level depends on how much parents care about their child's cognitive ability and risk attitudes. The model allows for the presence of domestic violence to be negatively correlated with parental investments due to preferences and not due to their impact on child development or preferences. Ex-ante, the model does not assume that parents care about their child's preferences or that domestic violence harms their development.

The model is estimated using the simulated method of moments. For a given set of parameters, 50 random draws are obtained for each household, decisions are calculated according to the value of these draws and moments of the simulated data are then calculated. Moments are weighted by the inverse of the estimated variance of the sample moments. The list of moments used in the estimation and the fit of the different moments are presented in Appendix D. The model is estimated in the subset of families with two parents present to

²²I follow Del Boca et al. (2014) in specifying preference parameters. In particular, I assume that $\alpha_i = \frac{exp(\kappa_i)}{\sum_j exp(\kappa_j)}$, and $\kappa \sim N(\mu, \Sigma)$

simplify the analysis.²³ Domestic violence is more likely to occur in these households (18% ever reporting domestic violence versus 11%) and most households are couples (87%). All the flow variables are defined in weekly terms.

Estimation of the household model requires observing maternal investments of time. Due to data limitations, this variable is proxied by the addition of the time a child is reported doing homework and the time that it takes to get the child to school. Both these variables are taken from the second wave of the study. These variables are used as a proxy of maternal time investments because the study time of a five year old child is mostly supervised by mothers and mothers are likely to take the child to school.²⁴ To improve identification of maternal time investments, the model is augmented by making additional measures of maternal time investments depend on this proxy variable. In particular, the responses to the questions 'Do you help your child with homework?', 'Do you know the names of your child's friends?' and 'Do you encourage your child to read?' are modelled as a function of maternal time investment. This approach mimics the construction of the parental investment index in the Section 6.4.

Utility is defined by the amount of domestic violence consumed. This consumption is assumed to be a latent variable that manifests itself in different measures, such as reporting incidents of domestic violence or asking for help due to domestic violence. This is, indeed, the assumption already made in the paper through the construction of an indicator of domestic violence from different variables. The difference is that the parameters of the factor model are estimated simultaneously with the rest of the household parameters. Additional advantages of this exercise are that these indicators are explicitly modelled as dichotomous variables and that discrete and continuous variables can be handled using single index models. Because domestic violence is a latent variable, the mean price of violence p_V is assumed to be one. This provides a scale for the domestic violence latent factor and allows for identification of the preference parameter for violence. Finally, the structural model allows lottery choices to be modelled as an ordered probit which is a function of latent risk preferences. That is, the structural model provides a consistent framework to accommodate continuous and discrete measures.

Table E.2 in Appendix D shows the fit of the model for the 120 moments used to calibrate it. The model is able to reproduce many features of the data and the variables used in the analysis. For instance, the model captures the prevalence of domestic violence and behavior in the lottery experiment as well as the cognitive development of the child. Due to the presence of outliers in the earning variables, these variables follow the empirical

 $^{^{23}\}mathrm{A}$ complete model would require modelling the decision to form a two-parent household.

 $^{^{24}\}mathrm{This}$ is confirmed in the data.

moments less closely. This suggests that using quantiles of these variables might produce more robust estimates. Figure 3 compares the simulated and observed distributions of the cognitive development index. The model follows the main patterns of this variable closely.

Table 7 presents estimates of preference parameters and the cognitive development and risk attitudes functions. Parents seem to care about both the cognitive development and the risk attitudes of their children. The sum of the weight of these two elements exceeds that of consumption. The estimates also suggest that there is a high degree of preference heterogeneity as expressed by the coefficient of variation of the preference parameters. Regarding cognitive development, the model estimates suggest that both parental investment of time and money are important inputs. The negative effect of domestic violence on cognitive development at age 8 is estimated to be small. Regarding the risk attitudes of children, the estimates confirms that domestic violence has a negative effect on risk taking behavior. Parental investment seems to have a relatively small effect on risk attitudes. These results should be taken with caution since they depend on the assumptions made on preferences. technology and household dynamics. It is possible that domestic violence affects cognitive development at earlier ages when the role of schooling is less important and because parental investment might affect preferences at earlier ages as well. To identify these possibilities, the model would have to be extended to earlier periods. This would be challenging, however, due to data limitations.

The structural model exercise confirms that domestic violence is not capturing parental investments only. The model could be extended in several directions. One direction is to allow complementarities in the production function. A second direction is to look at the dynamics of domestic violence and development using all waves of the Young Lives study. That might afford identification of the effect of household environment at earlier and later years. The model can also be extended to include other measures of child development like physical health and self-assessments.

To test the robustness of the results, Appendix E discusses an alternative model of household decisions which is hierarchical. The model assumes that *altruistic* husbands might use violence as a way to extract effort from their wives whenever they cannot afford nonviolent incentives. Being altruistic, husbands consider the trade-off of extracting labor and decreasing the wellbeing of their wives and their offsprings. Estimates of this model fit the data less well, but they confirm that domestic violence is deleterious to child development.

In sum, the structural estimation exercise supports the idea that parents care about their child's preferences as they do about their cognitive development. A potential reason for this is that risk attitudes are associated with other aspects of a child's development that are not captured by cognitive measures.

8 Conclusions

Using a longitudinal study of a random sample of Peruvian children, I investigate the relationship between the presence of domestic violence in the household in early childhood and the cognitive development and risk attitudes of children. My results confirm previous research that a child's cognitive development is negatively associated with domestic violence, and I provide evidence that this effect is causal. I also find that the presence of domestic violence in a household early in a child's life is associated with the manifestation of more risk averse behavior. While domestic violence affects cognitive development and risk attitudes separately, they are not correlated. That is, bad early life experiences, such as living in a household with domestic violence, can have multiple and separate effects. Structural estimation of a household model with domestic violence reveals that parents behave as if they care about not only their child's cognitive development but also her preferences. Parents seem to favor risk taking behavior during childhood.

Some research shows that exposure to domestic violence during childhood increases the likelihood of committing crimes and engaging in risky activities as an adult (Carrell and Hoekstra, 2010; Currie and Tekin, 2006). I find instead that children who grew up in a household experiencing domestic violence are more risk averse. This apparent contradiction might be due to the fact that children exposed to domestic violence are also more likely to have higher costs to human capital accumulation, through its effect on cognitive abilities and household dynamics, and this could drive crime and risky behavior as an adult (Freeman, 1999). The findings of this paper combined with previous results highlight the complementarity of experimental and survey methods in the identification of why past experiences affect future behavior.

Recent evidence shows that interventions during infancy can produce behavioral changes later in life (Gertler et al., 2013; Heckman et al., 2013). This paper shows that individual preferences themselves might be altered as well. Policies aimed at improving a child's household conditions might be a necessary complement for the success of interventions at the school level.

References

- A. Aizer and F. Cunha. The production function of human capital: endowments, investments and fertility. NBER WP 18429, sep 2012.
- Anna Aizer. The gender wage gap and domestic violence. American Economic Review, 100 (4):1847–1859, September 2010.
- Christopher T. Andersen, Sarah A. Reynolds, Jere R. Behrman, Benjamin T. Crookston, Kirk A. Dearden, Javier Escobal, Subha Mani, Alan Sanchez, Aryeh D. Stein, and Lia C. H. Femald. Participation in the Juntos Conditional Cash Transfer Program in Peru Is Associated with Changes in Child Anthropometric Status but Not Language Development or School Achievement. *Journal of Nutrition*, 145(10):2396–2405, OCT 2015. ISSN 0022-3166. doi: 10.3945/jn.115.213546.
- Orazio Attanasio, Costas Meghir, and Emily Nix. Human capital development and parental investment in india. NBER Working Paper 21740, Nov 2015.
- Megan H. Bair-Merritt, Sara B. Johnson, Sande Okelo, and Gayle Page. Intimate partner violence exposure, salivary cortisol, and childhood asthma. *Child Abuse & Neglect*, 36 (7-8):596–601, Jul-Aug 2012.
- Marianne Bertrand and Jessica Pan. The trouble with boys: social influences and the gender gap in disruptive behavior. *American Economic Journal-Applied Economics*, 5(1):32–64, January 2013.
- Eric Bettinger and Robert Slonim. Patience among children. *Journal of Public Economics*, 91(1-2):343–363, February 2007.
- H. P. Binswanger. Attitudes toward risk experimental-measurement in rural india. American Journal of Agricultural Economics, 62(3):395–407, 1980.
- David W. Brown, Robert F. Anda, Henning Tiemeier, Vincent J. Felitti, Valerie J. Edwards, Janet B. Croft, and Wayne H. Giles. Adverse childhood experiences and the risk of premature mortality. *American Journal of Preventive Medicine*, 37(5):389–396, November 2009.
- Stephen V. Burks, Jeffrey P. Carpenter, Lorenz Goette, and Aldo Rustichini. Cognitive skills affect economic preferences, strategic behavior, and job attachment. *Proceedings of* the National Academy of Sciences of the United States of America, 106(19):7745–7750, May 12 2009.

- C. Caldji, B. Tannenbaum, S. Sharma, D. Francis, P. M. Plotsky, and M. J. Meaney. Maternal care during infancy regulates the development of neural systems mediating the expression of fearfulness in the rat. *Proceedings of the National Academy of Sciences of the United States of America*, 95(9):5335–5340, April 28 1998.
- Michael Callen, Mohammad Isaqzadeh, James D. Long, and Charles Sprenger. Violence and Risk Preference: Experimental Evidence from Afghanistan. American Economic Review, 104(1):123–148, Jan 2014.
- J. M. Campbell, S. K. Bell, and L. K. Keith. Concurrent validity of the peabody picture vocabulary test-third edition as an intelligence and achievement screener for low ses african american children. Assessment, 8(1):85–94, March 2001.
- Juan-Camilo Cardenas, Anna Dreber, Emma von Essen, and Eva Ranehill. Gender differences in competitiveness and risk taking: comparing children in colombia and sweden. Journal of Economic Behavior & Organization, 83(1, si):11–23, June 2012.
- Scott E. Carrell and Mark L. Hoekstra. Externalities in the Classroom: How Children Exposed to Domestic Violence Affect Everyone's Kids. American Economic Journal -Applied Economics, 2(1):211–228, January 2010.
- Marco Castillo, Jeffrey L. Jordan, and Ragan Petrie. Children's rationality, risk attitudes and field behavior. *European Economic Review*, 102:62–81, 2018.
- Marco Castillo, Jeffrey Jordan, and Ragan Petrie. Discount rates of children and high school graduation. *The Economic Journal*, 129(619):1153–1181, 2019.
- Marco Castillo, Paul J. Ferraro, Jeffrey L. Jordan, and Ragan Petrie. The today and tomorrow of kids: time preferences and educational outcomes of children. *Journal of Public Economics*, 95(11-12):1377–1385, December 2011.
- Edith Chen, Sheldon Cohen, and Gregory E. Miller. How Low Socioeconomic Status Affects 2-Year Hormonal Trajectories in Children. *Psychological Science*, 21(1):31–37, Jan 2010.
- Victor Chernozhukov and Christian Hansen. The reduced form: A simple approach to inference with weak instruments. *Economics Letters*, 100(1):68–71, JUL 2008a. ISSN 0165-1765. doi: 10.1016/j.econlet.2007.11.012.
- Victor Chernozhukov and Christian Hansen. The reduced form: A simple approach to inference with weak instruments. *Economic Letters*, 100(1):68–71, Jul 2008b.

- K Chisholm, M. C. Carter, E. W. Ames, and S. J. Morison. Attachment security and indiscriminately friendly behavior in children adopted from romanian orphanages. *Development and Psychopathology*, 7(2):283–294, Spring 1995. ISSN 0954-5794.
- F. Cirulli, A. Berry, and E. Alleva. Early disruption of plasticity the mother-infant relationship: effects on brain and implications for psychopathology. *Neuroscience and Biobehavioral Reviews*, 27(1-2):73–82, jan-mar 2003.
- Santiago Cueto and J. Leon. Psychometric characteristics of cognitive development and achievement instruments in round 3 of young lives. Young lives technical note no. 25, December 2012.
- Santiago Cueto, J. Leon, G. Guerrero, and I. Munoz. Psychometric characteristics of cognitive development and achievement instruments in round 2 of young lives. Young lives technical note no. 15, January 2009.
- Flavio Cunha, James J. Heckman, and Susanne M. Schennach. Estimating the Technology of Cognitive and Noncognitive Skill Formation. *Econometrica*, 78(3):883–931, May 2010.
- Janet Currie and Erdal Tekin. Does child abuse cause crime?, 2006.
- Gordon B. Dahl and Lance Lochner. The Impact of Family Income on Child Achievement: Evidence from the Earned Income Tax Credit. American Economic Review, 102(5):1927– 1956, Aug 2012.
- Andrea Danese and Bruce S. McEwen. Adverse childhood experiences, allostasis, allostatic load, and age-related disease. *Physiology & Behavior*, 106(1, si):29–39, April 12 2012.
- Andrea Danese, Carmine M. Pariante, Avshalom Caspi, Alan Taylor, and Richie Poulton. Childhood maltreatment predicts adult inflammation in a life-course study. *Proceedings* of the National Academy of Sciences of the United States of America, 104(4):1319–1324, January 23 2007.
- Richard J. Davidson and Bruce S. McEwen. Social influences on neuroplasticity: stress and interventions to promote well-being. *Nature Neuroscience*, 15(5):689–695, May 2012.
- Daniela Del Boca, Christopher Flinn, and Matthew Wiswall. Household Choices and Child Development. Review of Economic Studies, 81(1):137–185, Jan 2014.
- Thomas Dohmen, Armin Falk, David Huffman, and Uwe Sunde. Are risk aversion and impatience related to cognitive ability? *American Economic Review*, 100(3):1238–1260, June 2010.

- Catherine C. Eckel and Philip J. Grossman. Forecasting risk attitudes: An experimental study using actual and forecast gamble choices. *Journal of Economic Behavior & OrganizationN*, 68(1):1–17, OCT 2008. ISSN 0167-2681. doi: 10.1016/j.jebo.2008.04.006.
- Catherine C. Eckel, Mahmoud A. el Gamal, and Rick K. Wilson. Risk loving after the storm: a bayesian-network study of hurricane katrina evacuees. *Journal of Economic Behavior & Organization*, 69(2):110–124, February 2009.
- M. Ellsberg, I. Heise, R. Pena, S Agurto, and A. Winkvist. Researching domestic violence against women: methodological and ethical considerations. *Studies in Family Planning*, 32(1):1–16, March 2001.
- Mary Ellsberg, Diana J Arango, Matthew Morton, Floriza Gennari, Sveinung Kiplesund, Manuel Contreras, and Charlotte Watts. Prevention of violence against women and girls: what does the evidence say? *The Lancet*, 14:61703–61707, Nov 2014.
- Clifton R. Emery. Controlling for selection effects in the relationship between child behavior problems and exposure to intimate partner violence. *Journal of Interpersonal Violence*, 26(8):1541–1558, May 2011.
- Javier Escobal and Eva Flores. An assessment of the. young lives sampling. approach in peru. Young Lives Technical note no. 3, march 2008.
- J. W. Fantuzzo and W. K. Mohr. Prevalence and effects of child exposure to domestic violence. *Future of Children*, 9(3):21–32, Winter 1999.
- V. J. Felitti, R. F. Anda, D. Nordenberg, D. F. Williamson, A. M. Spitz, V. Edwards, M. P. Koss, and J. S. Marks. Relationship of childhood abuse and household dysfunction to many of the leading causes of death in adults the adverse childhood experiences (ace) study. American Journal of Preventive Medicine, 14(4):245–258, May 1998.
- D. Filmer and L. Pritchett. The effect of household wealth on educational attainment: evidence from 35 countries. *Population and Development Review*, 25(1):85+, March 1999.
- D. Filmer and L. Pritchett. Estimating wealth effects without expenditure data or tears: an application to educational enrollments in states of india. *Demography*, 38(1):115–132, February 2001.
- FONCODES. Mapa de la pobreza 2000. Foncodes, 2001.
- D. D. Francis, J. Diorio, P. M. Plotsky, and M. J. Meaney. Environmental enrichment reverses the effects of maternal separation on stress reactivity. *Journal of Neuroscience*, 22(18):7840–7843, Sep 15 2002. ISSN 0270-6474.

- Richard B. Freeman. The Economics of Crime. Handbooks in Economics, vol. 5. Amsterdam; New York and Oxford: Elsevier Science, North-Holland, 1999.
- Paul Gertler, James Heckman, Rodrigo Pinto, Arianna Zanolini, Christel Vermeersch, Susan Walker, Susan M. Chang, and Sally Grantham-Zmcgregor. Labor market returns to early childhood stimulation: a 20-year followup to an experimental intervention in jamaica. National Bureau of Economic Research w19185, 2013.
- U. Gneezy and A. Rustichini. Gender and competition at a young age. American Economic Review, 94(2):377–381, May 2004.
- Bart H. H. Golsteyn, Hans Gronqvist, and Lena Lindahl. Adolescent Time Preferences Predict Lifetime Outcomes. *Economic Journal*, 124(580, F):F739–F761, Nov 2014.
- J Gruber and E Saez. The elasticity of taxable income: evidence and implications. *Journal* of *Public Economics*, 84(1):1–32, Apr 2002.
- Megan R. Gunnar and Karina M. Quevedo. Early care experiences and HPA axis regulation in children: a mechanism for later trauma vulnerability. In Dekloet, E. R. and Vermetten, E., editor, Stress hormones and post traumatic stress disorder: basic studies and clinical perspectives, volume 167 of Progress in brain research, pages 137+, 2007.
- William T. Harbaugh, Kate Krause, and Lise Vesterlund. Risk attitudes of children and adults: choices over small and large probability gains and losses. *Experimental Economics*, 5(1):53–84, June 2002.
- Glenn W. Harrison, John A. List, and Charles Towe. Naturally occurring preferences and exogenous laboratory experiments: A case study of risk aversion. *Econometrica*, 75(2): 433–458, Mar 2007.
- Johannes Haushofer and Ernst Fehr. On the psychology of poverty. *Science*, 344(6186): 862–867, May 23 2014.
- James Heckman, Rodrigo Pinto, and Peter Savelyev. Understanding the Mechanisms Through Which an Influential Early Childhood Program Boosted Adult Outcomes. American Economic Review, 103(6):2052–2086, Oct 2013.
- James J. Heckman, Jora Stixrud, and Sergio Urzua. The effects of cognitive and noncognitive abilities on labor market outcomes and social behavior. *Journal of Labor Economics*, 24(3):411–482, July 2006.

- L. W. Hedin and P. O. Janson. Domestic violence during pregnancy the prevalence of physical injuries, substance use, abortions and miscarriages. Acta Obstetricia et Gynecologica Scandinavica, 79(8):625–630, August 2000.
- John L. Horn. A rationale and test for the number of factors in factor analysis. *Psychometrika*, 30(2):179–185, 1965.
- P. R. Huttenlocher. Synaptic density in human frontal cortex developmental changes and effects of aging. *Brain Research*, 163(2):195–206, 1979.
- Narayanan Kandasamy, Ben Hardy, Lionel Page, Markus Schaffner, Johann Graggaber, Andrew S. Powlson, Paul C. Fletcher, Mark Gurnell, and John Coates. Cortisol shifts financial risk preferences. *Proceedings of the National Academy of Sciences of the United States of America*, 111(9):3608–3613, Mar 4 2014.
- M. A. Kernic, M. E Wolf, V. L. Holt, B. McKnight, C. E. Huebner, and F. P. Rivara. Behavioral problems among children whose mothers are abused by an intimate partner. *Child Abuse & Neglect*, 27(11):1231–1246, November 2003.
- K. C. Koenen, T. E. Moffitt, A. Caspi, A. Taylor, and S. Purcell. Domestic violence is associated with environmental suppression of IQ in young children. *Development and Psychopathology*, 15(2):297–311, Spring 2003.
- SM Korte. Corticosteroids in relation to fear, anxiety and psychopathology. Neuroscience and Biobehavioral Reviews, 25(2):117–142, Mar 2001.
- I. P. Levin and S. S. Hart. Risk preferences in young children: early evidence of individual differences in reaction to potential gains and losses. *Journal of Behavioral Decision Making*, 16(5):397–413, December 2003.
- Ulrike Malmendier and Stefan Nagel. Depression babies: do macroeconomic experiences affect risk taking? *Quarterly Journal of Economics*, 126(1):373–416, February 2011.
- J. McFarlane, B. Parker, and K. Soeken. Abuse during pregnancy: associations with maternal health and infant birth weight. *Nursing Research*, 45(1):37–42, jan-feb 1996.
- M. J. Meaney. Maternal care, gene expression, and the transmission of individual differences in stress reactivity across generations. Annual Review of Neuroscience, 24:1161–1192, 2001.

- Michael J. Meaney, David H. Aitken, Shari R. Bodnoff, Linda J. Iny, Joseph E. Tatarewicz, and Robert M. Sapolsky. Early Postnatal Handling Alters Glucocorticoid Receptor Concentrations in Selected Brain Regions. *Behavioral Neuroscience*, 127(5):637–641, Oct 2013.
- Bruno Moreira, Raul Matsushita, and Sergio da Silva. Risk seeking behavior of preschool children in a gambling task. *Journal of Economic Psychology*, 31(5, si):794–801, October 2010.
- L. Murray and P. J. Cooper. Postpartum depression and child development. *Psychological Medicine*, 27(2):253–260, March 1997.
- Charles A. Nelson, III, Charles H. Zeanah, Nathan A. Fox, Peter J. Marshall, Anna T. Smyke, and Donald Guthrie. Cognitive recovery in socially deprived young children: the bucharest early intervention project. *Science*, 318(5858):1937–1940, December 21 2007.
- J. D. Osofsky. The impact of violence on children. *Future of Children*, 9(3):33–49, Winter 1999.
- Isabelle Ouellet-Morin, Andrea Danese, Lucy Bowes, Sania Shakoor, Antony Ambler, Carmine M. Pariante, Andrew S. Papadopoulos, Avshalom Caspi, Terrie E. Moffitt, and Louise Arseneault. A Discordant Monozygotic Twin Design Shows Blunted Cortisol Reactivity Among Bullied Children. Journal of the American Academy of Child and Adolescent Psychiatry, 50(6):574–582, JUN 2011.
- Ingo Outes-Leon and Stefan Dercon. Survey attrition and attrition bias in young lives. Young Lives Technical note no. 5, march 2008.
- Paul M. Pronyk, James R. Hargreaves, Julia C. Kim, Linda A. Morison, Godfrey Phetla, Charlotte Watts, Joanna Busza, and John D. H. Porter. Effect of a structural intervention for the prevention of intimate-partner violence and HIV in rural South Africa: a cluster randomised trial. *Lancet*, 368(9551):1973–1983, Dec 2006.
- A Schlottmann. Children's probability intuitions: understanding the expected value of complex gambles. *child development*, 72(1):103–122, January-February 2001.
- Carmit Segal. Misbehavior, education, and labor market outcomes. Journal of the European Economic Association, 11(4):743–779, August 2013.
- Matthias Sutter, Martin G. Kocher, Daniela Glaetzle-Ruetzler, and Stefan T. Trautmann. Impatience and uncertainty: experimental decisions predict adolescents' field behavior. *American Economic Review*, 103(1):510–531, 2013.

- R. A. Thompson and C. A. Nelson. Developmental science and the media early brain development. American Psychologist, 56(1):5–15, January 2001.
- M. Tomlinson, P. Cooper, and L. Murray. The mother-infant relationship and infant attachment in a South African peri-urban settlement. *Child Development*, 76(5):1044–1054, September-October 2005.
- Iva Trako, Micaela Sviatschi, and Guadalupe Kavanaugh. Inter-generational benefits of an integral approach to reduce gender-based violence: Evidence from peru. working paper, January 15 2018.
- UNICEF. Behind closed doors. the impact of domestic violence on children., 2006.
- Maarten J. Voors, Eleonora E. M. Nillesen, Philip Verwimp, Erwin H. Bulte, Robert Lensink, and Daan P. van Soest. Violent conflict and behavior: a field experiment in burundi. american economic review, 102(2):941–964, April 2012.
- Bruce A. Weinberg. An Incentive Model of the Effect of Parental Income on Children. Journal of Political Economy, 109(2):266–280, April 2001.
- Joshua A. Weller, Irwin P. Levin, and Natalie L. Denburg. Trajectory of risky decision making for potential gains and losses from ages 5 to 85. *Journal of Behavioral Decision Making*, 24(4):331–344, Oct 2011.
- WHO. A user's guide to self-reporting questionnaire (SRQ). World Health Organization, 1994.
- WHO. Global and regional estimates of violence against women: prevalence and health effects of intimate partner violence and non-partner sexual violence. World Health Organization, 2013.



Figure 1: Lottery Instrument



Figure 2: Distribution of lottery choices





(b) Domestic violence (3rd survey)

	Ν	Mean	S.D.
Wealth index (3rd wave)	1919	0.54	0.21
Household income (3rd wave)	1919	11126.2	21728.9
Female average wage (2009)	1919	560.5	80.2
Male average wage (2009)	1919	748.5	130.6
Household size	1919	5.7	2.3
Two-parent household	1919	86.9	33.8
Lottery choice $(1=\text{safest}, 6=\text{riskiest})$	1919	4.6	1.8
Child's BMI (3rd wave)	1915	16.9	2.3
Child's BMI (4th wave)	1843	19.6	3.3
Hours in a typical day sitting	1897	4.2	0.8
Days being active at least 60 minutes (3rd wave)	1907	3.6	2.7
Hours a day playing (3rd wave)	1913	4.2	1.7
Hours a day studying (3rd wave)	1912	1.9	0.8
Hours a day playing (4th wave)	1859	3.7	1.4
Hours a day studying (4th wave)	1859	1.8	0.9

Table 1: Descriptive Statistics of Covariates and Outcomes

Note: 1st wave in 2002 was completed when the child was between 6 and 18 months old. Second wave was in 2006, 3rd wave in 2009, 4th in 2012.

	Ν	Mean	S.D.
Cognitive development at age 5			
Peabody test (2nd wave)	1845	29.2	17.9
Cognitive test (2nd wave)	1888	8.4	2.1
Cognitive development at age 8			
Peabody test (3rd wave)	1836	46.8	13.5
Math test (3rd wave)	1882	11.9	4.9
Reading & writing test (3rd wave)	1739	7.7	3.0
Cognitive development at age 12			
Peabody test (4th wave)	1875	85.6	17.5
Math test (4th wave)	1871	16.1	5.5
Reading & writing test (4th wave)	1871	14.4	3.6
PARENTAL INVESTMENT			
Expenditure on child (3rd wave)	1912	553.5	1365.2
Value of gifts to child (3rd wave)	1912	342.7	497.7
Number of books at home	1915	1.7	1.3
Helps child with homework $(\%)$	1919	61.7	48.6
Encourages child to read $(\%)$	1915	2.5	0.6
Knows the name of child's friends $(\%)$	1912	84.1	36.6
Reason with child when misbehaves $(\%)$	1913	45.1	32.4
Mother's capital			
Mother's school level (category)	1909	31.3	9.1
Spanish speaker $(\%)$	1919	85.2	35.5
Mother's ability to read in first language			
(0=not at all, $1=$ w/diffic., $3=$ w/ease)	1919	1.6	0.7
DOMESTIC VIOLENCE			
Reported domestic violence in 3rd wave $(\%)$	1919	7.0	25.6
Reported domestic violence in 2nd wave $(\%)$	1919	8.2	27.4
Ask for help due to Dom. Viol. (3rd wave) $(\%)$	1919	12.5	33.0
Number of smokers (excl. mother)	1905	0.1	0.4

Table 2: Descriptive Statistics of Measure Variables for 8 year old sample

Table 3: MEASUREMENT SYSTEM

PROPORTION	OF	TH	Ε	VARI-	
ANCE EXPLAIN	NED	BY	\mathbf{L}	ATENT	1
VARIABLE					

Gognitive ability at 12 (4th round)					
Peabody test (4th survey)	0.478				
Math knowledge exam	0.549				
Early grade reading assessment	0.645				
Gognitive ability at 8 (3rd round)					
Peabody test (3rd survey)	0.534				
Math knowledge exam	0.556				
Early grade reading assessment	0.645				
Parental investments					
Monetary expenses on child	0.230				
Gifts	0.288				
No. books at home	0.124				
Helps child w/homework	0.007				
Encourages child to read	0.006				
No. names of child's friends known	0.013				
Reasons with child when misbehaves	0.009				
Domestic violence/Household environment					
Reported DV in 3rd survey	0.386				
Reported DV in 2nd survey	0.108				
Searched for help (3rd survey)	0.044				
No. people who smoke in the house	0.094				
Gognitive ability at 5 (2nd round)					
Peabody test (2nd survey)	0.238				
Cognitive test	0.867				
Mother's capital					
Years of schooling	0436				
Spanish speaker	0.388				
Mother can read in first language learned	0.684				

Note: Measures control for age, location and sex of child at the time of data collection. All variables are standardized.

			Table 4:	Correla	tion table								
Variables	у	y_2	y3	uecision y4	y_5	y_6	Wave 1	Wave 2	yen. mataise Wave 3	Wave 4	Wave 1	Wave 2	Wave 3
y	1.000												
y2	0.629^{***}	1.000											
<i>y</i> 3	0.847^{***}	0.628^{***}	1.000										
	(0.000)	(0.000) 0.512***	0.816***	1 000									
ずみ 	(0.000)	(0.000)	(0.000)	***	000								
<i>y</i> 5	0.000)	(0.000)	(0000)	(0.000)	ппп-т								
ye	0.833***	0.323^{***}	0.515^{***}	0.631^{***}	0.821^{***}	1.000							
Mother's psych. malaise index (wave 1)	0.015	-0.003	-0.016	0.019	0.028	0.024	1.000						
Mother's psych. malaise index (wave 2)	-0.025	-0.041	-0.014	-0.012	(T / 7.0)	-0.026	0.288^{***}	1.000					
Mother's psych. malaise index (wave 3)	(0.336) 0.004	(0.111) -0.024	(0.599) -0.006	(0.629) 0.017	(0.508) 0.006	(0.309) 0.012	(0.000) 0.169^{***}	0.166^{***}	1.000				
Mother's psych. malaise index (wave 4)	$(0.873) \\ 0.011$	(0.347) 0.021	$(0.825) \\ 0.014$	(0.506) 0.008	(0.821) 0.013	(0.643) -0.004	(0.000) 0.346^{***}	(0.000) 0.257^{***}	0.224^{***}	1.000			
Serious injury (wave 1)	(0.672) - 0.047^{*}	(0.418) -0.042*	(0.580) -0.039	(0.764) -0.026	(0.626) -0.030	$(0.891) - 0.057^{**}$	(0.000) 0.153^{***}	(0.000)	(0.000) 0.058^{**}	0.054^{**}	1.000		
Serious injury (wave 2)	(0.070) 0.014	(0.099) - 0.014	(0.129) 0.001	(0.304) 0.026	(0.243) 0.032	(0.026) 0.003	(0.000) 0.052^{**}	$(0.134) \\ 0.064^{**}$	(0.024) 0.059^{**}	(0.034) 0.034	0.067^{***}	1.000	
Serious injury (wave 3)	(0.576) 0.015	(0.589) -0.028	(0.957) 0.000	(0.305) 0.013	(0.216) 0.027	(0.894) 0.030	(0.043) 0.036	(0.013) 0.036	$(0.022) \\ 0.045 *$	$(0.188) \\ 0.051^{**}$	(0.009) -0.003	0.076***	1.000
Female	$(0.565) - 0.063^{**}$	(0.280) -0.013	(0.996) -0.031	$(0.606) -0.062^{**}$	(0.290) -0.067***	(0.242) - 0.072^{***}	(0.161) -0.016	(0.162) 0.030	(0.080) -0.009	(0.046) -0.023	(0.914) -0.004	(0.003) -0.094***	-0.102^{***}
Domestic violence renorted (wave 1)	(0.014)	(0.625)	(0.236)	(0.017)	(0.009) -0.016	(0.005)	(0.546) 0 143 * * *	(0.238)	(0.714)	(0.365) 0.111***	(0.883)	(0.000)	(0.000) -0.020
	(0.516)	(0.988)	(0.316)	(0.846)	(0.523)	(0.482)	(0000)	(0.020)	(0.019)	(0.00)	(0.839)	(0.349)	(0.435)
Domestic violence reported (wave 2)	-0.044 [*] (0.088)	-0.079^{***} (0.002)	-0.037 (0.155)	-0.028 (0.271)	-0.031 (0.230)	-0.026 (0.309)	0.029 (0.258)	0.100^{***} (0.000)	0.049° (0.056)	0.073^{***} (0.004)	0.007 (0.779)	-0.005 (0.849)	-0.014 (0.577)
Domestic violence reported (wave 3)	-0.059**	-0.060**	-0.039	-0.028	-0.046*	-0.075*** (0.004)	0.079*** (0.009)	0.082^{***}	0.098***	0.122^{***}	0.050* (0.052)	0.013	0.033
Child cognitive development (wave 3)	-0.035	0.009	-0.019	-0.042	-0.046*	-0.033	-0.039	-0.053**	-0.073***	-0.099***	0.019	0.003	0.032
Child cognitive development (wave 2)	-0.018	0.034	-0.005	(101.0)	-0.025	-0.045 *	0.002	-0.079***	0.004	0.002	(0.470) (0.020)	0.029	-0.016
Mother's capital	(0.496) 0.019	$(0.191) \\ 0.050^{*}$	(0.833) 0.030	(0.751) -0.005	(0.326) 0.013	(0.080) 0.006	$(0.945) \\ -0.104^{***}$	(0.002) -0.016	$(0.986) -0.057^{**}$	$(0.923) \\ -0.087^{***}$	(0.445) -0.011	(0.252) -0.011	(0.528) 0.016
Reasons with child when misbehaves (wave 2)	$(0.471) \\ -0.074^{***}$	(0.052) 0.003	$(0.249) \\ -0.056^{**}$	(0.833)-0.099***	$(0.603) -0.066^{**}$	$(0.826) -0.066^{**}$	(0.000) -0.012	(0.547) -0.017	(0.028) -0.041	(0.001) -0.028	(0.670) 0.008	(0.659) 0.009	(0.533) 0.009
Reasons with child when misbehaves (wave 3)	$(0.004) - 0.056^{**}$	(0.896) -0.015	(0.031) - 0.041	$(0.000) - 0.065^{**}$	$(0.010) -0.054^{**}$	$(0.010) - 0.048^{*}$	(0.652) -0.006	(0.504) 0.026	$(0.108) \\ -0.063^{**}$	(0.284) -0.015	(0.765) 0.013	(0.729) -0.008	(0.725) 0.016
Corporal punishment if child misbehaves (wave 2)	$(0.029) \\ 0.086^{***}$	(0.560) 0.033	$(0.109) \\ 0.064^{**}$	$(0.012) \\ 0.084^{***}$	(0.035) 0.080^{***}	$(0.062) \\ 0.084^{***}$	$(0.829) \\ 0.052^{**}$	(0.320) 0.031	$(0.014) \\ 0.062^{**}$	(0.558) 0.025	(0.621) 0.023	(0.750) -0.031	(0.536) -0.006
Corporal punishment if child misbehaves (wave 2)	$(0.001) \\ 0.051^{**}$	(0.205) 0.002	(0.012) 0.042	(0.001) 0.041	(0.002) 0.048^{*}	$(0.001) \\ 0.062^{**}$	(0.044) 0.006	(0.227) 0.067^{**}	(0.017) 0.021	(0.335) 0.034	(0.363) 0.035	(0.232) 0.042	(0.804) 0.016
Excessive drinking (wave 1)	(0.049) -0.027	(0.944) -0.029	(0.103) -0.019	(0.113) -0.026	(0.060) -0.021	(0.016) -0.020	(0.826) 0.004	(0.010) 0.023	(0.419) -0.015	(0.193) 0.023	(0.174) -0.014	(0.106) 0.007	(0.527) 0.003
Excessive drinking (wave 2)	(0.293) -0.023	(0.263) - 0.031	(0.455) -0.012	(0.306) -0.010	(0.409) -0.018	(0.430) -0.027	(0.871) 0.026	(0.363) 0.097^{***}	(0.549) - 0.002	(0.374) 0.004	(0.591) 0.026	(0.791) -0.021	(0.910) 0.007
Excessive drinking (wave 3)	(0.375) -0.026	$(0.226) -0.046^{*}$	(0.630) 0.002	(0.698) -0.009	(0.490) -0.018	(0.290) -0.041	(0.321) 0.013	(0.000) 0.053^{**}	$(0.945) \\ 0.078^{***}$	$(0.883) \\ 0.061^{**}$	(0.314) 0.027	(0.411) 0.006	(0.798) 0.053^{**}
No of smokow in household (mana 2)	(0.319)	(0.073)	(0.946)	(0.729)	(0.494)	(0.110)	(0.621)	(0.040)	(0.002)	(0.017)	(0.292)	(0.813)	(0.038)
	(0.684)	(0.948)	(0.777)	(0.787)	(0.615)	(0.652)	(0.776)	(0.870)	(0.800)	(0.667)	(0.077)	(0.222)	(0.687)
No. of smokers in household (wave 3)	0.006 (0.803)	(0.897)	0.007 (0.783)	0.006 (0.801)	(0.772)	0.002 (0.932)	0.043 [*] (0.096)	(0.003)	(0.043°)	(0.725)	0.053^{**} (0.039)	(0.175)	(0.295)
Searched for help (wave 3)	-0.006	-0.009	-0.013	0.001	0.004	-0.011	0.091***	0.045*	0.160***	0.102***	0.107***	0.016	0.010
Wealth index (wave 3)	(one-n) ***770.0-	-0.002	-0.039	-0.051^{**}	-0.086***	-0.113^{***}	-0.046^{*}	-0.128^{***}	-0.060**	-0.071***	0.004	0.054^{**}	0.062^{**}
Mathau's monobalamian wanting. Na monome ala	(0.003)	(0.937)	(0.133)	(0.049)	(0.001)	(0.000) (0.000)	(0.071)	(0.000)	(0.019)	(0.006)	(0.876)	(0.036)	(0.016)
questionnaire for the first, second and fourth surveys.	For the this	rd survey mot	n survey wer ther's life sat	e conecteu. n isfaction is tl	nother s ueprive of the second s	ession/lack of	r based on 8	item question	naire of life sa	tisfaction. Th	he questions	were rescaled	to
measure lack of satisfaction. The WHO SQ20 was colfor the 1st. 2nd 3rd and 4th set of questions are 0.86	llected in the 05. 0.8467. (e 3rd survey, 0.7523, and 0.	but they are 8695. Child	not available iniuries: B	e to the publi urns. laceration	c. The Cronb ons. broken b	ach α scale r ones, etc. C c	eliability coef enitive deve	ficient of these slopment: Fau	e 4 measures ctor estimate	is 0.4702. T d from meas	he Cronbach	a's ive
development. No measures are available for the first s	survey. Dom	nestic violen	ce: Mother's	report of ine	sidence of dor	nestic violenc	e. No report	is available fo	r the 4th surv	ey. Lottery	decision: ¹	Variable y is	the
option chosen in the experiment: Option $1 = 5$ token: chance, Option $6 = 0$ or 20 with equal chance. Variab	s for sure, C ole $y \ge i$ me	Dption $2 = 4$ of ans that the a	or 10 with eq	ual chance, (or higher for	Dption $3 = 3$ $i \in \{2, 3, 4, 5$	or 13 with eq 6}.	ual chance, C	Dption $4 = 2$	or 16 with equ	al chance, O _l	ption $5 = 1$	or 19 with eq	ual

Table 5: Fixed-Effect regression of domestic violence on several outcomes							
	Mother's ps	sych. malaise	Ch	ild			
	All	First	had so	erious	Cog	nitive	
VARIABLES	data	3 surveys	inju	ıry	Devel	opment	
Domestic violence reported		0.076^{***}	0.002	-0.002	-0.164***	-0.153***	
		[0.023]	[0.020]	[0.021]	[0.056]	[0.057]	
		(0.001)	(0.929)	(0.917)	(0.004)	(0.008)	
Mother's psych. malaise				0.029^{**}		-0.105***	
				[0.015]		[0.037]	
				(0.046)		(0.004)	
HH per-capita consumption	-0.000	0.000	0.000	0.000	-0.000	-0.000	
	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]	
	(0.962)	(0.963)	(0.952)	(0.993)	(0.876)	(0.838)	
Wealth index	-0.167***	-0.242***	-0.136***	-0.135***	0.057	0.034	
	[0.032]	[0.046]	[0.041]	[0.042]	[0.129]	[0.131]	
	(0.000)	(0.000)	(0.001)	(0.001)	(0.658)	(0.798)	
Constant	0.098^{***}	0.128^{***}	0.168^{***}	0.167^{***}	0.009	0.023	
	[0.017]	[0.023]	[0.020]	[0.020]	[0.069]	[0.070]	
	(0.000)	(0.000)	(0.000)	(0.000)	(0.895)	(0.745)	
Observations	$7,\!489$	$5,\!669$	5,721	$5,\!669$	$3,\!533$	$3,\!495$	
R-squared	0.005	0.010	0.003	0.004	0.005	0.010	
Number of individuals	$1,\!919$	1,919	1,919	1,919	1,907	1,906	

s.e. in brackets, p-values in parentheses

*** p<0.01, ** p<0.05, * p<0.10

Mother's psych. malaise: No reports on domestic violence for the 4th survey were collected. Mother's depression/lack of life satisfaction are the predicted common factor based on WHO SQ20 depression questionnaire for the first, second and fourth surveys. For the third survey mother's life satisfaction is the predicted common factor based on 8 item questionnaire of life satisfaction. The questions were rescaled to measure lack of satisfaction. The WHO SQ20 was collected in the 3rd survey, but they are not available to the public. The Cronbach α scale reliability coefficient of these 4 measures is 0.4702. The Cronbach α 's for the 1st, 2nd, 3rd and 4th set of questions are 0.8605, 0.8467, 0.7523, and 0.8695. Child injuries: Burns, lacerations, broken bones, etc. Cognitive development: Factor estimated from measures of cognitive development. No measures are available for the first survey. **Domestic violence:** Mother's report of incidence of domestic violence. No report is available for the 4th survey.

	Cog	Dev. (wave	3)	L	ottery Choice	
	Parental	Dom Viol	Cog	Parental	Dom Viol	Lot
	Investment	Index	Develp.	Investment	Index	Choic
Simulated change in income	0 547***	0.194		0 526***	0.141	
Simulated change in income	[0.080]	0.124		[0.077]	$\begin{bmatrix} 0.141 \\ [0.121] \end{bmatrix}$	
	(0.000)	(0.125] (0.316)		(0.000)	(0.121] (0.246)	
Were more	0.049**	0.057**		0.020**	0.057**	
wage gap	[0.042]	-0.057		0.039	-0.057**	
	(0.017) (0.015)	(0.027] (0.033)		(0.010] (0.017)	(0.020] (0.027)	
Opening of Women Emergency Center	0.014	-0 078***		0.010	-0 062***	
opening of Women Emergency conter	[0.015]	[0.023]		[0,014]	[0 022]	
	(0.363)	(0.001)		(0.468)	(0.005)	
Parental investments			0.349**			0.218
	-		[0.172]	-		[0.28]
			(0.043)			(0.442
DV index			-0.339*			-0.733
			[0.196]			[0.345]
			(0.084)			(0.033)
Cog. development (w. 2)	0.118***	-0.054	0.383***	0.127***	-0.041	-0.02
	[0.029]	[0.045]	[0.043]	[0.026]	[0.041]	[0.067]
	(0.000)	(0.232)	(0.000)	(0.000)	(0.322)	(0.709)
Mother's capital	0.048**	0.045	0.230***	0.031	0.059^{*}	0.102*
	[0.023]	[0.036]	[0.030]	[0.019]	[0.030]	[0.04]
	(0.038)	(0.208)	(0.000)	(0.107)	(0.050)	(0.013)
Wealth index (w. 1)	-0.104***	0.034	0.155***	-0.098***	0.026	0.034
	[0.031]	[0.048]	[0.039]	[0.028]	[0.045]	[0.059]
	(0.001)	(0.480)	(0.000)	(0.001)	(0.556)	(0.567)
Family size	-0.028*	0.011	-0.028	-0.021	0.018	-0.00
	[0.015]	[0.023]	[0.019]	[0.014]	[0.021]	[0.029]
	(0.059)	(0.615)	(0.129)	(0.125)	(0.404)	(0.938)
Two-parent HH	0.029**	0.019	-0.013	0.027**	0.009	-0.01
	[0.015]	[0.023]	[0.019]	[0.014]	[0.021]	[0.029
	(0.047)	(0.396)	(0.505)	(0.046)	(0.670)	(0.713)
Income (w. 2)	0.638***	0.087	-0.012	0.627***	0.111	-0.07
	[0.065]	[0.100]	[0.053]	[0.062]	[0.097]	[0.087]
	(0.000)	(0.382)	(0.817)	(0.000)	(0.256)	(0.42]
Income sq. (w. 2)	-0.021	0.035	0.008	-0.021	0.030	0.072
	[0.036]	[0.056]	[0.046]	[0.035]	[0.055]	[0.075
	(0.564)	(0.536)	(0.861)	(0.546)	(0.586)	(0.340
No income reported (w. 2)	0.028	-0.003	-0.036	0.029	-0.003	0.103
	[0.029] (0.324)	[0.044] (0.939)	[0.035] (0.302)	[0.028] (0.287)	[0.044] (0.947)	0.057 (0.069
	(0.024)	(0.303)	(0.002)	(0.201)	(0.041)	(0.008
Proxy means test for CCT	-0.233***	0.120***	0.047	-0.224***	0.119***	0.196*
	[0.030]	[0.046]	[0.059]	[0.027]	[0.043]	[0.096
	(0.000)	(0.008)	(0.431)	(0.000)	(0.006)	(0.042
Observations	1.615	1.615	1.615	1.819	1.819	1.819

Table 6: FIRST AND SECOND STAGE IV REGRESSIONS FOR COGNITIVE DEVELOPMENT AND LOTTERY CHOICES

standard errors in brackets, p-values in parentheses *** p<0.01, ** p<0.05, * p<0.10

Figure 3: Model 1. Cumulative Distribution of Observed and predicted index of cognitive development



Table 7: MODEL 1:	STRUCTURAL	PARAMETERS
-------------------	------------	------------

Preference parameters							
Variable	coefficient	s.e.	CV				
α_C (Consumption)	0.224	0.0038	4.41				
α_m (Mother's leisure)	0.232	0.0011	4.51				
α_f (Father's leisure)	0.219	0.0034	7.33				
α_{θ} (Cognitive development)	0.174	0.0010	3.67				
α_{σ} (Risk attitudes)	0.126	0.0012	2.00				
α_V (Domestic violence)	0.026	0.0001					

Correlation between preferences parameters

	$lpha_C$	α_m	α_f	$lpha_{ heta}$
α_m	0.267			
$lpha_f$	-0.222	0.657		
$lpha_{ heta}$	0.193	0.675	0.731	
$lpha_{\sigma}$	0.716	0.329	0.122	0.385
Cognitive production	function			
Variable				
variable	coencient	s.e.		
a_{-} (Mother's investment (time))	0 114	0.0010		
a_{I_m} (Parental investment (money))	0.194	0.0010 0.0052		
a_{i} (Cognitive development (lagged))	1.337	0.0321		
a_{V_0} (Domestic Violence)	-0.020	0.0006		
a_{ℓ} (Mother's capital)	0.296	0.0051		
a_{b_m} (Household size)	-0.043	0.0026		
$a_{\mu\nu}$ (Wealth)	-0.030	0.0006		
- ()				
Risk attitudes fu	nction			
Variable	$\operatorname{coefficient}$	s.e.		
b_{τ_m} (Mother's investment (time))	0.005	0.0003		
b_I (Parental investment (money))	0.000	0.0000		
b_{θ_0} (Cognitive development (lagged))	0.023	0.0019		
b_V (Domestic Violence)	-0.174	0.0010		
b_{θ_m} (Mother's capital)	0.002	0.0001		
b_{hh} (Household size)	-0.153	0.0252		
b_{ω} (Wealth)	0.039	0.0014		

Standard errors are calculated as the standard deviation of parameter estimates of 200 bootstrap iterations.

A Additional Results

	(1)	(2)	(3)	(4)
VARIABLES	2004-2005	2004-2008	2004-2009	2004-2013
WEC in the district in 2007	0.055			
	[0.081]			
	(0.498)			
	0.045			
WEC in the district in 2008	-0.045			
	[0.092]			
	(0.624)			
WEC in the district in 2010	-0.046	-0.012		
	[0.092]	[0.026]		
	(0.618)	(0.636)		
	(01020)	(01000)		
WEC in the district in 2011	0.080	0.038	0.029	
	[0.089]	[0.026]	[0.018]	
	(0.370)	(0.137)	(0.108)	
WEC in the district in 2012	0.083	0.021	0.019	
	[0.084]	[0.034]	[0.023]	
	(0.325)	(0.535)	(0.406)	
WEC in the district in 2013	0 139	0.054	0.034	
	[0.097]	[0, 040]	[0, 031]	
	(0.157)	(0.181)	(0.267)	
	(01201)	(01101)	(0.201)	
WEC in the district in 2014	-0.002	0.080^{*}	0.043	0.014
	[0.096]	[0.047]	[0.027]	[0.012]
	(0.984)	(0.090)	(0.116)	(0.246)
Voor Fired offecta	Vac	Voc	Voc	Vec
rear rixed enects	res	res	res	res
Observations	101	268	383	839
R-squared	0.096	0.036	0.044	0.033
F p-value	0.100	0.199	0.221	0.246

 Table A.1: Relationship between WECs in the district and four windows of

 PRE-PROGRAM DOMESTIC VIOLENCE

Robust s.e. in brackets, p-values in parentheses *** p<0.01, ** p<0.05, * p<0.10

Notes: Clustering at the district level. The dependent variable is the change in domestic violence at the district level. Domestic violence is defined as experiencing physical or sexual violence. The observations correspond to the pre-program period of the WEC center rollout for each district. All districts that ever had a WEC center that opened between 2006-2014, 2009-2014, 2010-2014 and 2013-2014 are included. The data correspond to the Peruvian DHS.

	Tabl	le A.2: I	V REG	RESSION	IS OF TI	HE EFFE	CT OF]	DOMEST	fic Vio	LENCE	ON VAR	TOUS C	UTCON	IES				
	Cog. Devlop.	$\begin{array}{c} \text{Lottery} \\ \text{choice} \\ (y) \end{array}$	Lottery $y \ge 2$	$y \ge 3$	$y \ge 4$	$y \ge 5$	$y \ge 6$	Self- esteem	Locus of control	BMI at 8 years	Days phys. active	Daily hours sitting	Daily hours playing	Daily hours studying	Cog. Dev.	1th wave - BMI at 12 years	12 year old Daily hours playing	Daily hours studying
DV index	-0.354^{*} $[0.213]$ (0.096)	-0.888^{**} [0.399] (0.026)	-0.577 [0.355] (0.104)	-0.788** [0.387] (0.042)	-0.801^{**} [0.389] (0.039)	-0.833**[0.392](0.034)	-0.660° [0.363] (0.069)	$\begin{array}{c} 0.023 \\ [0.199] \\ (0.908) \end{array}$	$\begin{array}{c} 0.089 \\ [0.111] \\ (0.424) \end{array}$	$\begin{array}{c} 1.047^{**} \\ [0.441] \\ (0.018) \end{array}$	-1.129^{**} [0.459](0.014)	$\begin{array}{c} 0.110 \\ [0.320] \\ (0.730) \end{array}$	-1.687^{***} [0.580] [0.580] (0.004)	$\begin{array}{c} 0.534 \\ [0.364] \\ (0.142) \end{array}$	-0.411 [0.264] (0.119)	$\begin{array}{c} 0.493 \\ [0.360] \\ (0.171) \end{array}$	-1.631^{***} [0.578] (0.005)	$\begin{array}{c} 1.165^{**} \\ [0.474] \\ (0.014) \end{array}$
Cog ability (lagged)	$\begin{array}{c} 0.434^{***} \\ [0.039] \\ (0.000) \end{array}$	-0.016 [0.060] (0.790)	$\begin{array}{c} 0.057 \\ [0.053] \\ (0.282) \end{array}$	-0.004 [0.058] (0.942)	$\begin{array}{c} -0.013 \\ [0.058] \\ (0.829) \end{array}$	-0.035 [0.059] (0.546)	-0.039 [0.054] (0.477)	$\begin{array}{c} 0.068^{**} \\ [0.028] \\ (0.015) \end{array}$	$\begin{array}{c} 0.057^{***} \\ [0.017] \\ (0.001) \end{array}$	$\begin{array}{c} 0.164^{**} \\ [0.066] \\ (0.013) \end{array}$	$\begin{array}{c} -0.089 \\ [0.068] \\ (0.189) \end{array}$	-0.000 [0.048] (0.992)	-0.073 [0.085] (0.393)	$\begin{array}{c} 0.068 \\ [0.054] \\ (0.204) \end{array}$	$\begin{array}{c} 0.400^{***} \\ [0.035] \\ (0.000) \end{array}$	$\begin{array}{c} 0.107^{**} \\ [0.052] \\ (0.041) \end{array}$	-0.162^{*} $[0.085]$ (0.055)	$\begin{array}{c} 0.113\\ [0.069]\\ (0.103) \end{array}$
Mother's capital	$\begin{array}{c} 0.258^{***} \\ [0.028] \\ (0.000) \end{array}$	$\begin{array}{c} 0.102^{**}\\ [0.041]\\ (0.012) \end{array}$	$\begin{array}{c} 0.057 \\ [0.036] \\ (0.115) \end{array}$	$\begin{array}{c} 0.100^{**} \\ [0.039] \\ (0.011) \end{array}$	$\begin{array}{c} 0.075^{*} \\ [0.040] \\ (0.058) \end{array}$	$\begin{array}{c} 0.105^{***} \\ [0.040] \\ (0.009) \end{array}$	$\begin{array}{c} 0.080^{**}\\ [0.037]\\ (0.031) \end{array}$	$\begin{array}{c} 0.097^{***} \\ [0.019] \\ (0.000) \end{array}$	$\begin{array}{c} 0.016 \\ [0.012] \\ (0.173) \end{array}$	$\begin{array}{c} 0.010 \\ [0.045] \\ (0.819) \end{array}$	$\begin{array}{c} 0.094^{**} \\ [0.047] \\ (0.045) \end{array}$	$\begin{array}{c} 0.029 \\ [0.033] \\ (0.381) \end{array}$	0.224^{***} [0.059] (0.000)	$\begin{array}{c} 0.094^{**} \\ [0.037] \\ (0.011) \end{array}$	$\begin{array}{c} 0.276^{***} \\ [0.027] \\ (0.000) \end{array}$	$\begin{array}{c} 0.052 \\ [0.038] \\ (0.175) \end{array}$	$\begin{array}{c} 0.289^{***}\\ [0.062]\\ (0.000) \end{array}$	-0.075 [0.051] (0.143)
Wealth index (w. 1)	$\begin{array}{c} 0.186^{***} \\ [0.019] \\ (0.000) \end{array}$	-0.100^{**} [0.032] (0.002)	$\begin{array}{c} -0.034 \\ [0.029] \\ (0.239) \end{array}$	-0.083^{***} [0.031] (0.008)	-0.077^{**} $[0.031]$ (0.014)	-0.092^{***} $[0.032]$ (0.004)	$\begin{array}{c} -0.110^{***} \\ [0.029] \\ (0.000) \end{array}$	$\begin{array}{c} 0.059^{***}\\ [0.015]\\ (0.000) \end{array}$	0.039^{***} [0.009] (0.000)	$\begin{array}{c} 0.229^{***} \\ [0.036] \\ (0.000) \end{array}$	-0.180^{***} [0.037] (0.000)	$\begin{array}{c} 0.026 \\ [0.026] \\ (0.312) \end{array}$	-0.091^{*} $[0.047]$ (0.052)	$\begin{array}{c} 0.146^{***} \\ [0.029] \\ (0.000) \end{array}$	$\begin{array}{c} 0.153^{***} \\ [0.020] \\ (0.000) \end{array}$	$\begin{array}{c} 0.210^{***} \\ [0.029] \\ (0.000) \end{array}$	$\begin{array}{c} -0.016 \\ [0.047] \\ (0.728) \end{array}$	$\begin{array}{c} 0.148^{***} \\ [0.039] \\ (0.000) \end{array}$
Family size	-0.031^{*} [0.018] (0.099)	$\begin{array}{c} 0.007 \\ [0.031] \\ (0.828) \end{array}$	$\begin{array}{c} -0.012 \\ [0.027] \\ (0.671) \end{array}$	$\begin{array}{c} 0.031 \\ [0.030] \\ (0.299) \end{array}$	$\begin{array}{c} 0.004 \\ [0.030] \\ (0.888) \end{array}$	$\begin{array}{c} -0.002 \\ [0.030] \\ (0.957) \end{array}$	$\begin{array}{c} 0.003 \\ [0.028] \\ (0.909) \end{array}$	-0.036^{**} $[0.015]$ (0.015)	-0.020^{**} [0.009] (0.024)	-0.027 [0.034] (0.424)	$\begin{array}{c} 0.050 \\ [0.035] \\ (0.153) \end{array}$	-0.035 [0.025] (0.149)	$\begin{array}{c} 0.108^{**} \\ [0.044] \\ (0.014) \end{array}$	-0.068^{**} [0.028] (0.014)	$\begin{array}{c} -0.002 \\ [0.018] \\ (0.919) \end{array}$	-0.015 [0.026] (0.569)	$\begin{array}{c} 0.051 \\ [0.043] \\ (0.233) \end{array}$	$\begin{array}{c} -0.046 \\ [0.035] \\ (0.196) \end{array}$
Two-parent HH	$\begin{array}{c} 0.001 \\ [0.019] \end{array}$	-0.008 $[0.030]$ (0.790)	-0.019 [0.026] (0.479)	$\begin{array}{c} 0.005 \\ [0.029] \\ (0.869) \end{array}$	-0.008 $[0.029]$ (0.792)	-0.000 [0.029] (0.995)	-0.014 [0.027] (0.598)	-0.001 [0.014] (0.922)	$\begin{array}{c} 0.006 \\ [0.009] \\ (0.467) \end{array}$	-0.028 [0.033] (0.393)	$\begin{array}{c} 0.015 \\ [0.034] \\ (0.661) \end{array}$	-0.030 [0.024] (0.206)	$\begin{array}{c} 0.028 \\ [0.043] \\ (0.509) \end{array}$	-0.006 [0.027] (0.834)	$\begin{array}{c} 0.028\\ [0.018]\\ (0.121) \end{array}$	-0.033 $[0.026]$ (0.212)	$\begin{array}{c} 0.001 \\ [0.043] \\ (0.975) \end{array}$	-0.013 [0.035] (0.719)
bservations B-squared	1,615 0.154	1,822 -0.515	1,822 -0.200	1,822 -0.418	1,822 -0.446	1,822 -0.463	1,822 -0.256	1,7920.037	1,794	1,828 -0 847	1,828 -1.024	1,817-0.007	1,831-2.369	1,830 -0.273	1,7460.070	1,765 - 0.151	1,760-2.084	1,760-1.056
Anderson-Rubin test A-R F-test	14.63 8.792	12.13 3.975	12.13 1.590	12.13 2.967	12.13 3.081	12.13 3.545	12.13 3.566	10.81 1.357	13.39 0.356	12.28 11.26	12.29 14.16	12.16 3.581	12.15 34.06	12.15 2.491	10.19 5.633	11.40 8.019	11.97 20.51	$11.97 \\ 6.412$
p-value	0.000159	0.0189	0.204	0.0517	0.0461	0.0291	0.0285	0.258	0.701	1.38e-05	7.91e-07	0.0280	0	0.0831	0.00364	0.000341	1.57e-09	0.00168
						stan	dard errors *** p<	in brackets 0.01, ** p<	, p-values ir <0.05, * p<	a parenthes 0.10	es							

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	S OF THE EFFECT OF PARENT erv tice Lottery $y \ge 2$ $y \ge 5$ 1) $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ 1) $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ 11 0.004 0.269 0.277 0.077 123 0.0331 0.0333 0.0341 0.0383 131 0.0343 0.0353 0.0341 0.0783 141 0.0343 0.054 0.0331 0.0343 157 0.054 0.0533 0.0341 0.023 167 0.054 0.0631 0.0333 0.044 133 0.0341 0.0553 0.0563 0.0563 0.0563 133 0.01931 0.0253 0.0333 0.0044 0.074 133 0.0244 0.0253 0.0346 0.0253 0.0246 133 0.0244 0.0233 0.0246 0.0243 0.0243 141 0.0231 0.0243 0.0246 0.074	S OF THE EFFECT OF PARENTAL INN arr ice $y \ge 2$ $y \ge 3$ $y \ge 5$ $y \ge 6$ 1 $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ 1 $y \ge 2$ $y \ge 3$ $y \ge 7$ 0.077 0.2623 2 0.004 0.269 0.277 0.077 0.2633 2 0.0383 (0.317) (0.371) (0.319) 2 0.0383 (0.317) (0.378) (0.319) 2 0.058 (0.326) (0.378) (0.319) 3 (0.024) 0.053 (0.381) (0.331) 3 (0.024) (0.023) (0.040) (0.310) 3 (0.023) (0.023) (0.040) (0.104) 3 (0.023) (0.023) (0.023) (0.033) 3 (0.023) (0.023) (0.023) (0.031) 3 (0.023) (0.023) (0.023) (0.031)	S OF THE EFFECT OF DARENTAL INVESTME erv Self-) $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ esteenn 18 0.004 0.269 0.277 0.077 0.2301 0.123 23 [0.2578] [0.3177] [0.782] [0.3177] [0.381] 0.036 33 [0.058] [0.3278] [0.3377] [0.381] 0.036 33 [0.058] [0.3278] [0.3377] [0.381] 0.036 33 [0.054] [0.3278] [0.3377] [0.381] [0.164] 33 [0.055] [0.055] [0.378] [0.377] [0.381] 33 [0.065] [0.065] [0.065] [0.065] [0.070] 33 [0.065] [0.065] [0.074] [0.070] [0.071] 33 [0.065] [0.065] [0.065] [0.070] [0.070] 33 [0.065] [0.065] [0.065] [0.070] [0.070] 34	SOF THE EFFECT OF PARENTAL INVESTMENT AND reverses Soft Locus error ice Lottery $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ Soft $oottal$ 3 0.004 0.269 0.277 0.077 0.233 0.033 4 $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ Soft $oottal$ 43 0.0383 [0.277] 0.077 0.233 [0.319] [0.0377] [0.733] 43 0.0333 [0.327] [0.341] [0.319] [0.733] [0.0383] [0.703] 5 0.054 -0.034 -0.031 -0.033 [0.741] [0.583] [0.703] [0.703] 5 0.053 [0.704] [0.043] [0.043] [0.043] [0.704] [0.703] 67 [0.033] [0.704] [0.703] [0.043] [0.704] [0.703] 7 0.053 [0.744] [0.733] [0.733] [0.763] [0.763] 1 0.053	S OF THE EFFECT OF PARENTAL INVESTMENT AND DOMES err Lottery MI 1 (re 10.101 0.289 0.277 0.077 0.289 0.113 0.004 0.289 0.317 0.077 0.3317 0.0703 0.041 8 0.041 8 0.041 8 0.041 8 0.041 8 0.041 8 0.041 8 0.045 0.0317 0.0317 0.0317 0.0377 0.0327 0.0337 0.0323 0.0449 0.0377 0.0323 0.0493	S. OF THE EFFECT OF PARENTAL INVESTMENT AND DOMESTIC VI i $y \ge 2$ Locus BMI by a Domestic of transments Locus BMI by a Domestic of transments MIL by a Dows of transments 1 $y \ge 2$ $y \ge 4$ $y \ge 6$ $y \ge 6$ $y \ge 6$ $y \ge 6$ $y \ge 3$	S OF THE EFFECT OF PARENTAL INVESTMENT AND DOMESTIC VIOLENCY in $1 \ge 2 \ge 3 = 1 \ge 3 \ge$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{ c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		
THE EFFECT OF $v \ge 2$ $y \ge 3$ $y \ge 4$ $v \ge 2$ $y \ge 3$ $y \ge 4$ 0.004 0.265 [0.277] 0.983 [0.332] [0.377] 0.084 0.0583 [0.062] 0.0843 [0.0533] [0.062] 0.0843 [0.0533] [0.062] 0.054 -0.033 [0.062] 0.053 [0.053] [0.040] 0.054 -0.024 -0.031 0.0553 [0.040] [0.040] 0.0553 [0.022] [0.041] 0.0553 [0.022] [0.042] 0.0553 [0.023] [0.023] 0.0553 [0.024] [0.025] 0.0533 [0.025] [0.026] 0.1933 [0.025] [0.025] 0.1933 [0.025] [0.025] 0.1933 [0.025] [0.026] 0.1933 [0.025] [0.026] 0.1933 [0.026] [0.026]	THE EFFECT OF PARENT Jentery $y \ge 3$ $y \ge 4$ $y \ge 5$ oottery $y \ge 3$ $y \ge 4$ $y \ge 5$ 0.004 0.269 0.277 0.077 0.0581 0.3371 0.0781 0.0781 0.0583 0.0337 0.0341 0.079** 0.0541 0.0623 0.0337 0.0341 0.0541 0.0631 0.0631 0.023 0.0553 0.0244 0.070* 0.011 0.0553 0.0244 0.070* 0.011 0.0553 0.0254 0.0631 0.023 0.0553 0.0254 0.031 0.023 0.0553 0.0254 0.0263 0.0246 0.074 0.0254 0.0255 0.0233 0.0231 0.0231 0.0254 0.0255 0.0233 0.0231 0.0251 0.0331 0.0255 0.0333 0.0231 0.0231 0.0331 0.0254 0.0333 0.0231 0.0231 0.0331<	THE FFFCT OF PARENTAL INI $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ 0.004 0.268 (0.317) (0.77) (0.262) (0.383) (0.317) (0.782) (0.381) 0.558^* -0.640^* -0.629^* -0.709^{***} -0.518 (0.332) (0.317) (0.782) (0.331) (0.262) (0.332) (0.317) (0.782) (0.381) (0.048) (0.053) (0.038) (0.063) (0.043) (0.104) (0.053) (0.063) (0.714) (0.630) (0.722) (0.033) (0.023) (0.040) (0.031) (0.053) (0.025) (0.031) (0.031) (0.025) (0.041) (0.031) (0.031) (0.053) (0.025) (0.74) (0.35) (0.053) (0.025)	THE EFFECT OF PARENTAL INVESTME $y \ge 2$ $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ esteen 0.004 0.258 0.3312 0.3317 0.2371 0.3319 0.3317 0.0383 0.0377 0.3317 0.3319 0.3319 0.3319 0.3319 0.558* -0.640* -0.629* -0.709** 0.3319 0.3319 0.3377 0.558* -0.640* -0.629 -0.7341 0.3319 0.346 0.3777 0.558* -0.641 0.3319 0.0323 0.045 0.3477 0.558* -0.024 -0.031 0.0233 0.045 0.035 0.056 0.0323 0.0411 0.338 0.045 0.025 0.051 0.0255 0.031 0.0255 0.045 0.025 0.051 0.0251 0.031 0.0351 0.045 0.025 0.041 0.031 0.0251 0.031 0.025 0.040 0.0531 <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{ c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td> <td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td>	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	FFECT OF PARENTAL INN $y \ge 3$ $y \ge 5$ $y \ge 6$ $y \ge 3$ $y \ge 5$ $y \ge 6$ 0.258 0.277 0.077 0.262 0.3323 0.317 0.7823 0.3811 0.640^* 0.377 0.7823 0.3811 0.640^* 0.3817 0.7823 0.3811 0.0533 0.0331 0.03431 0.1044 0.0531 0.0331 0.048 0.1044 0.05361 0.0661 0.0233 0.048 0.0531 0.0631 0.0233 0.0431 0.02231 0.0341 0.0311 0.0321 0.0231 0.0232 0.0441 0.0311 0.0225 0.0441 0.0321 0.0321 0.0231 0.02321 0.0321 0.0321 0.02321 0.03231 0.02321 0.0321 0.03321 0.02321 0.0321 0.0321 0.03321 0.0321	EFFECT OF PARENTAL INVESTME $y \ge 3$ $y \ge 4$ $y \ge 5$ $y \ge 6$ esteem 0.2569 0.277 0.077 0.2362 0.123 0.3323 0.3177 0.2869 0.277 0.077 0.3383 0.0317 0.2380 0.3191 0.1401 0.3383 0.0317 0.2341 0.3177 0.3319 0.1401 0.0548 0.0327 0.0331 0.045 0.1401 0.3411 0.0538 0.0311 0.0233 0.045 0.167 0.3411 0.0551 0.0311 0.0331 0.045 0.031 0.167 0.0533 0.0411 0.3383 0.0441 0.331 0.167 0.0524 0.0311 0.0323 0.1673 0.167 0.222 0.045 0.045 0.045 0.022 0.045 0.227 0.0533 0.1417 0.2338 0.167 0.202 0.0543 0.0411 0.031 0.022 0.045	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	T OF PARENTAL INVESTME $y \ge 4$ $y \ge 5$ $y \ge 6$ esteem $y \ge 4$ $y \ge 5$ $y \ge 6$ esteem 0.277 0.077 0.2301 0.123 0.277 0.7821 0.3311 0.317 0.377 0.7821 0.3341 0.377 0.377 0.7821 0.3341 0.377 0.3371 0.7323 0.1431 0.3461 0.0331 0.045 0.3461 0.3461 0.0536 0.7222 0.3431 0.1671 0.0651 0.0453 0.045 0.0451 0.0651 0.0453 0.0451 0.0211 0.0774 0.0331 0.0621 0.0221 0.0411 0.0331 0.045 0.0211 0.0451 0.0232 0.0451 0.0211 0.0411 0.0331 0.0451 0.0211 0.0411 0.0331 0.0461 0.0211 0.04117	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	T OF PARENTAL INVESTMENT AND DOMESTIC VI $y \ge 5$ $y \ge 6$ selfer ontrol BMI Days. $y \ge 7$ $y \ge 5$ $y \ge 6$ selfer ontrol BMI Days. 0.377 0.077 0.2381 0.1430 0.0333 0.2345 0.294 0.377 0.077 0.2381 0.1430 0.0333 0.2345 0.2346 0.0331 0.0345 0.0144 0.0345 0.0144 0.0324 0.0333 0.0331 0.0333 0.0144 0.045 0.0324 0.0324 0.0331 0.0333 0.0431 0.0333 0.0324 0.0324 0.0411 0.0333 0.0433 0.0433 0.0233 0.0324 0.0774 0.0433 0.0433 0.0443 0.0433 0.053 0.0331 0.0433 0.0433 0.0433 0.0233 0.0343 0.0433 0.0332 0.043	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
	$\begin{array}{c} y \geq 5 \\ y \geq 5 \\ 0.077 \\ 0.077 \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.038) \\ 0.023] \\ 0.023] \\ 0.023] \\ 0.004 \\ 0.004) \\ 0.004 \\ 0.004) \\ 0.004) \\ 0.004 \\ 0.004) \\ 0.0060 \\ 0.0000 \\ 0.0$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ARENTAL INVESTMENT AND DOMESTIC VIOLENCE $y \ge 5$ $y \ge 6$ ener setter of x Days bar Days 0.7732 (0.319) (0.733) (0.167) (0.733) (0.319) (0.737) (0.491) (0.665) (0.665) 0.0177 (0.333) (0.167) (0.033) (0.043) (0.073) (0.665) (0.732)	ARENTAL INVESTMENT AND DOMESTIC VIOLENCE ON VA $\gamma \geq 5$ $y \geq 6$ settern corf BMI bays bars prive baily bai	ARENTAL INVESTMENT AND DOMESTIC VIOLENCE ON VARIOUS $y \ge 5$ $y \ge 6$ enterm control BMI pays builty builty	ARENTAL INVESTMENT AND DOMESTIC VIOLENCE ON VARIOUS OUTCON $y \ge 5$ $y \ge 6$ selfer control BMI Daily Daily	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	

Table A.4: IV REC EXCLUDING WECS	GRESSIO INSTRU	NS OF MENT	THE F	IFFECT	OF P	ARENT	AL INV	ESTMEI	UT AND	DOME	STIC V	IOLENC	CE ON	VARIOU	JS OUTO	COMES -	2 year old	
	Cog. Devlop.	$\begin{array}{c} \text{Lottery} \\ \text{choice} \\ (y) \end{array}$	$\underset{y \ge 2}{\text{Lottery}}$	$y \ge 3$	$y \ge 4$	$y \ge 5$	$y \ge 6$	Self- esteem	Locus of control	BMI at 8 years	Days phys. active	Daily hours sitting	Daily hours playing	Daily hours studying	Cog. Dev.	BMI at 12 years	Daily hours playing	st
Parental investment	$\begin{array}{c} 0.473 \\ [0.316] \\ (0.135) \end{array}$	$\begin{array}{c} 0.243 \\ [0.310] \\ (0.433) \end{array}$	$\begin{array}{c} 0.005 \\ [0.272] \\ (0.986) \end{array}$	$\begin{array}{c} 0.248 \\ [0.272] \\ (0.363) \end{array}$	$\begin{array}{c} 0.261 \\ [0.274] \\ (0.340) \end{array}$	$\begin{array}{c} 0.112 \\ [0.314] \\ (0.722) \end{array}$	$\begin{array}{c} 0.316 \\ [0.332] \\ (0.342) \end{array}$	$\begin{array}{c} 0.172 \\ [0.153] \\ (0.259) \end{array}$	-0.035 [0.089] (0.694)	-0.524 [0.502] (0.297)	$\begin{array}{c} 0.505 \\ [0.594] \\ (0.395) \end{array}$	-0.251 [0.342] (0.463)	$\begin{array}{c} 0.819 \\ [0.809] \\ (0.311) \end{array}$	$\begin{array}{c} 0.545^{**} \\ [0.258] \\ (0.035) \end{array}$	$\begin{array}{c} 0.702^{**} \\ [0.308] \\ (0.023) \end{array}$	-0.693 $[0.502]$ (0.168)	$\begin{array}{c} 1.149 \\ [0.740] \\ (0.120) \end{array}$	920
DV index	-1.259**[0.629](0.045)	-0.890 [0.560] (0.112)	-0.563 [0.492] (0.253)	-0.505 [0.492] (0.304)	-0.532 [0.496] (0.283)	-0.923 [0.568] (0.104)	-1.051* [0.601] (0.081)	-0.249 $[0.284]$ (0.382)	$\begin{array}{c} 0.077 \\ [0.157] \\ (0.626) \end{array}$	$\begin{array}{c} 1.885 * * \\ [0.883] \\ (0.033) \end{array}$	-2.371^{**} [1.065] (0.026)	1.020^{*} $[0.612]$ (0.096)	-3.416^{**} $[1.457]$ (0.019)	$\begin{array}{c} 0.185 \\ [0.467] \\ (0.691) \end{array}$	-0.868° $[0.494]$ (0.079)	$\begin{array}{c} 1.499 * * \\ [0.755] \\ (0.047) \end{array}$	$\begin{array}{c} -2.511^{**} \\ [1.158] \\ (0.030) \end{array}$	020
Lagged index of cog ability	$\begin{array}{c} 0.317^{***} \\ [0.085] \\ (0.000) \end{array}$	-0.034 [0.075] (0.652)	$\begin{array}{c} 0.054 \\ [0.066] \\ (0.417) \end{array}$	$\begin{array}{c} -0.016 \\ [0.066] \\ (0.806) \end{array}$	-0.025 [0.066] (0.704)	-0.036 [0.076] (0.639)	$\begin{array}{c} -0.079 \\ [0.081] \\ (0.327) \end{array}$	$\begin{array}{c} 0.031 \\ [0.036] \\ (0.386) \end{array}$	$\begin{array}{c} 0.059^{***} \\ [0.021] \\ (0.005) \end{array}$	$\begin{array}{c} 0.222^{*} \\ [0.120] \\ (0.063) \end{array}$	$\begin{array}{c} -0.187 \\ [0.143] \\ (0.190) \end{array}$	$\begin{array}{c} 0.075 \\ [0.083] \\ (0.367) \end{array}$	-0.223 $[0.196]$ (0.253)	-0.048 [0.062] (0.443)	$\begin{array}{c} 0.290^{***} \\ [0.065] \\ (0.000) \end{array}$	$\begin{array}{c} 0.197^{*} \\ [0.111] \\ (0.077) \end{array}$	-0.321^{**} [0.160] (0.045)	000
Mother's capital	$\begin{array}{c} 0.240^{***} \\ [0.055] \\ (0.000) \end{array}$	$\begin{array}{c} 0.105^{**} \\ [0.045] \\ (0.019) \end{array}$	$\begin{array}{c} 0.050 \\ [0.039] \\ (0.205) \end{array}$	$\begin{array}{c} 0.089^{**}\\ [0.039]\\ (0.024) \end{array}$	$\begin{array}{c} 0.067^{*} \\ [0.040] \\ (0.089) \end{array}$	$\begin{array}{c} 0.122^{***} \\ [0.046] \\ (0.007) \end{array}$	$\begin{array}{c} 0.095^{**}\\ [0.048]\\ (0.050) \end{array}$	$\begin{array}{c} 0.089^{***} \\ [0.022] \\ (0.000) \end{array}$	$\begin{array}{c} 0.016 \\ [0.013] \\ (0.228) \end{array}$	-0.032 $[0.072]$ (0.653)	$\begin{array}{c} 0.127 \\ [0.086] \\ (0.139) \end{array}$	$\begin{array}{c} 0.019 \\ [0.049] \\ (0.702) \end{array}$	$\begin{array}{c} 0.269^{**} \\ [0.117] \\ (0.021) \end{array}$	$\begin{array}{c} 0.050 \\ [0.037] \\ (0.179) \end{array}$	0.265^{**} $[0.043]$ (0.000)	-0.010 [0.069] (0.888)	$\begin{array}{c} 0.304^{***} \\ [0.102] \\ (0.003) \end{array}$	7 <u>0</u> 0
Wealth index (w. 1)	$\begin{array}{c} 0.221^{***} \\ [0.079] \\ (0.005) \end{array}$	$\begin{array}{c} 0.044 \\ [0.069] \\ (0.523) \end{array}$	-0.044 $[0.060]$ (0.469)	$\begin{array}{c} -0.021 \\ [0.060] \\ (0.726) \end{array}$	$\begin{array}{c} 0.039 \\ [0.061] \\ (0.516) \end{array}$	$\begin{array}{c} 0.091 \\ [0.070] \\ (0.193) \end{array}$	$\begin{array}{c} 0.072 \\ [0.074] \\ (0.326) \end{array}$	$\begin{array}{c} 0.057* \\ [0.034] \\ (0.093) \end{array}$	$\begin{array}{c} 0.009 \\ [0.020] \\ (0.651) \end{array}$	-0.116 $[0.110]$ (0.290)	$\begin{array}{c} 0.030 \\ [0.131] \\ (0.820) \end{array}$	$\begin{array}{c} 0.009 \\ [0.075] \\ (0.900) \end{array}$	$\begin{array}{c} 0.276 \\ [0.178] \\ (0.121) \end{array}$	-0.023 $[0.057]$ (0.687)	$\begin{array}{c} 0.159^{***} \\ [0.061] \\ (0.009) \end{array}$	-0.097 [0.100] (0.327)	$\begin{array}{c} 0.200 \\ [0.148] \\ (0.177) \end{array}$	9 <u>0</u> 0
Family size	$\begin{array}{c} -0.015 \\ [0.034] \\ (0.665) \end{array}$	$\begin{array}{c} 0.001 \\ [0.032] \\ (0.974) \end{array}$	-0.018 $[0.028]$ (0.514)	$\begin{array}{c} 0.023 \\ [0.028] \\ (0.423) \end{array}$	-0.004 [0.028] (0.876)	-0.006 [0.032] (0.844)	$\begin{array}{c} 0.006 \\ [0.034] \\ (0.858) \end{array}$	-0.033^{**} [0.017] (0.044)	$\begin{array}{c} -0.018^{*} \\ [0.009] \\ (0.055) \end{array}$	-0.043 $[0.051]$ (0.403)	$\begin{array}{c} 0.076 \\ [0.061] \\ (0.211) \end{array}$	-0.054 [0.035] (0.126)	$\begin{array}{c} 0.134 \\ [0.083] \\ (0.106) \end{array}$	-0.058^{**} [0.026] (0.027)	$\begin{array}{c} 0.004 \\ [0.026] \\ (0.869) \end{array}$	-0.036 [0.043] (0.404)	$\begin{array}{c} 0.066 \\ [0.064] \\ (0.307) \end{array}$	9 <u>0</u> 0
Two-parent HH	-0.002 $[0.035]$ (0.962)	$\begin{array}{c} -0.011 \\ [0.031] \\ (0.735) \end{array}$	$\begin{array}{c} -0.024 \\ [0.027] \\ (0.382) \end{array}$	$\begin{array}{c} -0.003 \\ [0.027] \\ (0.909) \end{array}$	-0.013 [0.027] (0.639)	$\begin{array}{c} 0.004 \\ [0.031] \\ (0.898) \end{array}$	-0.014 [0.033] (0.678)	$\begin{array}{c} -0.007 \\ [0.016] \\ (0.654) \end{array}$	$\begin{array}{c} 0.008 \\ [0.009] \\ (0.362) \end{array}$	-0.029 $[0.050]$ (0.564)	$\begin{array}{c} 0.013 \\ [0.059] \\ (0.826) \end{array}$	-0.028 [0.034] (0.419)	$\begin{array}{c} 0.023 \\ [0.081] \\ (0.780) \end{array}$	-0.026 [0.026] (0.303)	$\begin{array}{c} 0.009 \\ [0.026] \\ (0.719) \end{array}$	-0.030 [0.044] (0.489)	-0.014 [0.066] (0.830)	0 <u>0</u> 0
Income (w. 2)	-0.039 [0.097] (0.689)	-0.074 [0.093] (0.428)	$\begin{array}{c} 0.036 \\ [0.082] \\ (0.663) \end{array}$	$\begin{array}{c} -0.061 \\ [0.082] \\ (0.458) \end{array}$	-0.096 [0.082] (0.245)	-0.060 [0.094] (0.522)	$\begin{array}{c} -0.088 \\ [0.100] \\ (0.379) \end{array}$	$\begin{array}{c} 0.020 \\ [0.046] \\ (0.662) \end{array}$	$\begin{array}{c} 0.005 \\ [0.027] \\ (0.863) \end{array}$	$\begin{array}{c} 0.198\\ [0.150]\\ (0.188) \end{array}$	-0.126 [0.178] (0.478)	$\begin{array}{c} 0.006 \\ [0.102] \\ (0.956) \end{array}$	-0.133 [0.243] (0.583)	$\begin{array}{c} 0.016 \\ [0.077] \\ (0.840) \end{array}$	-0.076 [0.094] (0.420)	0.332^{**} [0.153] (0.030)	-0.267 [0.229] (0.245)	0.3
Income sq. (w. 2)	$\begin{array}{c} 0.021 \\ [0.083] \\ (0.805) \end{array}$	$\begin{array}{c} 0.073 \\ [0.080] \\ (0.361) \end{array}$	$\begin{array}{c} -0.025 \\ [0.070] \\ (0.724) \end{array}$	$\begin{array}{c} 0.057 \\ [0.070] \\ (0.416) \end{array}$	$\begin{array}{c} 0.088 \\ [0.071] \\ (0.215) \end{array}$	$\begin{array}{c} 0.062 \\ [0.081] \\ (0.442) \end{array}$	$\begin{array}{c} 0.088\\ [0.086]\\ (0.309) \end{array}$	-0.006 [0.040] (0.882)	$\begin{array}{c} 0.003 \\ [0.023] \\ (0.891) \end{array}$	-0.144 $[0.129]$ (0.265)	$\begin{array}{c} 0.049 \\ [0.153] \\ (0.748) \end{array}$	$\begin{array}{c} -0.009 \\ [0.088] \\ (0.916) \end{array}$	$\begin{array}{c} 0.068 \\ [0.209] \\ (0.746) \end{array}$	$\begin{array}{c} -0.018 \\ [0.067] \\ (0.787) \end{array}$	$\begin{array}{c} 0.061 \\ [0.075] \\ (0.420) \end{array}$	-0.230^{*} $[0.124]$ (0.063)	$\begin{array}{c} 0.175 \\ [0.184] \\ (0.342) \end{array}$	000
No income reported (w. 2)	-0.049 [0.063] (0.441)	$\begin{array}{c} 0.101^{*} \\ [0.061] \\ (0.097) \end{array}$	$\begin{array}{c} 0.038 \\ [0.054] \\ (0.477) \end{array}$	$\begin{array}{c} 0.065 \\ [0.053] \\ (0.226) \end{array}$	$\begin{array}{c} 0.075 \\ [0.054] \\ (0.163) \end{array}$	$\begin{array}{c} 0.099 \\ [0.062] \\ (0.108) \end{array}$	$\begin{array}{c} 0.120^{*} \\ [0.065] \\ (0.066) \end{array}$	-0.015 [0.030] (0.609)	$\begin{array}{c} 0.000 \\ [0.018] \\ (0.992) \end{array}$	-0.036 [0.098] (0.717)	-0.099 [0.116] (0.395)	$\begin{array}{c} 0.034 \\ [0.067] \\ (0.615) \end{array}$	-0.071 [0.159] (0.653)	-0.043 [0.051] (0.398)	-0.041 [0.055] (0.447)	$\begin{array}{c} 0.023 \\ [0.091] \\ (0.797) \end{array}$	-0.004 [0.135] (0.978)	000
Proxy means test for CCT	$\begin{array}{c} 0.174 \\ [0.128] \\ (0.175) \end{array}$	$\begin{array}{c} 0.218^{*} \\ [0.119] \\ (0.067) \end{array}$	$\begin{array}{c} -0.003 \\ [0.105] \\ (0.976) \end{array}$	$\begin{array}{c} 0.115 \\ [0.105] \\ (0.272) \end{array}$	$\begin{array}{c} 0.176^{*} \\ [0.105] \\ (0.095) \end{array}$	0.237^{**} [0.121] (0.049)	$\begin{array}{c} 0.292^{**} \\ [0.128] \\ (0.023) \end{array}$	$\begin{array}{c} 0.053 \\ [0.059] \\ (0.369) \end{array}$	-0.043 [0.033] (0.188)	-0.530^{***} [0.192] (0.006)	$\begin{array}{c} 0.386^{*} \\ [0.228] \\ (0.090) \end{array}$	-0.109 [0.131] (0.404)	$\begin{array}{c} 0.664^{**} \\ [0.310] \\ (0.032) \end{array}$	-0.057 [0.099] (0.562)	$\begin{array}{c} 0.177\\ [0.117]\\ (0.132) \end{array}$	$\begin{array}{c} -0.510^{***} \\ [0.190] \\ (0.007) \end{array}$	$\begin{array}{c} 0.516^{*} \\ [0.278] \\ (0.064) \end{array}$	0 <u>0</u> 0
Observations R-squared Anderson-Rubin test A-R F-test P-Value	1,615 -1.642 5.221 9.787 5.96e-05	$\begin{array}{c} 1,819\\ -0.533\\ 6.158\\ 2.115\\ 0.121\end{array}$	$\begin{array}{c} 1,819\\ -0.189\\ 6.158\\ 0.806\\ 0.447\end{array}$	$\begin{array}{c} 1,819\\ -0.174\\ 6.158\\ 0.920\\ 0.399\end{array}$	$\begin{array}{c} 1,819\\ -0.202\\ 6.158\\ 1.031\\ 0.357\end{array}$	$\begin{array}{c} 1,819\\ -0.573\\ 6.158\\ 2.099\\ 0.123\end{array}$	$\begin{array}{c} 1,819\\ -0.763\\ 6.158\\ 3.053\\ 0.0475\end{array}$	$\begin{array}{c} 1,789\\ -0.090\\ 5.873\\ 0.954\\ 0.385\end{array}$	$\begin{array}{c} 1,791 \\ -0.017 \\ 6.602 \\ 0.178 \\ 0.837 \end{array}$	$\begin{array}{c} 1,826\\ -2.914\\ 6.448\\ 10.37\\ 3.32e-05\end{array}$	1,825 -4.559 6.215 14.73 4.51e-07	$\begin{array}{c} 1,814 \\ -0.895 \\ 6.184 \\ 2.776 \\ 0.0626 \end{array}$	$\begin{array}{c} 1,828\\ -9.944\\ 6.200\\ 33.45\\ 0\end{array}$	$\begin{array}{c} 1,827\\ -0.064\\ 6.141\\ 2.897\\ 0.0554\end{array}$	$\begin{array}{c} 1,743 \\ -0.807 \\ 5.467 \\ 7.528 \\ 0.000555 \end{array}$	$\begin{array}{c} 1,762\\ -1.909\\ 6.394\\ 7.115\\ 0.000837\end{array}$	$\begin{array}{c} 1,757\\ -5.430\\ 6.078\\ 18.88\\ 7.73e-09\end{array}$	0.0 % 0.1 0.0
						stand	ard errors *** p<	in brackets 0.01, ** p<	, p-values i <0.05, * p<	n parenthes 0.10	es							

Table A.5: IV regressions of the effect of Parental Investment and Domestic Violence on Various Outcomes -

EACLUDING NEVER	STO STO	OALIFI	ERS															
		Lotterv							Locus	BMI	Davs	Dailv	Dailv	Daily -	Cog.	4th wave - 1 BMI	.2 year old Dailv	Dailv
	Cog. Devlop.	$_{(y)}^{choice}$	$\underset{y \geq 2}{\text{Lottery}}$	$y \ge 3$	$y \ge 4$	$y \ge 5$	$y \ge 6$	Self- esteem	of control	at 8 years	phys. active	hours	hours playing	hours studying	Dev.	at 12 years	hours playing	hours studying
Parental investment	$1.048 \\ [0.810] \\ (0.196)$	-0.936 [1.462] (0.522)	$\begin{array}{c} 0.243 \\ [1.343] \\ (0.857) \end{array}$	$\begin{array}{c} 0.101 \\ [1.509] \\ (0.947) \end{array}$	$\begin{array}{c} -0.666 \\ [1.385] \\ (0.630) \end{array}$	-1.206 [1.418] (0.395)	-1.744 [1.456] (0.231)	-0.359 [0.787] (0.648)	-0.056 [0.413] (0.892)	-0.853 [1.150] (0.458)	5.715^{**} [2.862] (0.046)	-2.233 [1.547] (0.149)	$\begin{array}{c} 3.155 \\ [2.235] \\ (0.158) \end{array}$	-0.439 [1.172] (0.708)	$\begin{array}{c} 0.233 \\ [0.985] \\ (0.813) \end{array}$	-1.465 [1.099] (0.183)	$\begin{array}{c} 0.167 \\ [1.678] \\ (0.921) \end{array}$	$\begin{array}{c} 0.114 \\ [1.069] \\ (0.915) \end{array}$
DV index	-0.400*[0.214](0.061)	-0.783^{*} $[0.415]$ (0.060)	-0.631^{*} [0.381] (0.098)	-0.944^{**} [0.429] (0.028)	-0.681^{*} $[0.393]$ (0.083)	-0.617 [0.403] (0.126)	-0.454 [0.414] (0.272)	$\begin{array}{c} 0.281 \\ [0.244] \\ (0.249) \end{array}$	$\begin{array}{c} 0.005 \\ [0.117] \\ (0.966) \end{array}$	$\begin{array}{c} 0.393 \\ [0.326] \\ (0.227) \end{array}$	-0.676 [0.810] (0.404)	-0.370 [0.461] (0.422)	-1.146° [0.633] (0.070)	$\begin{array}{c} 0.171 \\ [0.331] \\ (0.605) \end{array}$	-0.485*[0.294](0.099)	$\begin{array}{c} 0.152 \\ [0.346] \\ (0.660) \end{array}$	-1.318^{**} [0.519] (0.011)	$\begin{array}{c} 0.394 \\ [0.331] \\ (0.234) \end{array}$
Lagged index of cog ability	$\begin{array}{c} 0.368^{***}\\ [0.070]\\ (0.000) \end{array}$	$\begin{array}{c} 0.104 \\ [0.128] \\ (0.414) \end{array}$	$\begin{array}{c} 0.100 \\ [0.117] \\ (0.393) \end{array}$	$\begin{array}{c} 0.007 \\ [0.132] \\ (0.957) \end{array}$	$\begin{array}{c} 0.084 \\ [0.121] \\ (0.486) \end{array}$	$\begin{array}{c} 0.098 \\ [0.124] \\ (0.428) \end{array}$	$\begin{array}{c} 0.136 \\ [0.127] \\ (0.286) \end{array}$	$\begin{array}{c} 0.065 \\ [0.066] \\ (0.323) \end{array}$	$\begin{array}{c} 0.041 \\ [0.034] \\ (0.229) \end{array}$	$\begin{array}{c} 0.129 \\ [0.100] \\ (0.200) \end{array}$	-0.503^{**} [0.250] (0.044)	$\begin{array}{c} 0.130 \\ [0.135] \\ (0.335) \end{array}$	-0.273 [0.195] (0.162)	$\begin{array}{c} 0.066 \\ [0.102] \\ (0.518) \end{array}$	$\begin{array}{c} 0.337^{***} \\ [0.081] \\ (0.000) \end{array}$	$\begin{array}{c} 0.107 \\ [0.101] \\ (0.291) \end{array}$	-0.128 [0.148] (0.386)	$\begin{array}{c} 0.040 \\ [0.094] \\ (0.667) \end{array}$
Mother's capital	$\begin{array}{c} 0.160^{***} \\ [0.058] \\ (0.005) \end{array}$	$\begin{array}{c} 0.162^{**} \\ [0.066] \\ (0.014) \end{array}$	$\begin{array}{c} 0.058 \\ [0.060] \\ (0.340) \end{array}$	$\begin{array}{c} 0.126^{*} \\ [0.068] \\ (0.064) \end{array}$	$\begin{array}{c} 0.123^{**} \\ [0.062] \\ (0.049) \end{array}$	$\begin{array}{c} 0.178^{***} \\ [0.064] \\ (0.005) \end{array}$	$\begin{array}{c} 0.157^{**} \\ [0.066] \\ (0.017) \end{array}$	$\begin{array}{c} 0.088^{**}\\ [0.036]\\ (0.014) \end{array}$	$\begin{array}{c} 0.018 \\ [0.019] \\ (0.354) \end{array}$	$\begin{array}{c} 0.005 \\ [0.052] \\ (0.919) \end{array}$	-0.106 [0.128] (0.409)	$\begin{array}{c} 0.127 ^{*} \\ [0.072] \\ (0.077) \end{array}$	$\begin{array}{c} 0.123 \\ [0.100] \\ (0.220) \end{array}$	$\begin{array}{c} 0.060 \\ [0.053] \\ (0.258) \end{array}$	$\begin{array}{c} 0.250^{***} \\ [0.040] \\ (0.000) \end{array}$	$\begin{array}{c} 0.055 \\ [0.050] \\ (0.270) \end{array}$	$\begin{array}{c} 0.301^{***} \\ [0.077] \\ (0.000) \end{array}$	-0.087^{*} [0.049] (0.075)
Wealth index (w. 1)	$\begin{array}{c} 0.197^{***} \\ [0.053] \\ (0.000) \end{array}$	$\begin{array}{c} 0.051 \\ [0.078] \\ (0.519) \end{array}$	$\begin{array}{c} -0.017 \\ [0.072] \\ (0.817) \end{array}$	$\begin{array}{c} 0.026 \\ [0.081] \\ (0.752) \end{array}$	$\begin{array}{c} 0.057 \\ [0.074] \\ (0.443) \end{array}$	$\begin{array}{c} 0.071 \\ [0.076] \\ (0.349) \end{array}$	$\begin{array}{c} 0.047 \\ [0.078] \\ (0.546) \end{array}$	$\begin{array}{c} 0.015 \\ [0.043] \\ (0.737) \end{array}$	$\begin{array}{c} 0.004 \\ [0.022] \\ (0.852) \end{array}$	-0.072 [0.061] (0.239)	-0.022 [0.152] (0.885)	$\begin{array}{c} 0.097 \\ [0.087] \\ (0.265) \end{array}$	$\begin{array}{c} 0.223^{*} \\ [0.119] \\ (0.061) \end{array}$	-0.047 [0.062] (0.454)	$\begin{array}{c} 0.145^{***} \\ [0.052] \\ (0.006) \end{array}$	-0.006 $[0.066]$ (0.932)	$\begin{array}{c} 0.101 \\ [0.100] \\ (0.312) \end{array}$	-0.053 [0.064] (0.404)
Family size	-0.078^{**} [0.032] (0.013)	$\begin{array}{c} 0.052 \\ [0.052] \\ (0.319) \end{array}$	$\begin{array}{c} 0.013 \\ [0.048] \\ (0.783) \end{array}$	$\begin{array}{c} 0.056 \\ [0.053] \\ (0.292) \end{array}$	$\begin{array}{c} 0.034 \\ [0.049] \\ (0.489) \end{array}$	$\begin{array}{c} 0.049 \\ [0.050] \\ (0.330) \end{array}$	$\begin{array}{c} 0.052 \\ [0.052] \\ (0.312) \end{array}$	-0.031 [0.028] (0.254)	-0.005 [0.015] (0.752)	$\begin{array}{c} 0.020 \\ [0.040] \\ (0.608) \end{array}$	$\begin{array}{c} -0.100 \\ [0.099] \\ (0.315) \end{array}$	$\begin{array}{c} -0.011 \\ [0.055] \\ (0.847) \end{array}$	$\begin{array}{c} -0.010 \\ [0.078] \\ (0.897) \end{array}$	-0.061 [0.041] (0.134)	$\begin{array}{c} -0.035 \\ [0.037] \\ (0.342) \end{array}$	$\begin{array}{c} 0.013 \\ [0.043] \\ (0.766) \end{array}$	$\begin{array}{c} 0.009 \\ [0.065] \\ (0.886) \end{array}$	-0.005 [0.041] (0.899)
Two-parent HH	-0.061^{*} [0.037] (0.093)	$\begin{array}{c} 0.028 \\ [0.057] \\ (0.623) \end{array}$	$\begin{array}{c} -0.043 \\ [0.052] \\ (0.414) \end{array}$	$\begin{array}{c} 0.019 \\ [0.059] \\ (0.748) \end{array}$	$\begin{array}{c} 0.030 \\ [0.054] \\ (0.577) \end{array}$	$\begin{array}{c} 0.047 \\ [0.055] \\ (0.394) \end{array}$	$\begin{array}{c} 0.035 \\ [0.057] \\ (0.539) \end{array}$	$\begin{array}{c} 0.008 \\ [0.031] \\ (0.787) \end{array}$	$\begin{array}{c} 0.012 \\ [0.016] \\ (0.468) \end{array}$	$\begin{array}{c} 0.018 \\ [0.045] \\ (0.683) \end{array}$	$\begin{array}{c} -0.135 \\ [0.111] \\ (0.225) \end{array}$	$\begin{array}{c} 0.049 \\ [0.063] \\ (0.442) \end{array}$	$\begin{array}{c} -0.061 \\ [0.087] \\ (0.487) \end{array}$	$\begin{array}{c} 0.005 \\ [0.045] \\ (0.920) \end{array}$	$\begin{array}{c} 0.005 \\ [0.037] \\ (0.890) \end{array}$	$\begin{array}{c} 0.026 \\ [0.045] \\ (0.566) \end{array}$	-0.006 [0.067] (0.924)	$\begin{array}{c} 0.029 \\ [0.043] \\ (0.495) \end{array}$
Income (w. 2)	$\begin{array}{c} 0.077 \\ [0.230] \\ (0.739) \end{array}$	$\begin{array}{c} 0.156 \\ [0.374] \\ (0.677) \end{array}$	$\begin{array}{c} -0.141 \\ [0.343] \\ (0.681) \end{array}$	-0.064 [0.386] (0.868)	$\begin{array}{c} 0.115 \\ [0.354] \\ (0.746) \end{array}$	$\begin{array}{c} 0.194 \\ [0.363] \\ (0.592) \end{array}$	$\begin{array}{c} 0.387 \\ [0.372] \\ (0.299) \end{array}$	$\begin{array}{c} 0.202 \\ [0.205] \\ (0.323) \end{array}$	$\begin{array}{c} 0.018\\ [0.110]\\ (0.873) \end{array}$	$\begin{array}{c} 0.432 \\ [0.295] \\ (0.143) \end{array}$	-1.450^{**} [0.733] (0.048)	$\begin{array}{c} 0.575 \\ [0.389] \\ (0.139) \end{array}$	-0.600 [0.573] (0.295)	$\begin{array}{c} 0.623^{**}\\ [0.300]\\ (0.038) \end{array}$	$\begin{array}{c} 0.222 \\ [0.261] \\ (0.395) \end{array}$	$\begin{array}{c} 0.823^{***} \\ [0.281] \\ (0.003) \end{array}$	-0.028 [0.441] (0.949)	$\begin{array}{c} 0.170 \\ [0.281] \\ (0.545) \end{array}$
Income sq. (w. 2)	-1.579^{*} [0.911](0.083)	$\begin{array}{c} 0.367 \\ [1.294] \\ (0.777) \end{array}$	$\begin{array}{c} 1.173 \\ [1.188] \\ (0.324) \end{array}$	$\begin{array}{c} 0.768 \\ [1.335] \\ (0.565) \end{array}$	$\begin{array}{c} 0.415 \\ [1.226] \\ (0.735) \end{array}$	$\begin{array}{c} 0.039 \\ [1.255] \\ (0.975) \end{array}$	-0.371 [1.288] (0.773)	-0.384 [0.716] (0.592)	$\begin{array}{c} 0.099\\ [0.382]\\ (0.795) \end{array}$	-0.749 [1.009] (0.458)	-0.752 [2.513] (0.765)	$\begin{array}{c} 0.730 \\ [1.444] \\ (0.613) \end{array}$	-0.626 [1.961] (0.749)	-1.765° [1.026] (0.085)	-0.443 $[0.859]$ (0.606)	-1.222 [1.125] (0.277)	-0.907 [1.726] (0.599)	-0.665 [1.099] (0.545)
No income reported (w. 2)	-0.018 [0.084] (0.827)	$\begin{array}{c} 0.020 \\ [0.127] \\ (0.877) \end{array}$	$\begin{array}{c} 0.019 \\ [0.117] \\ (0.869) \end{array}$	$\begin{array}{c} 0.014 \\ [0.131] \\ (0.913) \end{array}$	$\begin{array}{c} 0.013 \\ [0.120] \\ (0.916) \end{array}$	$\begin{array}{c} 0.012 \\ [0.123] \\ (0.922) \end{array}$	$\begin{array}{c} 0.024 \\ [0.127] \\ (0.848) \end{array}$	$\begin{array}{c} 0.004 \\ [0.070] \end{array}$	-0.023 [0.037] (0.525)	-0.072 [0.099] (0.470)	$\begin{array}{c} 0.086 \\ [0.247] \\ (0.729) \end{array}$	$\begin{array}{c} -0.071 \\ [0.140] \\ (0.612) \end{array}$	$\begin{array}{c} 0.057 \\ [0.193] \\ (0.769) \end{array}$	-0.137 [0.101] (0.174)	-0.045 [0.082] (0.585)	$\begin{array}{c} -0.023 \\ [0.106] \\ (0.827) \end{array}$	-0.020 [0.161] (0.902)	$\begin{array}{c} -0.001 \\ [0.103] \\ (0.991) \end{array}$
Proxy means test for CCT	$\begin{array}{c} 0.172 \\ [0.171] \\ (0.314) \end{array}$	-0.059 [0.337] (0.861)	$\begin{array}{c} 0.102 \\ [0.309] \\ (0.741) \end{array}$	$\begin{array}{c} 0.132 \\ [0.348] \\ (0.705) \end{array}$	-0.010 [0.319] (0.975)	-0.113 [0.327] (0.729)	-0.250 [0.336] (0.457)	-0.152 [0.186] (0.412)	-0.056 [0.093] (0.545)	-0.458^{*} $[0.264]$ (0.082)	$\begin{array}{c} 1.196^{*} \\ [0.656] \\ (0.069) \end{array}$	-0.334 [0.362] (0.356)	$\begin{array}{c} 1.002^{*} \\ [0.512] \\ (0.050) \end{array}$	-0.307 [0.268] (0.252)	$\begin{array}{c} 0.068 \\ [0.229] \\ (0.766) \end{array}$	-0.513** $[0.260]$ (0.049)	$\begin{array}{c} 0.181 \\ [0.388] \\ (0.641) \end{array}$	-0.218 [0.247] (0.377)
Observations R-squared Anderson-Rubin test A-R F-test	937 0.085 7.375 2.787	1,099 -0.490 4.618 3.329	1,099 -0.255 4.618 1.930	$\begin{array}{c} 1,099\\ -0.649\\ 4.618\\ 3.310\\ \end{array}$	1,099 -0.378 4.618 2.161	$\begin{array}{c} 1,099\\ -0.411\\ 4.618\\ 2.657\end{array}$	$\begin{array}{c} 1,099\\ -0.456\\ 4.618\\ 2.631\end{array}$	1,074 -0.238 4.617 0.841	1,077 0.013 4.972 0.273	1,104 -0.376 4.587 1.717	$1,103 \\ -4.424 \\ 4.570 \\ 10.62 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ $	1,094 -0.850 4.938 2.903	$1,104 \\ -2.375 \\ 4.587 \\ 12.25 \\ 0.000 \\ 0.0$	$\begin{array}{c} 1,103 \\ -0.066 \\ 4.571 \\ 2.219 \end{array}$	1,048 -0.031 4.059 2.583	1,062 -0.430 5.621 2.183	$\begin{array}{c} 1,059\\ -1.407\\ 5.729\\ 10.61\\ -10.61\end{array}$	$\begin{array}{c} 1,059\\ -0.137\\ 5.729\\ 0.736\\ 0.736\end{array}$
p-value	0.0397	0.0190	0.123	9610.0	0160.0	0.0472 standard	$0.0488 \\ errors in b \\ ** p < 0.01$	$\frac{0.471}{\text{rackets, p-}}, ** p<0.0$	$\frac{0.845}{-values in}$ 05, * p<0.	0.162 parenthese .10	6.92e-07 es	0.0339	6.92e-08	0.0843	0.0521	9880.0	7.13e-07	0.531

	Cogr	nitive D	evelopme	net]	Lottery	decision	
Variable	IV coeff.	S.E.	Correcte	d Diff.	IV coeff.	S.E.	Correcte	ed Diff.
			coeff.				coeff.	
Parental Investment	0.349	0.179	0.349	0.000	0.216	0.283	0.196	0.020
DV index	-0.341	0.196	-0.335	-0.006	-0.733	0.345	-0.664	-0.069
Cog. Development (w. 2)	0.383	0.043	0.439	-0.056	-0.024	0.067	-0.021	-0.002
Mother's capital	0.231	0.030	0.231	-0.001	0.102	0.041	0.130	-0.028
Wealth index $(w. 1)$	0.155	0.039	0.153	0.002	0.034	0.059	0.031	0.003
Family size	-0.028	0.019	-0.027	-0.001	-0.002	0.029	-0.002	0.000
Two-parent HH	-0.013	0.019	-0.014	0.001	-0.011	0.029	-0.008	-0.002
Income (w. 2)	-0.012	0.053	-0.016	0.004	-0.070	0.087	-0.068	-0.002
Income sq. $(w. 2)$	0.008	0.046	0.011	-0.002	0.072	0.075	0.071	0.001
No income reported (w. 2)	-0.037	0.035	-0.036	-0.001	0.097	0.057	0.098	-0.001
Proxy means test for CCT	0.047	0.059	0.051	-0.005	0.195	0.096	0.188	0.006

Table A.6: Effect of measurement error on estimated coefficients

Note: Measurement error correction of coefficient estimates follows the approach described in Section B

Figure A.1: Effect of domestic violence and mother's time investment. Pairs of parameters not rejected by data.



Coeff. of domestic violence



	Factor 1	Factor 2	Factor 3
Monetary investments			
School related expenses	0.7244	-0.0272	-0.1096
Children gifts	0.7211	0.0068	0.0140
No. Books at home	0.5575	-0.0058	0.2418
<u>Time & attention investments</u>			
Help with homework	-0.1317	-0.0627	0.6125
Encourages to read	0.0696	0.0316	0.5957
Know child' friends' names	0.0390	0.0162	0.6809
Environmental variables			
Reported violence (wave 3)	-0.0427	0.7009	-0.0366
Reported violence (wave 2)	-0.0346	0.6421	0.0408
Seeked help due to IPV	-0.0233	0.4524	0.0208
Partner smokes	0.1879	0.4875	-0.0108

Table A.7: Confirmatory factor analysis of parental investment and domestic violence measures

Table A.8: Negative Symmetry between Domestic Violence and (Time) Parental Investments

		Domestic v	iolence inde	<u>c</u>	Parental	investment	s of time &	attention
	First	Cog.	First	Lot.	First	Cog.	First	Lot
VARIABLES	stage	Dev.	stage	Choice	stage	Dev.	stage	Choice
Domestic violence index		-0.354^{*} [0.213] (0.096)		-0.888^{**} [0.399] (0.026)				
Parental investments (time & attention)						$\begin{array}{c} 0.813^{**} \\ [0.322] \\ (0.012) \end{array}$		1.445^{**} [0.646] (0.025)
Wage gap	-0.054^{**} [0.027] (0.042)		-0.055^{**} [0.026] (0.032)		$\begin{array}{c} 0.048^{***} \\ [0.014] \\ (0.000) \end{array}$		0.031^{**} [0.013] (0.023)	
Opening of Women Emergency Center	-0.076^{***} [0.023] (0.001)		-0.060^{***} [0.022] (0.006)		0.040^{***} [0.012] (0.001)		$\begin{array}{c} 0.038^{***} \\ [0.012] \\ (0.001) \end{array}$	
Lagged index of cog ability	-0.066 [0.045] (0.140)	$\begin{array}{c} 0.434^{***} \\ [0.039] \\ (0.000) \end{array}$	-0.050 [0.041] (0.227)	-0.016 [0.060] (0.790)	$\begin{array}{c} 0.037 \\ [0.023] \\ (0.103) \end{array}$	$\begin{array}{c} 0.429^{***} \\ [0.037] \\ (0.000) \end{array}$	0.050^{**} [0.022] (0.022)	-0.044 [0.064] (0.488)
Mother's capital	$\begin{array}{c} 0.032 \\ [0.035] \\ (0.360) \end{array}$	$\begin{array}{c} 0.258^{***} \\ [0.028] \\ (0.000) \end{array}$	$\begin{array}{c} 0.047 \\ [0.030] \\ (0.116) \end{array}$	0.102^{**} [0.041] (0.012)	$\begin{array}{c} 0.188^{***} \\ [0.018] \\ (0.000) \end{array}$	$\begin{array}{c} 0.091 \\ [0.071] \\ (0.201) \end{array}$	$\begin{array}{c} 0.158^{***} \\ [0.016] \\ (0.000) \end{array}$	-0.165 [0.119] (0.167)
Wealth index (w. 1)	-0.050^{**} [0.024] (0.035)	0.186^{***} [0.019] (0.000)	-0.050^{**} [0.023] (0.028)	-0.100^{***} [0.032] (0.002)	0.055^{***} [0.012] (0.000)	$\begin{array}{c} 0.162^{***} \\ [0.022] \\ (0.000) \end{array}$	0.050^{***} [0.012] (0.000)	-0.128*** [0.038] (0.001)
Family size	$\begin{array}{c} 0.018 \\ [0.022] \\ (0.426) \end{array}$	-0.031* [0.018] (0.099)	$\begin{array}{c} 0.025 \\ [0.021] \\ (0.233) \end{array}$	$\begin{array}{c} 0.007 \\ [0.031] \\ (0.828) \end{array}$	-0.023^{**} [0.011] (0.042)	-0.017 [0.019] (0.363)	-0.026^{**} [0.011] (0.021)	$\begin{array}{c} 0.025 \\ [0.033] \\ (0.452) \end{array}$
Two-parent HH	$\begin{array}{c} 0.016 \\ [0.023] \\ (0.489) \end{array}$	0.001 [0.019] (0.970)	$\begin{array}{c} 0.007 \\ [0.021] \\ (0.754) \end{array}$	-0.008 [0.030] (0.790)	$\begin{array}{c} 0.001 \\ [0.011] \\ (0.955) \end{array}$	-0.004 [0.018] (0.803)	-0.001 [0.011] (0.916)	-0.009 [0.029] (0.755)
Observations R-squared	$1,615 \\ 0.012$	$1,615 \\ 0.154$	$1,822 \\ 0.009$	$1,822 \\ -0.515$	$1,620 \\ 0.116$	$1,620 \\ 0.202$	$1,829 \\ 0.101$	$1,829 \\ -0.454$

pval in brackets *** p<0.01, ** p<0.05, * p<0.10

OUTCOMES																4th wave -]	2 year old	
	Cog. Devlop.	$\begin{array}{c} \text{Lottery} \\ \text{choice} \\ (y) \end{array}$	$\begin{array}{c} \text{Lottery} \\ y \geq 2 \end{array}$	$y \ge 3$	$y \ge 4$	$y \ge 5$	$y \ge 6$	Self- esteem	Locus of control	BMI at 8 years	Days phys. active	Daily hours sitting	Daily hours playing	Daily hours studying	Cog. Dev.	BMI at 12 years	Daily hours playing	Daily hours studying
Investments of money	-0.131 [0.233] (0.574)	-0.448 [0.355] (0.207)	-0.455 [0.334] (0.173)	-0.283 [0.346] (0.414)	-0.274 [0.346] (0.428)	-0.552 [0.350] (0.115)	-0.320 [0.334] (0.337)	$\begin{array}{c} 0.087 \\ [0.182] \\ (0.632) \end{array}$	$\begin{array}{c} 0.026 \\ [0.110] \\ (0.810) \end{array}$	$\begin{array}{c} 0.569 \\ [0.398] \\ (0.152) \end{array}$	-0.851*[0.448](0.057)	$\begin{array}{c} 0.178\\ [0.315]\\ (0.572) \end{array}$	-1.138** [0.548] (0.038)	0.779** [0.341] (0.022)	$\begin{array}{c} 0.092 \\ [0.212] \\ (0.666) \end{array}$	$\begin{array}{c} 0.141 \\ [0.298] \\ (0.637) \end{array}$	-0.549 $[0.486]$ (0.259)	$\begin{array}{c} 0.127 \\ [0.376] \\ (0.735) \end{array}$
Investment of time minus DV	$\begin{array}{c} 0.416^{**} \\ [0.184] \\ (0.024) \end{array}$	$\begin{array}{c} 0.669^{**} \\ [0.297] \\ (0.024) \end{array}$	$\begin{array}{c} 0.475^{*} \\ [0.279] \\ (0.089) \end{array}$	$\begin{array}{c} 0.572^{**}\\ [0.290]\\ (0.048) \end{array}$	$\begin{array}{c} 0.568^{**}\\ [0.289]\\ (0.049) \end{array}$	$\begin{array}{c} 0.632^{**}\\ [0.293]\\ (0.031) \end{array}$	$\begin{array}{c} 0.521^{*} \\ [0.279] \\ (0.062) \end{array}$	$\begin{array}{c} 0.014 \\ [0.156] \\ (0.930) \end{array}$	-0.060 [0.089] (0.501)	-0.856^{***} $[0.331]$ (0.010)	$\begin{array}{c} 1.080^{***} \\ [0.374] \\ (0.004) \end{array}$	-0.225 [0.264] (0.394)	$\begin{array}{c} 1.552^{***} \\ [0.459] \\ (0.001) \end{array}$	-0.344 [0.285] (0.227)	$\begin{array}{c} 0.406^{*} \\ [0.210] \\ (0.054) \end{array}$	-0.452 [0.283] (0.111)	$\begin{array}{c} 1.436^{***} \\ [0.454] \\ (0.002) \end{array}$	-0.837** [0.351] (0.017)
Lagged index of cog ability	$\begin{array}{c} 0.386^{***} \\ [0.044] \\ (0.000) \end{array}$	-0.024 [0.066] (0.720)	$\begin{array}{c} 0.054 \\ [0.062] \\ (0.380) \end{array}$	$\begin{array}{c} -0.024 \\ [0.064] \\ (0.705) \end{array}$	$\begin{array}{c} -0.031 \\ [0.064] \\ (0.629) \end{array}$	-0.022 [0.065] (0.738)	-0.045 [0.062] (0.466)	$\begin{array}{c} 0.047 \\ [0.033] \\ (0.154) \end{array}$	$\begin{array}{c} 0.058^{***} \\ [0.020] \\ (0.004) \end{array}$	$\begin{array}{c} 0.160^{**} \\ [0.074] \\ (0.030) \end{array}$	$\begin{array}{c} -0.111\\ [0.083]\\ (0.182) \end{array}$	$\begin{array}{c} 0.019 \\ [0.059] \\ (0.747) \end{array}$	-0.111 [0.102] (0.273)	-0.024 [0.063] (0.703)	$\begin{array}{c} 0.324^{***} \\ [0.045] \\ (0.000) \end{array}$	$\begin{array}{c} 0.114^{*} \\ [0.065] \\ (0.077) \end{array}$	-0.263^{**} [0.102] (0.010)	$\begin{array}{c} 0.169^{**} \\ [0.079] \\ (0.033) \end{array}$
Mother's capital	$\begin{array}{c} 0.227^{***} \\ [0.030] \\ (0.000) \end{array}$	$\begin{array}{c} 0.072^{*} \\ [0.039] \\ (0.062) \end{array}$	$\begin{array}{c} 0.019 \\ [0.036] \\ (0.602) \end{array}$	$\begin{array}{c} 0.072^{*} \\ [0.038] \\ (0.058) \end{array}$	$\begin{array}{c} 0.051 \\ [0.038] \\ (0.181) \end{array}$	$\begin{array}{c} 0.081^{**} \\ [0.038] \\ (0.034) \end{array}$	$\begin{array}{c} 0.064^{*} \\ [0.036] \\ (0.082) \end{array}$	$\begin{array}{c} 0.092^{***} \\ [0.020] \\ (0.000) \end{array}$	$\begin{array}{c} 0.018 \\ [0.012] \\ (0.141) \end{array}$	$\begin{array}{c} 0.024 \\ [0.044] \\ (0.575) \end{array}$	$\begin{array}{c} 0.046 \\ [0.049] \\ (0.343) \end{array}$	$\begin{array}{c} 0.045 \\ [0.035] \\ (0.193) \end{array}$	$\begin{array}{c} 0.159^{***} \\ [0.060] \\ (0.008) \end{array}$	0.097^{***} [0.037] (0.009)	$\begin{array}{c} 0.252^{***} \\ [0.026] \\ (0.000) \end{array}$	$\begin{array}{c} 0.039 \\ [0.036] \\ (0.287) \end{array}$	0.229^{***} [0.059] (0.000)	-0.061 [0.045] (0.184)
Wealth index (w. 1)	$\begin{array}{c} 0.157^{***} \\ [0.040] \\ (0.000) \end{array}$	$\begin{array}{c} 0.024 \\ [0.057] \\ (0.672) \end{array}$	$\begin{array}{c} -0.054 \\ [0.054] \\ (0.315) \end{array}$	-0.019 [0.056] (0.729)	$\begin{array}{c} 0.039 \\ [0.056] \\ (0.485) \end{array}$	$\begin{array}{c} 0.065 \\ [0.057] \\ (0.251) \end{array}$	$\begin{array}{c} 0.033 \\ [0.054] \\ (0.543) \end{array}$	$\begin{array}{c} 0.043 \\ [0.029] \\ (0.141) \end{array}$	$\begin{array}{c} 0.010 \\ [0.018] \\ (0.567) \end{array}$	-0.041 [0.065] (0.529)	-0.066 [0.073] (0.365)	$\begin{array}{c} 0.065 \\ [0.051] \\ (0.203) \end{array}$	$\begin{array}{c} 0.139\\ [0.089]\\ (0.116) \end{array}$	-0.026 [0.055] (0.634)	$\begin{array}{c} 0.124^{***} \\ [0.040] \\ (0.002) \end{array}$	-0.013 [0.056] (0.811)	$\begin{array}{c} 0.113 \\ [0.090] \\ (0.209) \end{array}$	-0.082 [0.070] (0.238)
Family size	-0.026 [0.020] (0.187)	$\begin{array}{c} 0.005 \\ [0.030] \\ (0.864) \end{array}$	-0.011 [0.028] (0.687)	$\begin{array}{c} 0.030 \\ [0.029] \\ (0.305) \end{array}$	$\begin{array}{c} 0.002 \\ [0.029] \\ (0.950) \end{array}$	-0.002 [0.029] (0.951)	$\begin{array}{c} 0.001 \\ [0.028] \\ (0.977) \end{array}$	-0.041^{***} [0.015] (0.008)	-0.018^{*} $[0.009]$ (0.051)	-0.032 [0.033] (0.336)	$\begin{array}{c} 0.065^{*} \\ [0.037] \\ (0.080) \end{array}$	-0.040 [0.026] (0.123)	$\begin{array}{c} 0.117^{**} \\ [0.046] \\ (0.010) \end{array}$	$\begin{array}{c} -0.077^{***} \\ [0.028] \\ (0.006) \end{array}$	-0.006 [0.019] (0.771)	-0.024 [0.027] (0.385)	$\begin{array}{c} 0.061 \\ [0.044] \\ (0.166) \end{array}$	-0.053 [0.034] (0.121)
Two-parent HH	-0.009 $[0.020]$ (0.652)	-0.008 [0.029] (0.780)	-0.022 [0.027] (0.430)	-0.002 [0.028] (0.945)	-0.012 [0.028] (0.680)	$\begin{array}{c} 0.007 \\ [0.029] \\ (0.808) \end{array}$	-0.012 [0.027] (0.675)	-0.008 [0.015] (0.611)	$\begin{array}{c} 0.008 \\ [0.009] \\ (0.359) \end{array}$	-0.031 [0.033] (0.337)	$\begin{array}{c} 0.016 \\ [0.037] \\ (0.656) \end{array}$	-0.029 [0.026] (0.268)	$\begin{array}{c} 0.027 \\ [0.045] \\ (0.548) \end{array}$	-0.029 [0.028] (0.290)	$\begin{array}{c} 0.008 \\ [0.019] \\ (0.657) \end{array}$	-0.031 [0.027] (0.254)	-0.021 [0.044] (0.637)	$\begin{array}{c} 0.004 \\ [0.034] \\ (0.900) \end{array}$
Income (w. 2)	$\begin{array}{c} 0.004 \\ [0.055] \\ (0.945) \end{array}$	-0.070 [0.085] (0.411)	$\begin{array}{c} 0.033 \\ [0.080] \\ (0.679) \end{array}$	-0.069 [0.083] (0.407)	$\begin{array}{c} -0.102 \\ [0.083] \\ (0.217) \end{array}$	-0.054 [0.084] (0.516)	-0.066 [0.080] (0.409)	$\begin{array}{c} 0.034 \\ [0.043] \\ (0.429) \end{array}$	$\begin{array}{c} 0.004 \\ [0.026] \\ (0.873) \end{array}$	$\begin{array}{c} 0.154 \\ [0.096] \\ (0.108) \end{array}$	$\begin{array}{c} -0.073 \\ [0.107] \\ (0.497) \end{array}$	-0.033 [0.075] (0.661)	-0.055 [0.131] (0.672)	$\begin{array}{c} 0.036 \\ [0.081] \\ (0.654) \end{array}$	-0.027 [0.061] (0.654)	0.255^{***} [0.086] (0.003)	-0.218 [0.139] (0.118)	$\begin{array}{c} 0.314^{***} \\ [0.108] \\ (0.003) \end{array}$
Income sq. (w. 2)	-0.006 [0.048] (0.897)	$\begin{array}{c} 0.070 \\ [0.074] \\ (0.346) \end{array}$	-0.024 [0.070] (0.727)	$\begin{array}{c} 0.061 \\ [0.072] \\ (0.396) \end{array}$	$\begin{array}{c} 0.091 \\ [0.072] \\ (0.207) \end{array}$	$\begin{array}{c} 0.057 \\ [0.073] \\ (0.431) \end{array}$	$\begin{array}{c} 0.073 \\ [0.070] \\ (0.291) \end{array}$	$\begin{array}{c} -0.014 \\ [0.037] \\ (0.711) \end{array}$	$\begin{array}{c} 0.004 \\ [0.023] \\ (0.879) \end{array}$	-0.116 [0.083] (0.165)	$\begin{array}{c} 0.015 \\ [0.094] \\ (0.875) \end{array}$	$\begin{array}{c} 0.015 \\ [0.066] \\ (0.823) \end{array}$	$\begin{array}{c} 0.018 \\ [0.1114] \\ (0.875) \end{array}$	-0.029 $[0.071]$ (0.685)	$\begin{array}{c} 0.030 \\ [0.051] \\ (0.551) \end{array}$	-0.182^{**} [0.072] (0.011)	$\begin{array}{c} 0.145 \\ [0.115] \\ (0.209) \end{array}$	-0.227^{**} [0.089](0.011)
No income reported (w. 2)	-0.035 [0.036] (0.331)	$\begin{array}{c} 0.109^{*} \\ [0.056] \\ (0.051) \end{array}$	$\begin{array}{c} 0.045 \\ [0.053] \\ (0.398) \end{array}$	$\begin{array}{c} 0.067 \\ [0.055] \\ (0.221) \end{array}$	$\begin{array}{c} 0.078 \\ [0.055] \\ (0.154) \end{array}$	$\begin{array}{c} 0.110^{**} \\ [0.055] \\ (0.047) \end{array}$	0.132^{**} [0.053] (0.013)	-0.013 [0.028] (0.645)	-0.000 [0.017] (0.983)	-0.057 [0.063] (0.370)	-0.070 [0.071] (0.320)	$\begin{array}{c} 0.020 \\ [0.050] \\ (0.688) \end{array}$	-0.031 [0.087] (0.718)	-0.049 [0.054] (0.360)	-0.032 [0.040] (0.419)	$\begin{array}{c} 0.006 \\ [0.057] \end{array}$	$\begin{array}{c} 0.009\\ [0.092] \end{array}$	$\begin{array}{c} 0.093 \\ [0.071] \\ (0.191) \end{array}$
Proxy means test for CCT	$\begin{array}{c} 0.039 \\ [0.062] \\ (0.531) \end{array}$	$\begin{array}{c} 0.186^{*} \\ [0.096] \\ (0.052) \end{array}$	$\begin{array}{c} -0.012 \\ [0.090] \\ (0.891) \end{array}$	$\begin{array}{c} 0.130 \\ [0.094] \\ (0.166) \end{array}$	$\begin{array}{c} 0.185^{**}\\ [0.093]\\ (0.048) \end{array}$	$\begin{array}{c} 0.195^{**}\\ [0.095]\\ (0.039) \end{array}$	$\begin{array}{c} 0.206^{**} \\ [0.090] \\ (0.022) \end{array}$	$\begin{array}{c} 0.013 \\ [0.048] \\ (0.788) \end{array}$	-0.041 [0.029] (0.158)	-0.360^{***} [0.108] (0.001)	$\begin{array}{c} 0.178\\ [0.121]\\ (0.143) \end{array}$	$\begin{array}{c} 0.024 \\ [0.086] \\ (0.783) \end{array}$	$\begin{array}{c} 0.364^{**} \\ [0.148] \\ (0.014) \end{array}$	-0.094 [0.092] (0.307)	$\begin{array}{c} 0.078 \\ [0.071] \\ (0.272) \end{array}$	-0.293^{***} $[0.100]$ (0.003)	0.325^{**} $[0.158]$ (0.040)	$\begin{array}{c} -0.374^{***} \\ [0.122] \\ (0.002) \end{array}$
Observations R-squared Anderson-Rubin test A-R F-test p-value	$\begin{array}{c} 1,615\\ 0.116\\ 17.31\\ 6.523\\ 0.000220\end{array}$	$\begin{array}{c} 1,819\\ -0.310\\ 16.33\\ 2.301\\ 0.0754\end{array}$	$\begin{array}{c} 1,819\\ -0.166\\ 16.33\\ 1.185\\ 0.314\end{array}$	$\begin{array}{c} 1,819\\ -0.243\\ 16.33\\ 1.904\\ 0.127\end{array}$	$\begin{array}{c} 1,819\\ -0.247\\ 16.33\\ 1.864\\ 0.134\end{array}$	$\begin{array}{c} 1,819\\ -0.278\\ 16.33\\ 2.059\\ 0.104\end{array}$	1,819 - 0.160 16.33 2.081 0.101	$\begin{array}{c} 1,789\\ 0.035\\ 14.99\\ 0.980\\ 0.401\\ \end{array}$	$\begin{array}{c} 1,791\\ -0.003\\ 17.75\\ 0.182\\ 0.909\\ \end{array}$	$\begin{array}{c} 1,826\\-0.638\\16.69\\6.921\\0.000124\end{array}$	$\begin{array}{c} 1,825\\ -1.086\\ 16.47\\ 9.869\\ 1.87e-06\end{array}$	$\begin{array}{c} 1,814\\ -0.059\\ 16.17\\ 2.386\\ 0.0674\end{array}$	$\begin{array}{c} 1,828\\-2.293\\16.35\\22.37\\0\end{array}$	1,827 -0.216 16.34 3.037 0.0281	$\begin{array}{c} 1,743\\ 0.032\\ 14.20\\ 5.016\\ 0.00183\end{array}$	$\begin{array}{c} 1,762\\ -0.147\\ 15.75\\ 5.335\\ 0.00117\end{array}$	$\begin{array}{c} 1,757\\ -1.979\\ 16.09\\ 13.93\\ 5.59e-09\end{array}$	$\begin{array}{c} 1,757\\ -0.770\\ 16.09\\ 4.388\\ 0.00439\end{array}$
						stand.	ard errors *** p<	in brackets 0.01, ** p<	s, p-values i <0.05, * p<	n parenthese 0.10	es							

Table A.9: IV REGRESSIONS OF THE EFFECT OF "PARENTAL INVESTMENT OF TIME MINUS DOMESTIC VIOLENCE" ON VARIOUS

B Measurement error corrections

The estimates of the latent factor θ_i cannot be used directly in estimations due to the fact that there are measured with error. In particular, let the equation of interest be:

$$Y_i = \alpha \theta_i + \gamma X_i + \varepsilon_i \tag{4}$$

with $E[\varepsilon_i] = 0$ and (θ_i, X_i) statistically independent of ε_i . Let $\theta_{S,i}$ denote the estimate of θ . Assuming that $\theta_{S,i} = \theta_i + V_i$, $E[V_i] = 0$ and (θ_i, X_i) are statistically independent of V_i , the linear regression will produce biased estimates. In particular, Heckman et al. (2013) show that

$$plim\begin{pmatrix} \hat{\alpha}\\ \hat{\gamma} \end{pmatrix} = \begin{pmatrix} COV(\theta_S, \theta_S) & COV(\theta_S, X)\\ COV(X, \theta_S) & COV(X, X) \end{pmatrix}^{-1} \begin{pmatrix} COV(\theta, \theta) & COV(\theta, X)\\ COV(X, \theta) & COV(X, X) \end{pmatrix} \begin{pmatrix} \alpha\\ \gamma \end{pmatrix}_{(5)}$$

In the formulas above, the terms in $COV(\theta, \theta)$ can be obtained from the estimates of the measurement system. Unbiased estimates for the linear regression can be obtained by pre-multiplying the OLS estimates by the inverse of the bias term.

There is reason to suspect that some of the latent variables are endogenous. This is the case for parental investment and domestic violence. Suppose that the equation of interest is now $Y_i = \alpha \theta_{i,1} + \gamma \theta_{i,2} + \varepsilon_i$ with $E[\varepsilon_i] = 0$ and $(\theta_{i,2}, Z_i)$ statistically independent of ε_i . In this context, Z_i is an instrumental variable for $\theta_{i,1}$. Using similar arguments, it can be shown that instrumental variable estimates will also be biased if some of the controls are latent variables measured with error. In particular,

$$plim\begin{pmatrix} \hat{\alpha}_{IV} \\ \hat{\gamma}_{IV} \end{pmatrix} = \begin{pmatrix} COV(Z,\theta_{S,1}) & COV(Z,\theta_{S,2}) \\ COV(\theta_{S,2},\theta_{S,1}) & COV(\theta_{S,2},\theta_{S,2}) \end{pmatrix}^{-1} \begin{pmatrix} COV(Z,\theta_1) & COV(Z,\theta_2) \\ COV(\theta_2,\theta_1) & COV(\theta_2,\theta_2) \end{pmatrix} \begin{pmatrix} \alpha \\ \gamma \end{pmatrix}$$
(6)

As in the case of linear regression, the terms in the correction matrix can be obtained from the estimates of the measurement system. In the case where there are more instruments than endogenous regressors, the formula is:

$$plim \ \hat{\beta}_{IV} = ((X'Z)_S (Z'Z)_S^{-1} (Z'X)_S)^{-1} ((X'Z)_S (Z'Z)_S^{-1} (Z'X))\beta$$
(7)

where the term $(X'Z)_S$ denotes the sample covariance between X and Z and (X'Z) denotes an unbiased estimator of the covariance between X and Z. These estimates can be obtained from the estimates of the measurement system.

Another remaining potential challenge to the validity of this study is the presence of

weak instruments or partially identified parameters. I use the reduce form method recently proposed by Chernozhukov and Hansen (2008b) which is robust to the presence of weak instruments. The idea of this method is to construct confidence intervals of the coefficients of interest by testing the significance of the instruments on the residuals of the outcome equation under the hypothesis that the assumed coefficient is correct. So, for instance, if the true equation is $Y_i = \alpha X_i + \varepsilon_i$ and Z_i is a valid instrument, then $Y_i - \alpha X_i$ must be independent of Z_i . I use this approach to determine the significance of parental investment and domestic violence. Since the method requires eliminating the influence of other covariates prior to performing the tests, this adjustments is made using estimates that account for the possibility of measurement error as explained above. In particular, I use the residuals of the regression of outcome Y_i , endogenous latent factor $\theta_{1,i}$ and instrument Z_i on variables $(\theta_{2,i}, X_i)$ whereby the estimates of these regressions account for the fact that latent variable $\theta_{2,i}$ is measured with error. The method can also be used to determine the set of partially identified parameters when the regression model is under-identified. In particular, if the true equation is $Y_i = \alpha_1 X_{1,i} + \alpha_2 X_{2,i} + \varepsilon_i$ and Z_i is a valid instrument for both variables, then $Y_i - \alpha_1 X_{1,i} - \alpha_2 X_{2,i}$ must be independent of Z_i .

C Elegibility formula for conditional cash transfer program - JUNTOS

Households qualify for a CCT if they lived in one of the districts scheduled for expansion of the program and if the household had a mother or pregnant woman, a widower or custodian with children below 15 years of age. To determined the individual level eligibility to receive a CCT, the Peruvian Institute of Statistics estimated a logit model of the probability that a household is poor based of the 2001 and 2004 Peruvian households survey. In particular, the estimated model is:

$$y = \alpha + \beta X + \varepsilon$$

where Y equals 1 if the household is below the poverty line. X is a set of proxy variables that include: 1. x_1 = percent of adult females in the households, 2. x_2 = Percent of minors in the household that attend school, 3. x_3 = A dummy variable if the household uses industrial sources of energy (gas, electricity or kerosene), 4. x_4 = Number of household appliances absent in the household, 5. x_5 = Access to electricity, water and sewage, 6. x_6 = Lowest quality construction materials, 7. x_7 = Low quality construction materials, 8. x_8 = Mid quality construction materials. The estimated parameters used in the assignment program are:

 $y = -1.346104 + 1.183232 * x_1 + 0.2275956 * x_2 - 0.7623876 * x_3 + 0.4445512 * x_4 - 0.376869 * x_5 - 0.2593025 * x_6 - 0.8584362 * x_7 - 1.317246 * x_8$

A household is considered poor, or eligible for the transfer, if $\hat{P} = \frac{exp(y)}{1+exp(y)} \ge .7567477$. I use this method, together with the deployment plan of the CCT, to determine if a household is predicted to receive a transfer. The variable \hat{P} is used as an additional control in the regressions.

D Simulated and sample moments used in the calibration of Model 1

Mean	Observed	Simulated
Cognitive Development	0.035	0.044
Log of wage (mother)	0.389	0.217
Log of wage (father)	0.466	0.706
Hours worked for wage (mother)	21.628	27.477
Hours worked for wage (father)	47.545	54.497
Other income	16.784	1.736
Weakly consumption	253.129	178.849
Parental money investments	23.604	20.991
Reported domestic violence in 3rd wave	0.071	0.045
Reported domestic violence in 2nd wave	0.087	0.072
Asked for help regarding domestic violence (3rd wave)	0.109	0.101
Time investment (mother)	6.513	6.833
Knows children's friends' names	0.841	1.000
Encourage child to read $(1,2,3 \text{ levels})$	2.449	2.417
Help child with homework	0.623	1.000
Nonzero hours worked for wage (mother)	0.705	0.539
Nonzero hours worked for wage (father)	0.996	0.894
Nonzero hours worked for wage (mother & father)	0.701	0.434
Reported domestic violence in 2nd and 3rd wave	0.020	0.000
Reported domestic violence and asked for help in 3rd wave	0.017	0.023
Reported domestic violence in 2nd wave and asked for help in 3rd wave	0.017	0.000
Knows child's friends' name and Encourage to read sometimes	0.772	0.681
Knows child's friends' name and helps with homework	0.552	0.973
Encourages to read sometimes and helps with homework	0.596	0.682
Knows child's friends' name and Encourage to read frequently	0.445	0.446
Helps with homework and Encourage to read frequently	0.341	0.446
Reported domestic violence in 3rd wave X Maternal time investment	0.445	0.659
Did not report domestic violence in 3rd wave X Maternal time investment	6.068	6.174
Reported domestic violence in 2nd wave X Maternal time investment	0.571	0.562
Did not report domestic violence in 2nd wave X Maternal time investment	5.942	6.271
Reported domestic violence in 3rd wave and knows child's friends' names	0.056	0.043
Reported domestic violence in 3rd wave and encourages child to read sometimes	0.065	0.044
Reported domestic violence in 3rd wave and encourages child to read frequently	0.036	0.044
Reported domestic violence in 3rd wave and helps child with homework	0.039	0.045
Reported domestic violence in 2nd wave and knows child's friends' names	0.074	0.070

Reported domestic violence in 2nd wave and encourages child to read sometimes	0.081	0.070
Reported domestic violence in 2nd wave and encourages child to read frequently	0.039	0.037
Reported domestic violence in 2nd wave and helps child with homework	0.050	0.072
Asked for help regarding domestic violence and knows child's friends' names	0.092	0.098
Asked for help regarding domestic violence and encourages child to read sometimes	0.101	0.078
Asked for help regarding domestic violence and encourages child to read frequently	0.052	0.042
Asked for help regarding domestic violence and helps child with homework	0.074	0.101
Lottery choice equals 1	0.088	0.077
Lottery choice equals 2	0.103	0.122
Lottery choice equals 3	0.081	0.074
Lottery choice equals 4	0.118	0.104
Lottery choice equals 5	0.092	0.080
Lottery choice equals 6	0.519	0.543

Standard deviations	Observed	Simulated
Cognitive development index	0.796	0.764
Log of wage (mother)	1.342	1.039
Log of wage (father)	1.193	1.072
Hours worked for wage (mother)	23.997	32.532
Hours worked for wage (mother)	19.828	34.599
Other income	128.874	1.598
Weakly consumption	168.485	229.045
Parental money investments	50.134	30.037
Time investment (mother)	4.455	4.627

Correlations	Observed	Simulated
correlation of cognitive and risk attitude	-0.037	0.055
correlation of mother's and father's wages	0.374	0.120
correlation of mother's and father's hours worked	0.010	-0.135
correlation of mother's wage and mother's hours worked	-0.232	0.404
correlation of father's wage and father's hours worked	-0.045	0.167
correlation of father's wage and mother's hours worked	-0.031	-0.532
correlation of mother's wage and father's hours worked	0.090	-0.284

Conditional means of father's wages	Observed	Simulated
Father finished at least high school	0.941	0.868
Mother finished at least high school	1.036	0.846
Father finished at least primary school	0.004	0.565
Mother finished at least primary school	0.135	0.655

Father is at least 40 years old	0.443	0.700
Mother is at least 35 years old	0.455	0.707
Father is less than 40 years old	0.317	0.664
Mother is less than 35 years old	0.379	0.692

Conditional means of mother's wages	Observed	Simulated
Father finished at least high school	0.762	0.320
Mother finished at least high school	0.859	0.389
Father finished at least primary school	-0.301	0.056
Mother finished at least primary school	-0.167	0.074
Father is at least 40 years old	0.405	0.216
Mother is at least 35 years old	0.448	0.230
Father is less than 40 years old	0.304	0.184
Mother is less than 35 years old	0.186	0.181

Conditional mean of father's hours worked	Observed	Simulated
Father finished at least high school	49.814	55.160
Mother finished at least high school	50.894	53.130
Father finished at least primary school	44.902	53.992
Mother finished at least primary school	47.150	55.020
Father is at least 40 years old	46.663	54.980
Mother is at least 35 years old	47.552	54.824
Father is less than 40 years old	47.232	53.817
Mother is less than 35 years old	47.476	54.373

Conditional mean of mother's hours worked	Observed	Simulated
Father finished at least high school	23.122	26.886
Mother finished at least high school	23.071	28.748
Father finished at least primary school	22.177	27.930
Mother finished at least primary school	20.644	26.970
Father is at least 40 years old	21.614	27.034
Mother is at least 35 years old	24.284	27.182
Father is less than 40 years old	19.455	28.103
Mother is less than 35 years old	19.038	27.591

Other	Observed	Simulated
Time investment conditional on mother finishing HS	8.599	6.693
Time investment conditional on mother finishing primary education	5.404	6.887
Time investment conditional on father finishing HS	8.078	6.897

Time investment conditional on father finishing primary education	4.818	6.784
correlation of mother's time investment and cognitive development	0.272	0.129
correlation of mother's time investment and lottery choices	-0.009	0.105
correlation of monetary investment and cognitive development	0.140	0.224
correlation of mother's wage and father's hours worked	-0.006	-0.144
correlation of mothers wage and cognitive development	0.252	0.173
correlation of mother's wage and lottery choices	-0.019	-0.093
correlation of father's wage and cognitive development	0.226	0.189
correlation of father's wage and lottery choices	-0.003	0.124
correlation of violence and cognitive development	-0.062	-0.010
correlation of violence and lottery choices	-0.052	-0.048
correlation of mother's wage and violence	-0.031	-0.053
correlation of father's wage and violence	0.002	0.000

Conditional mean of domestic violence	Observed	Simulated
Father finished at least high school	0.256	0.211
Mother finished at least high school	0.253	0.197
Father finished at least primary school	0.234	0.224
Mother finished at least primary school	0.318	0.223
Father is at least 40 years old	0.260	0.224
Mother is at least 35 years old	0.273	0.224
Father is less than 40 years old	0.259	0.212
Mother is less than 35 years old	0.249	0.214

E A strategic model of household decisions with domestic violence

In this section I discuss the results of an alternative structural model of household dynamics in which domestic violence is instrumental. This exercise is intended to probe the sensitivity of results to modelling assumptions. In this model husbands are principals that sometimes use violence to extract work from wives, the agents. The model builds on Weinberg (2001)'s research on the use of corporal punishment by parents. Husbands care about the time wives devote to households chores (t_h) , but this effort is not observable.²⁵ Husbands only receive a signal, high or low, which is correlated with effort. The probability of receiving a high signal is increasing in the time a wife devotes to household chores. In order to extract effort, husbands make transfers conditional on the observed signal. These transfers are above the minimum required level set by law or custom.²⁶ In this model, wives do not derive any utility from household chores. They perform these tasks either because they receive compensation or to avoid violence. Husbands are altruistic towards their wives (love which is assumed not reciprocated for simplicity). Husbands refraining from violence need to compensate for their wives opportunity cost of time. Husbands unable to compensate their wives enough trade-off between enjoying the benefits of household chores and the utility cost imposed on their wives.

I will sketch the main elements of the model and present a complete derivation of the estimated model in Appendix F. Using the notation from the previous section, mothers are assumed to solve the problem below,

$$Max \ \alpha_c ln(c_m) + \alpha_l ln(l_m) + \alpha_{\theta} ln(\theta_1) + \alpha_{\sigma_1} ln(\sigma_1) - \alpha_V V \tag{8}$$

$$c_m + I + w_m (l_m + \tau_m + t_h) \le Y_m + \underline{Y} + \Phi(t_h) \Delta Y + w_m T \tag{9}$$

$$ln\theta_1 = A + a_\theta ln(\theta_0) + a_{\theta_m} ln(\theta_m) + a_{\tau_m} ln(\tau_m) + a_I ln(I) + a_V V + a_X X + \epsilon_\theta \tag{10}$$

$$ln\sigma_1 = B + b_\sigma ln(\sigma_0) + b_{\theta_m} ln(\theta_m) + b_{\tau_m} ln(\tau_m) + b_I ln(I) + b_V V + b_X X + \epsilon_\sigma$$
(11)

Wives/mothers derive utility from consumption (c_m) , leisure (l_m) , the cognitive development of the child (θ_1) and the child's preferences (σ_1) subject to a budget constraint and technology constraints. I assume, for the moment, that violence (V) is 0. The transfer wives receive from fathers is $\underline{Y} + \Delta Y$ if the signal is good and \underline{Y} if the signal is bad. The probability of obtaining a good signal is $\Phi(t_h)$ where $\Phi(.)$ is a cumulative distribution

²⁵This assumption is reasonable in the current context. Most husbands in the sample work and only observe signals of their wife's efforts like cleanliness of the household, quality of meals, etc.

²⁶In Peru, fathers are required to give at least a third of their earned income to ex-wives and the law requires that transfers are made conditional on the number of children. This is an extra parameter in the model.

function. The budget constraint is written under the assumption that wives can obtain an actuarially fair insurance on the signal-contingent transfers. This assumption simplifies the analysis greatly and explains the way the budget constraint is written. Wives have then an opportunity to obtain transfers from their husbands by performing household chores and husbands can *choose* the level of t_h by varying ΔY .

Let $V_m(w_m, Y_m, \underline{Y}, \Delta Y)$ be the indirect utility function of the wife when facing market wage w_m , independent income Y_m and contract $(\underline{Y}, \Delta Y)$. The problem that the husband solves is:

$$max \ \beta_c ln(c_f) + \beta_l ln(l_f) + \beta_h ln(t_h) + \beta_m V_m(w_m, Y_m, \underline{Y}, \Delta Y)$$
(12)

$$c_f + w_f l_f + \underline{Y} + \Delta Y \le Y_f + w_f T \tag{13}$$

$$t_h is optimal$$
 (14)

The equation above assumes that husbands can insure income outlays as wives do. Again, this is done to simplify the analysis and estimations. Note that husbands care about their children only because they care about their wives' utility.

Suppose that $\Phi(t_h) = \frac{t_h}{T}$. That is, the probability of receiving a good signal is uniformly distributed. Since wives do not derive utility from performing household chores, they will find working for a wage or for their husbands as perfect substitutes. If a husband cannot pay their wives' market wage, they will not provide any house work. Since husbands consider household chores essential goods, they will use violence as a way to extract effort from their wives any time this is more profitable. In other words, if wives dislike violence, husbands can use violence whenever a bad signal is observed. The "optimal" amount of violence will depend on the preference parameters of both husbands and wives. Even in this simple set-up, the model does not have a closed-form solution. Additional details on the solution of the model are provided in F.

Table E.2 in this Appendix shows the fit of the model for the 122 moments used to calibrate it.²⁷ While this model is able to reproduce many features of the data, the fit is worse than the model in the previous section. The weighted sum of moments (divided by the number of moments) is over two and half times larger than in the previous model. Figure E.1 compares the simulated and observed distribution of the cognitive development index and shows that the model fit is less precise. Regarding the prevalence of domestic violence, Table E.2 in Appendix F shows that the model predicts this variable reasonably well.

²⁷The consumption of the wife and husband are separate in this specification adding two extra moments to the model, one for the mean and one for the variance.

Table E.1 presents estimates of preference parameters, the cognitive development function and the risk attitudes function. In this version of the model, only mothers care about the cognitive development and preferences of their children. While the weight of cognitive development in the mother's utility is small, there is a high degree of preferences heterogeneity as expressed by the coefficient of variation of the preference parameter. I find that the parameters of the cognitive development function and risk attitudes function are quite similar across models with the exception that this model estimates the effect of domestic violence on cognitive development to be much larger.

In sum, the structural estimation produce similar results as the instrumental variable approach.

Figure E.1: MODEL 2. CUMULATIVE DISTRIBUTION OF OBSERVED AND PREDICTED INDEX OF COGNITIVE DEVELOPMENT



Mean	Observed	Simulated
Cognitive Development	0.035	-0.050
Log of wage (mother)	0.389	0.342
Log of wage (father)	0.466	0.567
Hours worked for wage (mother)	21.628	29.984
Hours worked for wage (father)	47.545	50.410
Other income	16.784	6.575
Weakly consumption (mother)	84.376	71.334
Weakly consumption (father)	168.753	177.031
Parental money investments	23.604	0.036
Reported domestic violence in 3rd wave	0.071	0.100
Reported domestic violence in 2nd wave	0.087	0.080
Asked for help regarding domestic violence (3rd wave)	0.109	0.224
Time investment (mother)	6.513	0.047
Knows children's friends' names	0.841	1.000
Encourage child to read $(1,2,3 \text{ levels})$	2.449	1.938
Help child with homework	0.623	0.240
Nonzero hours worked for wage (mother)	0.705	0.934
Nonzero hours worked for wage (father)	0.996	0.981
Nonzero hours worked for wage (mother & father)	0.701	0.915
Reported domestic violence in 2nd and 3rd wave	0.020	0.060
Reported domestic violence and asked for help in 3rd wave	0.017	0.080
Reported domestic violence in 2nd wave and asked for help in 3rd wave	0.017	0.060
Knows child's friends' name and Encourage to read sometimes	0.772	0.972
Knows child's friends' name and helps with homework	0.552	0.233
Encourages to read sometimes and helps with homework	0.596	0.234
Knows child's friends' name and Encourage to read frequently	0.445	0.064
Helps with homework and Encourage to read frequently	0.341	0.019
Reported domestic violence in 3rd wave X Maternal time investment	0.445	0.001
Did not report domestic violence in 3rd wave X Maternal time investment	6.068	0.046
Reported domestic violence in 2nd wave X Maternal time investment	0.571	0.000
Did not report domestic violence in 2nd wave X Maternal time investment	5.942	0.046
Reported domestic violence in 3rd wave and knows child's friends' names	0.056	0.097
Reported domestic violence in 3rd wave and encourages child to read sometimes	0.065	0.097
Reported domestic violence in 3rd wave and encourages child to read frequently	0.036	0.000
Reported domestic violence in 3rd wave and helps child with homework	0.039	0.000
Reported domestic violence in 2nd wave and knows child's friends' names	0.074	0.078
Reported domestic violence in 2nd wave and encourages child to read sometimes	0.081	0.078
Reported domestic violence in 2nd wave and encourages child to read frequently	0.039	0.000

Reported domestic violence in 2nd wave and helps child with homework	0.050	0.000
Asked for help regarding domestic violence and knows child's friends' names	0.092	0.218
Asked for help regarding domestic violence and encourages child to read sometimes	0.101	0.218
Asked for help regarding domestic violence and encourages child to read frequently	0.052	0.000
Asked for help regarding domestic violence and helps child with homework	0.074	0.040
Lottery choice equals 1	0.088	0.150
Lottery choice equals 2	0.103	0.133
Lottery choice equals 3	0.081	0.017
Lottery choice equals 4	0.118	0.042
Lottery choice equals 5	0.092	0.089
Lottery choice equals 6	0.519	0.569

Standard deviations	Observed	Simulate
Cognitive development index	0.796	0.972
Log of wage (mother)	1.342	1.044
Log of wage (father)	1.193	1.083
Hours worked for wage (mother)	23.997	34.251
Hours worked for wage (mother)	19.828	34.307
Other income	128.874	16.671
Weakly consumption (mother)	56.162	153.540
Weakly consumption (father)	112.323	282.496
Parental money investments	50.134	0.105
Time investment (mother)	4.455	0.048

Correlations	Observed	Simulated
correlation of cognitive and risk attitude	-0.037	0.319
correlation of mother's and father's wages	0.374	0.122
correlation of mother's and father's hours worked	0.010	-0.363
correlation of mother's wage and mother's hours worked	-0.232	0.017
correlation of father's wage and father's hours worked	-0.045	0.238
correlation of father's wage and mother's hours worked	-0.031	-0.144
correlation of mother's wage and father's hours worked	0.090	-0.469

Conditional means of father's wages	Observed	Simulated
Father finished at least high school	0.941	0.780
Mother finished at least high school	1.036	0.752
Father finished at least primary school	0.004	0.383
Mother finished at least primary school	0.135	0.505
Father is at least 40 years old	0.443	0.502

Mother is at least 35 years old	0.455	0.528	
Father is less than 40 years old	0.317	0.564	
Mother is less than 35 years old	0.379	0.583	

Conditional means of mother's wages	Observed	Simulated
Father finished at least high school	0.762	0.456
Mother finished at least high school	0.859	0.532
Father finished at least primary school	-0.301	0.163
Mother finished at least primary school	-0.167	0.185
Father is at least 40 years old	0.405	0.333
Mother is at least 35 years old	0.448	0.347
Father is less than 40 years old	0.304	0.314
Mother is less than 35 years old	0.186	0.312

Conditional mean of father's hours worked	Observed	Simulated
Father finished at least high school	49.814	50.433
Mother finished at least high school	50.894	50.430
Father finished at least primary school	44.902	50.391
Mother finished at least primary school	47.150	50.404
Father is at least 40 years old	46.663	50.400
Mother is at least 35 years old	47.552	50.404
Father is less than 40 years old	47.232	50.411
Mother is less than 35 years old	47.476	50.413

Conditional mean of mother's hours worked	Observed	Simulated
Father finished at least high school	23.122	30.108
Mother finished at least high school	23.071	30.090
Father finished at least primary school	22.177	30.004
Mother finished at least primary school	20.644	29.985
Father is at least 40 years old	21.614	29.911
Mother is at least 35 years old	24.284	29.946
Father is less than 40 years old	19.455	29.998
Mother is less than 35 years old	19.038	30.005

Other	Observed	Simulated
Time investment conditional on mother finishing HS	8.599	0.047
Time investment conditional on mother finishing primary education	5.404	0.047
Time investment conditional on father finishing HS	8.078	0.047
Time investment conditional on father finishing primary education	4.818	0.047

correlation of mother's time investment and cognitive development	0.272	0.222
correlation of mother's time investment and lottery choices	-0.009	0.073
correlation of monetary investment and cognitive development	0.140	0.087
correlation of mother's wage and father's hours worked	-0.006	-0.173
correlation of mothers wage and cognitive development	0.252	0.104
correlation of mother's wage and lottery choices	-0.019	-0.168
correlation of father's wage and cognitive development	0.226	-0.145
correlation of father's wage and lottery choices	-0.003	-0.090
correlation of violence and cognitive development	-0.062	-0.503
correlation of violence and lottery choices	-0.052	-0.489
correlation of mother's wage and violence	-0.031	-0.134
correlation of father's wage and violence	0.002	0.248

Conditional mean of domestic violence	Observed	Simulated
Father finished at least high school	0.256	0.400
Mother finished at least high school	0.253	0.401
Father finished at least primary school	0.234	0.401
Mother finished at least primary school	0.318	0.404
Father is at least 40 years old	0.260	0.407
Mother is at least 35 years old	0.273	0.406
Father is less than 40 years old	0.259	0.404
Mother is less than 35 years old	0.249	0.403

Preference parar	neters		
Variable	$\operatorname{coefficient}$	s.e.	CV
Wife's profess	1005		
α_{α} (Consumption)	$\frac{1005}{0}$ 5740	0.0044	1 88
α_C (Consumption)	0.3740	0.0044	2.04
α_l (Leisure)	0.3000	0.0039	468.85
α_{θ} (Bisk attitudes)	0.0020	0.0000	400.02
α_{σ} (Demostic violence)	26 0211	0.0000 0.3614	0.03
α_V (Domestic violence) Husband's profes	-20.0211	0.3014	0.05
β_{α} (Consumption)	$\frac{10000}{0.2271}$	0.0001	6 21
β_C (Consumption) β_i (Leisure)	0.2271 0.6704	0.0001	0.31 2.70
β_l (Leisure) β_r (Wife's housework)	0.0794 0.0715	0.0002	2.10
β_h (Wife's nousework) β_h (Wife's utility)	0.0715	0.0000	20.07
ρ_m (whe submity)	0.0220	0.0000	
Variable	acofficient		
Variable	coenicient	s.e.	
a_{τ} (Mother's investment (time))	0.0442	0.0011	
a_I (Parental investment (money))	0.0131	0.0000	
a_{θ_0} (Cognitive development (lagged))	1.5436	0.0007	
a_V (Domestic Violence)	-0.0792	0.0001	
$a_{\theta_{}}$ (Mother's capital)	-0.1091	0.0001	
a_{bb} (Household size)	-0.0437	0.0000	
a., (Wealth)	-0.0350	0.0000	
Risk attitudes fu	nction		
Variable	coefficient	s.e.	
b_{τ_m} (Mother's investment (time))	0.0170	0.0000	
b_I (Parental investment (money))	0.0004	0.0000	
b_{θ_0} (Cognitive development (lagged))	0.0235	0.0000	
b_V (Domestic Violence)	-0.2596	0.0003	
b_{θ_m} (Mother's capital)	0.0017	0.0000	
b_{hh} (Household size)	-0.1323	0.0001	
b_{α} (Wealth)	0.0890	0.0001	

Table E.1: MODEL 2: STRUCTURAL PARAMETERS

Standard errors are calculated as the standard deviation of parameter estimates of 20 bootstrap iterations.

F An alternative model of household decisions - Derivations

No violence. As a first approximation, I assume that $\Phi(t_h) = \frac{t_c}{T}$. The problem of the mother when violence is exogenous is:

$$Max \ \alpha_c ln(c_m) + \alpha_l ln(T - \tau_m - t_h - t_w) + \alpha_\theta ln(\theta_1) + \alpha_{\sigma_1} ln(\sigma_1) - \alpha_{\mathcal{V}} \mathcal{V}$$
(15)

$$c_m + I \le Y_m + \underline{Y} + \frac{\Delta I}{T} t_h + w_m t_w \tag{16}$$

$$ln\theta_1 = A + a_\theta ln(\theta_0) + a_{\theta_m} ln(\theta_m) + a_{\tau_m} ln(\tau_m) + a_I ln(I) + a_\mathcal{V} \mathcal{V} + a_X X + \epsilon_\theta$$
(17)

$$ln\sigma_1 = B + b_\sigma ln(\sigma_0) + b_{\theta_m} ln(\theta_m) + b_{\tau_m} ln(\tau_m) + b_I ln(I) + b_{\mathcal{V}} \mathcal{V} + b_X X + \epsilon_\sigma$$
(18)

This problem has a corner solution with $t_h = 0$ if $w_m > \frac{\Delta Y}{T}$. If t_h is an essential good for the husband, then $\frac{\Delta Y}{T} > w^* > w_m$ if this contract is to be used. Note that in this specific case, the father acts as if he is hiring the mother to work at a wage $\frac{\Delta Y}{T}$.

The FOCs for this special case are:

$$\frac{\alpha_c}{c_m} - \lambda = 0 \tag{19}$$

$$-\frac{\alpha_l}{T - \tau_m - t_h} + \lambda \frac{\Delta Y}{T} = 0$$
⁽²⁰⁾

$$-\frac{\alpha_l}{T - \tau_m - t_h} + \frac{\alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}}{\tau_m} = 0$$
(21)

$$\frac{\alpha_{\theta}a_I + \alpha_{\sigma}b_I}{I} - \lambda = 0 \tag{22}$$

$$c_m + I = Y_m + \underline{Y} + \frac{\Delta Y}{T} t_h \tag{23}$$

and
$$t_h(Y_m, \underline{Y}, \Delta Y) = \frac{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I}{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}} T - \frac{\alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}}{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}} \frac{Y_m + \underline{Y}}{\Delta \underline{Y}}.$$

Or, $t_h = \xi T - (1 - \xi) \frac{Y_m + \underline{Y}}{\Delta Y} T$. Then $l_m + \tau_m = (1 - \xi) T + (1 - \xi) \frac{Y_m + \underline{Y}}{\Delta Y} T = (1 - \xi) \frac{Y_m + \underline{Y} + \Delta Y}{\Delta Y} T$.

To derive the complete solution of the mother's problem, an expression is obtained: $Y_m + \underline{Y} + \frac{\Delta Y}{T} t_h. \text{ That is, } Y_m + \underline{Y} + \xi \Delta Y - (1 - \xi)(Y_m + \underline{Y}) = \xi(Y_m + \underline{Y} + \Delta Y) = \frac{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I}{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}} (Y_m + \underline{Y} + \Delta Y).$ This implies that $c_m = \frac{\alpha_c}{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}} (Y_m + \underline{Y} + \Delta Y), I = \frac{\alpha_\theta a_I + \alpha_\sigma b_I}{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}} (Y_m + \underline{Y} + \Delta Y), A = \frac{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}}{Y_m + \underline{Y} + \Delta Y}.$

The father's problem is:

$$max \ \beta_c ln(c_f) + \beta_l ln(l_f) + \beta_h ln(t_h(Y_m, \underline{Y}, \Delta Y)) + \beta_m V_m(Y_m, \underline{Y}, \Delta Y)$$
(24)

$$c_f + w_f l_f + \underline{Y} + \frac{\Delta Y}{T} t_h(Y_m, \underline{Y}, \Delta Y) \le Y_f + w_f T$$
(25)

A simpler version of this model considers that there is a minimum transfer to be made to mothers, \underline{Y} . This reduces the complexity of the model and gives the following FOCs.

$$\frac{\beta_c}{c} - \lambda = 0 \tag{26}$$

$$\frac{\beta_l}{l_f} - \lambda w_f = 0 \tag{27}$$

$$\frac{\beta_h}{t_h(Y_m,\underline{Y},\Delta Y)}\frac{\partial t_h}{\partial \Delta Y} + \beta_m \lambda^* t_h(Y_m,\underline{Y},\Delta Y) - \lambda\xi = 0$$
(28)

$$c_f + w_f l_f + \xi(\underline{Y} + \Delta Y) - (1 - \xi) Y_m \le Y_f + w_f T$$

$$(29)$$

$$\lambda^* = \frac{\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m}}{Y_m + \underline{Y} + \Delta Y}$$
(30)

The model has a nonlinear solution.

We have that $t_h = \xi T - (1 - \xi) \frac{Y_m + \underline{Y}}{\Delta Y} T$. After substitutions: $\frac{\beta_h(1-\xi)(Y_m + \underline{Y})}{\xi \Delta Y (\xi \Delta Y - (1-\xi)(Y_m + \underline{Y}))} + \frac{\beta_m(\alpha_c + \alpha_\theta a_I + \alpha_\sigma b_I + \alpha_l + \alpha_\theta a_{\tau_m} + \alpha_\sigma b_{\tau_m})(\xi \Delta Y - (1-\xi)(Y_m + \underline{Y}))T}{(Y_m + \underline{Y} + \Delta Y)\Delta Y} = \lambda(\Delta Y)$ ΔY solves the following equation: $\frac{\beta_c + \beta_l}{\lambda(\Delta Y)} + \xi(Y_m + \underline{Y} + \Delta Y) = Y_f + w_f T + Y_m$

Violence. I consider now the situation in which a father cannot implement the optimal contract due to the existence of a minimum transfer constraint. As before, fathers make a minimum transfer to mothers in the amount \overline{Y} unconditionally and use a level of domestic violence \mathcal{V} if they observed a bad signal.

After collecting terms, the mothers' indirect utility function is defined as:

$$V_m(w_m, Y_m, \underline{Y}, \mathcal{V}) = max_{c_m, l_m, t_h, \tau_m, I} \{\alpha_c ln(c_m) + \alpha_l ln(l_m) + (\alpha_\theta a_1 + \alpha_{\sigma_1} b_1) ln(\tau_m) + (\alpha_\theta a_2 + \alpha_{\sigma_1} b_2) ln(I) - \gamma_{\mathcal{V}} (1 - \frac{t_h}{T}) \mathcal{V} : c_m + I + w_m (l_m + \tau_m + t_h) \leq Y_m + \underline{Y} + w_m T \}$$

where $\gamma_{\mathcal{V}} = \alpha_{\mathcal{V}} + \alpha_{\theta}a_{\mathcal{V}} + \alpha_{\sigma}b_{\mathcal{V}}$. In this set-up, the utility of the mother is quasi-linear on household chores which simplifies the analysis enormously. The optimal decisions of the mother are:

$$c_m = \frac{\alpha_c}{\gamma_{\mathcal{V}} \mathcal{V}} w_m T \tag{31}$$

$$l_m = \frac{\alpha_l}{\gamma_{\mathcal{V}} \mathcal{V}} T \tag{32}$$

$$\tau_m = \frac{\alpha_\theta a_1 + \alpha_{\sigma_1} b_1}{\gamma_{\mathcal{V}} \mathcal{V}} T \tag{33}$$

$$I = \frac{\alpha_{\theta} a_2 + \alpha_{\sigma_1} b_2}{\gamma_{\mathcal{V}} \mathcal{V}} w_m T \tag{34}$$

$$t_h = \frac{Y_m + \underline{Y}}{w_m} + \frac{\gamma_{\mathcal{V}}\mathcal{V} - (\alpha_c + \alpha_l + \alpha_{\theta}a_1 + \alpha_{\sigma_1}b_1 + \alpha_{\theta}a_2 + \alpha_{\sigma_1}b_2)}{\gamma_{\mathcal{V}}\mathcal{V}}T$$
(35)

If the mother does not work $(t_w = 0)$, then: $c_m = \frac{\alpha_c}{\alpha_c + \alpha_{\theta}a_2 + \alpha_{\sigma_1}b_2}(Y_m + \underline{Y}), I = \frac{\alpha_{\theta}a_2 + \alpha_{\sigma_1}b_2}{\alpha_c + \alpha_{\theta}a_2 + \alpha_{\sigma_1}b_2}(Y_m + \underline{Y}), \tau_m = \frac{\alpha_{\theta}a_1 + \alpha_{\sigma_1}b_1}{\alpha_l}\frac{1}{\gamma_{\mathcal{V}}\mathcal{V}}T$, and $t_h = T - \frac{\alpha_l + \alpha_{\theta}a_1 + \alpha_{\sigma_1}b_1}{\alpha_l}\frac{1}{\gamma_{\mathcal{V}}\mathcal{V}}T$.

The problem of the father is:

$$max \ \beta_c ln(c_f) + \beta_l ln(l_f) + \beta_h ln(t_h) + \beta_m V_m(w_m, Y_m, \underline{Y}, \mathcal{V})$$
(36)

$$c_f + w_f l_f \le Y_f + w_f T - \underline{Y} \tag{37}$$

The optimal level of violence is separate from the consumption decisions since violence has an emotional cost, not a monetary one. The optimal level of \mathcal{V} is the solution of the following equation:

$$\frac{\beta_h}{t_h(w_m, Y_m, \underline{Y}, \mathcal{V})} \frac{\partial t_h(w_m, Y_m, \underline{Y}, \mathcal{V})}{\partial \mathcal{V}} - \gamma_{\mathcal{V}} \left(1 - \frac{t_h(w_m, Y_m, \underline{Y}, \mathcal{V})}{T}\right) = 0$$
(38)

where $t_h(w_m, Y_m, \underline{Y}, \mathcal{V})$ is the optimal decision of the mother facing parameters $(w_m, Y_m, \underline{Y}, \mathcal{V})$ and $\frac{\partial t_h}{\partial \mathcal{V}} = \frac{Y_m + \underline{Y}}{w_m \mathcal{V}} - \frac{t_h}{\mathcal{V}} > 0$. (If $t_w = 0$, then $\frac{\partial t_h}{\partial \mathcal{V}} = \frac{T - t_h}{\mathcal{V}} > 0$) The solution for c_f and l_f are standard. This model has some interesting predictions. First, violence can occur even if both fathers and mothers are relatively rich. Violence can occur if the father does not have the resources to compensate for the mother's time. In that situation, a father will use violence to extract work from the mother. The model can also explain household relationships between parents and children as in Weinberg (2001).